NSF/RA-800640

CALIFORNIA INSTITUTE OF TECHNOLOGY

EARTHQUAKE ENGINEERING RESEARCH LABORATORY

A LINEARIZATION TECHNIQUE FOR THE DYNAMIC RESPONSE OF NONLINEAR CONTINUA

By

Charles Morton Krousgrill, Jr.

EERL 80-08

A Report on Research Conducted Under Grants from the National Science Foundation

> Pasadena, California September, 1980

INFORMATION RESOURCES NATIONAL SCIENCE FOUNDATION

REPRODUCED BY NATIONAL TECHNICAL INFORMATION SERVICE U.S. DEPARTMENT OF COMMERCE SPRINGFIELD, VA 22161 This investigation was sponsored by Grant No. PFR77-23687 from the National Science Foundation, Division of Problem-Focused Research Applications, under the supervision of W. D. Iwan. Any opinions, findings, and conclusions or recommendations expressed in this publication are those of the author and do not necessarily reflect the views of the National Science Foundation.

A LINEARIZATION TECHNIQUE FOR THE DYNAMIC RESPONSE OF NONLINEAR CONTINUA

Thesis by Charles Morton Krousgrill, Jr.

In Partial Fulfillment of the Requirements for the Degree of Doctor of Philosophy

> California Institute of Technology Pasadena, California

1981 (Submitted September 24, 1980)

1.2

ACKNOWLEDGEMENTS

I wish to express appreciation to my research advisor Professor W. D. Iwan for his guidance and encouragement during my stay at Caltech and especially for his patience and support during the preparation of this thesis. Special thanks go to Professor T. J. R. Hughes for invaluable discussions and for making available his finite element computer program LEARN which has been used in the analysis presented in this thesis. I also wish to express my appreciation for the financial support received through the National Science Foundation and the California Institute of Technology.

To the many members of the Society of Professional Students with whom I have had the pleasure of working while at Caltech, I want to extend my sincerest appreciation for their discussions and valuable suggestions. I also wish to thank my friends in the Pasadena community for their unending encouragement throughout the years.

Thanks are due to Eqn and Troff for their patient and skillful typing of this manuscript. The assistance of Ms. Gabi Kassinger, and especially that of Ms. Thelma Valdez, in the preparation of this manuscript is also acknowledged.

Lastly, I thank my parents, to whom this thesis is dedicated, for their support and encouragement throughout the years of my education.

ABSTRACT

The efforts of this dissertation are directed toward the development of a technique for understanding the dynamic response of structural elements governed by nonlinear partial differential equations. This technique is based on the concepts of the equivalent linearization method which relies on obtaining an optimal linear set of equations to model the original nonlinear set.

In this method, the linearization is performed at the continuum level. At this level, the equivalent linear stiffness and damping parameters are physically realizable and are defined in such a way that the method can be easily be incorporated into finite element computer codes.

Three different approaches to the method are taken with each approach based on the minimization of a distinct difference between the nonlinear system and its linear replacement. Existence and uniqueness properties of the minimization solutions are established.

The method is specialized for the treatment of steady-state solutions to harmonic excitation and of stationary response to random excitation. Procedures for solving the equivalent linearization are also discussed.

The method is applied to three specific examples: one dimensional, hysteretic shear beams, thin plates governed by nonlinear equations of motion and the same nonlinear thin plates but with cutouts. Solutions via the equivalent linearization method using the stress difference minimization compare well with Galerkin's method and numerical integration. The last example is easily handled by the continuum equivalent linearization technique, whereas other methods prove to be inadequate.

iii

TABLE OF CONTENTS

Acknowledgments			ii	
Abstract			iv	
Chapt	Chapter I Introduction			
Chapter II Governing Equations and Review of Some Existing Solution Procedures			6	
2.1	Introduction			
2.2	Governing Equations			
2.3	System Properties			
•	2.3.1	Linearizability of the system	9	
	2.3.2	Positive definiteness and self-adjointness of the linearized system	.10	
	2.3.3	Form of damping operator	12	
	2.3.4	Symmetry of nonlinear stress operators - steady- state harmonic response	12	
2.4	4 Existing Solution Techniques			
	2.4.1	Galerkin's method (periodic solutions)	14	
	2.4.2	Ritz method (periodic solutions)	19	
	2.4.3	Spatial discretization and equivalent linearization	21	
2.5	Remarks	S	24	
Chapte	er III Equ	uvalent Linearizaton for Continuous Systems	26	
3.1	Introduction			
3.2	Class of	Auxiliary Systems	27	
3.3	Definitio	n of Equation Differences	28	
3.4	Averagin	ng Operators	29	
3.5	System :	Difference Minimization	30	
	3.5.1	Formulation I minimization	32	
	3.5.2	Formulation II minimization	34	
	3.5.3	Formulation III minimization	35	
3.6	Examina	ation of the Minimum	37	
3.7	Spatial I	Discretization of the Auxiliary Equations	40	
3.8 Existence and Uniqueness of Equivalent Linear Parameters 42				
Chapter IV Special Cases of the General Approach 46				

.

4.1 Stea	ly-State Harmonic Response	47	
4.	1.1 Formulation I - steady-state harmonic response	48	
4.	1.2 Formulation II - steady-state harmonic response	51	
4.	1.3 Formulation III - steady-state harmonic response	54	
4.2 Stat	onary Random Response	55	
4.	2.1 Formulation I - stationary random response	58	
4.	2.2 Formulation II - stationary random response	59	
4.	2.3 Formulation III - stationary random response	60	
4.3 Discu	ssion of Solution Procedures	62	
4.	3.1 Free vibration response	62	
4.	3.2 Forced response to harmonic excitation	64	
4.	3.3 Forced response to stationary random excitation	66	
Chapter V	Example Studies	67	
5.1 Resp	onse of a One-Dimensional Yielding Continuum	67	
5.	1.1 Minimization relations and discretization of auxi- liary equation	69	
5.	1.2 Specification of the yielding model	72	
5.	1.3 Convergence of equivalent linear parameters	75	
5.	1.4 Numerical results	80	
5.2 Stea	ly-State Response of a Nonlinear Plate	84	
5.	2.1 Equations of motion	84	
5.	2.2 Equivalent linear system	87	
5.	2.3 Finite element analysis of the auxiliary equation	91	
5.	2.4 Convergence considerations for equivalent linear parameters	96	
5.	2.5 Numerical Results	97	
	5.2.5.1 Free vibration response	99	
	5.2.5.2 Forced response	100	
5.	2.6 Discussion	103	
5.3 Stea Hole	ly-State Harmonic Response of a Nonlinear Plate with	106	
5.	3.1 Ritz method	106	
5.	3.2 Fundamental natural frequency	109	
5.	3.3 Free vibration response	110	
5.	3.4 Forced response	116	
5.	3.5 Discussion	118	

Concluding Remarks		120
References	· ·	124

L INTRODUCTION

Linear models of dynamic systems have enjoyed widespread use throughout the history of engineering analysis. Due to the relative simplicity of such models and the amount of attention given to this type of analysis, a vast amount of both qualitative and quantitative information is available on the response of such systems. In fact, the simplicity of linear analysis is often the sole impetus for using a linear model in the first place.

However, the need for nonlinear analysis becomes particularly apparent when the dynamic systems experience large amplitudes of vibration. In particular, it is recognized that large scale civil engineering structures behave nonlinearly when excited by damaging earthquakes, wind loadings and wave loadings. Nonlinear, dynamic models are also needed for the analysis of aircraft and aerospace structural components when excited by strong acoustical loadings. The necessity for nonlinear structural models may arise from geometric considerations, nonlinear elastic and inelastic material behavior and/or from the inadequacy of linear models in describing the energy dissipation in the structure.

The difficulties of nonlinear analysis lie not so much in the lack of good mathematical models, but more in the inability to solve the equations of motion. Only in rare cases can exact solutions to the system equations be obtained, and generally, existing analytical techniques produce only qualitative information on the response. Therefore, recourse to numerical techniques or approximate analysis is generally made. Although purely numerical techniques provide detailed information on the response of specific systems, they tend to be expensive and are not well suited to general studies. Hence, emphasis in this dissertation will be placed on approximate analytical techniques. The inclusion of

purely numerical analysis will be limited to the provision of checks on the analytical solutions.

Considerable attention has been given to the understanding of the response of nonlinear discrete systems. The discrete models originate either from systems where discrete dynamic components are physically apparent or from spatial discretization of a continuous systems. The linearization procedures are then applied to the resulting ordinary differential equations of motion.

A discussion of some commonly used techniques of nonlinear analysis directed toward use in earthquake engineering is presented in reference [20]. For the analysis of systems subjected to deterministic excitation, most of these approximate techniques rely on the assumption of a particular solution form. The solutions are written in terms of undetermined coefficients for which the approximate method produces solutions. The classical methods of harmonic balance, energy balance and equivalent linearization can be used to obtain first-order approximate solutions for steady-state response of the system when excited by a harmonic input. Higher order approximations to such solutions can be obtained through the use of perturbation techniques, asymptotic methods [4] and Galerkin's method [27] by the inclusion of several terms in the approximation. The specification of the solution form in the above methods precludes transient solutions, nor can stability information on steady-state solutions can be obtained. The method of slowly varying parameters [20] does, however, permit such analysis.

A number of these approximate techniques used on systems described by the discrete, deterministic theory have been adapted for use on the nonlinear, stochastic vibration problem. A complete review of these methods can be found in references [20] and [7]. Three of the most widely used methods are the previ-

ously mentioned perturbation and equivalent linearization techniques along with the Fokker-Planck equation approach. The latter approach is one in which the joint probability density functions of the displacements and velocities are obtained, either exactly or through an approximate technique, from the corresponding Fokker-Planck equation.

When a nonlinear continuum model is used to describe the dynamic system, complexities are introduced in that the model now contains a spatial domain as well as the temporal domain present in a discrete model. Existing studies deal with the nonlinear analysis of structural elements such as beams, strings, plates, membranes and shells. Three rather comprehensive surveys of existing work with these nonlinear structural elements can be found in references [41], [42] and [11]. In most studies of these types of nonlinear continua, the partial differential equations of motion are first spatially discretized by an approximate technique such as modal decomposition, Galerkin's method or the finite element method. Then the previously described approximate discrete techniques are directly applicable to the resulting set of nonlinear, ordinary differential equations.

The approach taken in this dissertation for the analysis of nonlinear continua is to perform the linearization *prior to* the spatial discretization of the partial differential equations. In particular, the nonlinear system is replaced by a linear auxiliary system containing undetermined, spatially distributed damping and stiffness parameters. These parameters are then determined in such a way as to model the amplitude dependent energy dissipation and stiffness properties of the nonlinearity in the original system. The solution to the auxiliary system can then be established in terms of the equivalent linear parameters. This is a continuum analog to the discrete equivalent linearization method that has been previously mentioned.

It is recognized that the finite element method is widely used in structural analysis, especially for structural members with complicated boundary conditions and spatially distributed material properties. The motivation for defining the equivalent linear system at the continuum level is for the ease of implementation into finite element computer codes. Analysis presented in this dissertation has been directed toward such an implementation.

In Chapter II, a set of general equations of motion are defined for a linearizable system for which zero-mean solutions exist. A brief survey of existing techniques for solving such a system is presented. For later use, the Galerkin method and a Ritz technique are described in detail. The formulation for equivalent linearization method for discrete systems is presented to provide a basis for the new continuum equivalent linearization approach.

The general formulation for the continuum equivalent linearization method is developed in Chapter III. Three different formulations of the method are presented and discussed. The existence and uniqueness of solutions for the equivalent linear parameters are investigated for the three problem formulations. Considerations of the finite element discretization of the linear, auxiliary system are also presented.

The general equivalent linearization relations from Chapter III are specialized in Chapter IV for two types of analysis: steady-state solutions for harmonic excitation and stationary response to random excitation. Mechanization of the method is discussed for both types of analysis. The free vibration, amplitudefrequency relations can be obtained as a special case of the steady-state, harmonic analysis, and a technique is presented for solving the equivalent linear equations. A method for obtaining resonant response of the harmonically excited system is also introduced.

In Chapter V three example studies of steady-state, harmonic response is presented. In each study, the fundamental nonlinear response mode is investigated using the continuum equivalent linearization method, and results are compared with those from other existing techniques.

The response of a one-dimensional shear beam composed a hysteretic yielding material is investigated in section 5.1. The convergence of the equivalent linear solution with refinement of the mesh for the spatial domain is discussed. Results from the equivalent linearization analysis are compared with those from Galerkin's method.

Section 5.2 contains the nonlinear analysis of Kirchhoff plates described by the Berger formulation of the Von Karman nonlinear equations of motion. The linear auxiliary equation is discretized using the Bogner-Fox-Schmidt shape functions in the finite element formulation. The results from the three equivalent linearization problem formulations are compared with those from Galerkin's method and numerical integration of the equations of motion.

In section 5.3, a further example of the nonlinear plate of section 5.2 is treated: the response of a rectangular plate with *cutouts*. The matching of the additional boundary conditions at the hole makes analysis by standard techniques difficult, while usage of the finite element-equivalent linearization techniques can be as routine as for the plate without a hole. The effect of the hole size on the resonant frequencies and forced response has been determined by the equivalent linearization technique, and results are compared with those found using a Ritz formulation of the nonlinear problem.

IL. GOVERNING EQUATIONS AND REVIEW OF SOME EXISTING SOLUTION TECHNIQUES

2.1 Introduction

The first section of this chapter deals with the definition of the class of nonlinear continuous systems that will be investigated in this and subsequent chapters. The wide variety of vibration problems encountered in engineering practice makes the definition of an all-encompassing system of equations difficult, if not impossible. In spite of this obstacle, an attempt will be made to cover as many nonlinear problems as possible, while retaining a sufficient amount of conciseness to permit some general statements concerning their solutions.

Limitations placed on the system operators, boundary conditions, etc., as presented in section 2.3, should be viewed in the spirit of conciseness and not so much as being necessary conditions to obtain solutions to a specific problem. For example, the familiar property of self-adjointness of the differential operator in a eigenvalue problem is sufficient for the useful property of eigenfunction orthogonality, but some systems which are not self-adjoint can surely be solved.

In section 2.4, a number of existing techniques for solving nonlinear vibration response of continua are discussed. The list of methods is not intended to be all-inclusive; only those techniques that will be explored in later chapters are reviewed. These techniques, typical of all those encountered by the author, are formulated in such a way that the system is first spatially discretized, and the resulting time dependent relations are then linearized. This contrasts with the new technique to be presented in Chapter III in which the linearization is performed in the continuous domain, followed by the spatial discretization.

2.2 Governing Equations

A number of dynamic systems can be described by partial differential equations (PDE's) in which the independent variables are space and time. A subclass of the PDE's are those in which the PDE is approximately linear for small motion. For larger amplitude of motion, the nonlinear effect becomes more pronounced with the degree of nonlinearity depending on the size of subsequent motion. Herein, consideration will be given to a further subclass of equations in which the nonlinearities are independent of inertial stresses and will depend only on dissipative and restoring stresses of the system. Let $\vartheta(\mathbf{x})$ and $\vartheta \vartheta(\mathbf{x})$ represent the spatial domain and boundary of the spatial domain, respectively, described by spatial coordinates \mathbf{x} , and the time parameter be represented by t. Consider a continuous system in $\vartheta(\mathbf{x})$ described by the set of partial differential equations:

$$\nabla \cdot \tau^{L}(\mathbf{a}, \mathbf{w}) + \nabla \cdot \tau^{N}(\mathbf{c}, \mathbf{w}, \dot{\mathbf{w}}) + \dot{\mathbf{Q}}^{L}(\mathbf{b}, \mathbf{w}) + m(\mathbf{x}) \ddot{\mathbf{w}} = \mathbf{p}(\mathbf{x}, t)$$
(2.1)

where a dot above a variable (or operator) denotes differentiation with respect to $t, \nabla \cdot ()$ is the divergence operator with respect to the spatial coordinates x, and

- $\mathbf{w} = \mathbf{w}(\mathbf{x}, t)$ is the dependent variable representing "displacements"
- a, b, c are sets of 'material" and "geometry" properties of the system and are functions only of x
- $m(\mathbf{x})$ is the "mass" distribution of the system
- **p**(**x**, *t*) is the applied 'load''
- τ^L is a linear "stress" operator working only on the displacements
 w and "stiffness" parameters a
- τ^N is an operator (or functional) which is **nonlinear** in displacements and/or velocities, \dot{w} and involves the nonlinear "stiffness"

parameters b

• $\dot{\mathbf{Q}}^L$ is a linear "damping" operator working on the displacements **w** and damping parameters c

On every point on the boundary, $\partial \mathcal{D}(\mathbf{x})$, let either of the following hold:

$$\mathbf{B}^{g}(\mathbf{w}) = \mathbf{0} \tag{2.2}$$

$$\mathbf{B}^{n}(\mathbf{w}) = \mathbf{0} \tag{2.3}$$

where $B^{g}(w)$ and $B^{n}(w)$ are sets of geometric and natural boundary conditions, respectively, which are linear homogeneous operators containing derivatives normal to and along $\partial \mathcal{D}(\mathbf{x})$.

Before continuing the discussion of the nonlinear system of equations, a few remarks need to be made on notation and terminology:

- 1. The **bold** terms in equations (2.1) thru (2.3) represent sets of indexed elements. The number and range of indices depend on the given system under investigation. For example, the linear "stress" operation τ^L may be a scalar τ^L , a vector τ^L_i or a two tensor τ^L_{ij} , depending on the number of indices needed to describe the stress field. The boldface notation was chosen here for the sake of generality.
- 2. The terms enclosed in quotes in the text following equation (2.1), such as "stress", should be thought of as generic descriptions. For different systems, τ^L , for example, may have different physical meanings, but in each case τ^L would have the character of stress. In further discussion, the generic name will be used without being restricted to any physical interpretation.
- 3. In order for the mathematical statement of the nonlinear problem, equations (2.1) (2.3), to be well-posed, appropriate initial conditions

need to be specified for \mathbf{w} and $\mathbf{\dot{w}}$. However, attention will be focused mainly on obtaining steady state solutions when \mathbf{p} is harmonic in time or on stationary solutions for a temporally random \mathbf{p} . For these types of analysis, the periodicity or stationarity conditions will alleviate the need for specifying initial conditions.

2.3 System Properties

The following restrictions will be placed on the nonlinear continuous system described by equations (2.1) - (2.3). These properties will be used to simplify analysis in order that specific statements can be made about the response of the system.

2.3.1 Linearizability of the system

Most existing techniques, along with the technique to be presented in the next chapter, for solving nonlinear systems of the type introduced in the preceding section assume that the system response is nearly linear for small amplitudes of motion. In fact, the response of this linear system is often used for a first a approximation to the nonlinear response. The linearizability has already been partially satisfied by the statement of the problem in equations (2.1); that is, the restoring stress has been written in a linear part, τ^L , and a nonlinear part, τ^N . It only remains that the qualitative "effect" of τ^L be predominant over τ^N as the amplitude of vibration becomes small. The idea of linearizability is dependent on the approximate method used, and therefore detailed discussion of this point will be deferred to later discussion of the specific techniques, whenever possible. In all further discussion, the *linearized system* will be taken to be equations (2.1) with τ^N omitted, or

$$\nabla \cdot \tau^{L} + \mathbf{Q}^{L} + m(\mathbf{x})\mathbf{\ddot{w}} = \mathbf{p}(\mathbf{x}, t)$$
(2.4)

Definition - Inner product

For general variables, or operators, \mathbf{a} , \mathbf{b} and \mathbf{c} , the scalar inner product ()-() is defined to have the following properties:

1. $a \cdot b = b \cdot a$ 2. $a \cdot a > 0$ if $a \neq 0$ and $a \cdot a = 0$ iff a = 03. $(a + b) \cdot c = a \cdot c + b \cdot c$

Definition - Positive Definite

For a differential operator L(a, u) and for a and u defined on the spatial domain $\vartheta(\mathbf{x})$, L is said to be positive definite if

$$\int_{\Omega} \mathbf{u} \cdot \mathbf{L}(\alpha, \mathbf{u}) d \quad \mathfrak{g}(\mathbf{x}) > \mathbf{0}$$

and for $\alpha \neq 0$, the integral is zero for u = 0. The integral can vanish for $u \neq 0$ iff $\alpha = 0$.

Definition - Self Adjoint

For L and u described above and for v also defined on $\mathcal{D}(\mathbf{x})$, L is said to be self-adjoint if

$$\int_{\mathcal{D}} \mathbf{u} \cdot \mathbf{L}(\mathbf{v}) d \quad \mathcal{D}(\mathbf{x}) = \int_{\mathcal{D}} \mathbf{v} \cdot \mathbf{L}(\mathbf{u}) d \quad \mathcal{D}(\mathbf{x})$$

It should be noted that whether or not an operator is positive definite and selfadjoint can be established by integration by parts.

According to the definitions presented above, the divergence of the linear stress operator, $\nabla \cdot \tau^L$, will be required to be both positive definite and self

adjoint in all further discussion of the nonlinear system (2.1). In addition, the mass distribution $m(\mathbf{x})$ will be required to satisfy

$$m(\mathbf{x}) > 0 \tag{2.5}$$

Consider the eigenvalue problem of the undamped, linearized system in equation (2.4):

$$\nabla \cdot \tau^{L}(\mathbf{u}_{r}) = \lambda_{r} m(\mathbf{x}) \mathbf{u}_{r}$$
(2.6)

where λ_r and \mathbf{u}_r are the r^{th} eigenvalue and r^{th} set of eigenfunctions. As can be found in a number of vibrations texts, for example reference [34], the above properties for $\nabla \cdot \tau^L$ and $m(\mathbf{x})$ permit the following relationships for the eigenvalues of (2.6):

1. The eigenvalues are positive.

$$\lambda_r > 0 \quad ; r = 1, 2, \dots$$
 (2.7)

2. For distinct eigenvalues λ_r and λ_s , \mathbf{u}_r and \mathbf{u}_s are orthogonal and can be normalized in such a way that

$$\int_{\mathcal{D}} m(\mathbf{x}) \, \mathbf{u}_r \cdot \mathbf{u}_s \, d \, \mathcal{L}(\mathbf{x}) = \begin{cases} 1 & \text{if } r = s \\ 0 & \text{if } r \neq s \end{cases}$$
(2.8)
for r,s = 1,2,...

 A Galerkin discretization of the eigenvalue problem in equation (2.6) will produce symmetric, positive definite "mass" and "stiffness" matrices. 2.3.3 Form of damping operator

The damping operator $\dot{\mathbf{Q}}^{L}(\mathbf{w})$ will be assumed to be from the class of Rayleigh damping operations, where \mathbf{Q}^{L} can be written as a linear combination of $\nabla \cdot \boldsymbol{\tau}^{L}$ and $m(\mathbf{x})\mathbf{w}$, or

$$\mathbf{Q}^{L} = \gamma_1 \nabla \cdot \tau^{L} + \gamma_2 m(\mathbf{x}) \mathbf{w}$$
(2.9)

with γ_1 and γ_2 being constant coefficients.

Again referring to the eigenfunctions \mathbf{u}_r and eigenvalues λ_r of the linearized equation (2.6) and using properties 1 and 2 from section 2.3.2, the following orthogonality relations can be established for the Rayleigh damping operator of equation (2.9):

$$\int \mathbf{u}_r \cdot \mathbf{Q}^L(\mathbf{u}_s) \ d \quad (\mathbf{x}) = \begin{cases} \lambda_r \gamma_1 + \gamma_2 & \text{if } r = s \\ 0 & \text{if } r \neq s \end{cases}$$
(2.10)

provided $\lambda_r \neq \lambda_s$ for $r \neq s$; r, s = 1, 2, ... Furthermore, since \mathbf{Q}^L is a linear combination of $\nabla \cdot \tau^L$ and $m(\mathbf{x})\mathbf{w}$, a Galerkin discretization of the linearized equation will produce a symmetric, positive definite damping matrix.

2.3.4 Symmetry of nonlinear stress operator - steady-state harmonic response

Definition - Symmetric Operator

If $\mathbf{u}(\mathbf{x}, t)$ is a periodic function of period T in time, then the operation $\tau^{N}(\mathbf{u})$ is said to be symmetric if

$$\int_{0}^{T} \tau^{N}(\mathbf{u}) dt = \mathbf{0}$$
(2.11)

Consider the case of the nonlinear system of equations (2.1) - (2.3) where the right hand side of equation (2.1) is harmonic in time. Then,

$$\nabla \cdot \tau^{L} + \nabla \cdot \tau^{N} + \dot{\mathbf{Q}}^{L} + m(\mathbf{x}) \ddot{\mathbf{w}} = \mathbf{r}(\mathbf{x}) \cos \omega t$$
(2.12)

$$B^g(w) = 0 \tag{2.12a}$$

$$\mathbf{B}^{n}(\mathbf{w}) = \mathbf{0} \tag{2.12b}$$

If a steady-state solution of (2.12) exists, let the solution, w, be sought which is periodic of period T, where $T = \frac{2\pi}{\omega}$.

Since $w(\mathbf{x}, t)$ is periodic in time, it can be expanded in a temporal Fourier series [52]

$$\mathbf{w}(\mathbf{x},t) = \frac{1}{2}\mathbf{b}_0 + \sum_{n=1}^{\infty} \left[\mathbf{a}_n(\mathbf{x})\sin\frac{2n\pi}{T}t + \mathbf{b}_n(\mathbf{x})\cos\frac{2n\pi}{T}t\right]$$
(2.13)

where $\mathbf{a}_n(\mathbf{x})$ and $\mathbf{b}_n(\mathbf{x})$; n=1, 2, ..., are the Fourier coefficients of w. Then, if τ^N is symmetric and $\nabla \cdot \tau^L$ is positive definite, it can be shown that $\mathbf{b}_0(\mathbf{x}) = \mathbf{0}$.

In all further discussion of the steady-state solution of equation (2.12), it will be assumed that the nonlinearity is symmetric. The significance of this, as shown above, is that it will not be necessary to include a constant (in time) offset function in the approximate solution for the steady-state, periodic solutions of the nonlinear system. The inclusion of the offset function will generally involve more complexities in the analysis, as can be seen in the treatment of non-symmetric nonlinearities of discrete systems in reference [45].

2.4 Review of Existing Solution Techniques

Exact solutions to the class of nonlinear systems described in equations (2.1) - (2.3) are generally not available. Therefore, recourse to approximate techniques is usually made. Existing techniques fall into three general classifications: those which are purely numerical, those which are purely analytical and those which are both analytical and numerical in character. Methods which are purely numerical are generally quite expensive, and generalities to solutions of the same system using different parameters are lost. Therefore, further discussion will be focused only on purely analytical and numericalanalytical methods, except when purely numerical techniques are desired to serve as a check on other solution procedures.

A different technique is generally used to solve the temporal portion of the problem than is used to solve the spatial part. The method chosen for each part of the problem will depend on the type of analysis to be performed (stability, steady-state response, etc.), on the boundary conditions to be satisfied or on simply a personal preference of the analyst. Table 2.1 lists a number of publications that deal with the nonlinear vibration of continua by purely analytical or numerical-analytical methods. An attempt has been made in the table to include methods representative of current and past work, but it is by no means inclusive of all work.

In the following section, three methods will be discussed in detail: Galerkin's method applied to both spatial and temporal problems, a Ritz approximation in space and Hamilton's principle in time, and spatial discretization followed by the method of equivalent linearization on the temporal problem.

2.4.1 Galerkin's method (periodic solutions)

Consider the nonlinear system given by equation (2.12) which is forced by a load that is harmonic in time with period $T = \frac{2\pi}{\omega}$. Let a set of functions $\Phi_i(\mathbf{x})$; i=1, 2, ..., N, be comparison functions, where a comparison function has the following properties:

- 1. $\Phi_i(\mathbf{x})$; i=1, 2, ..., N, satisfy all the boundary conditions of (2.12); i.e. $B^g(\Phi_i) = 0$ and $B^n(\Phi_i) = 0$
- 2. $\Phi_i(\mathbf{x})$; i=1, 2, ..., N, are linearly independent and represent the first N of a group of functions $\Phi_i(\mathbf{x})$ i=1, 2, ..., N, ... which is complete in the

Type of System	Type of Analysis	Spatial Method	Temporal Method	Ref.
plana r string	forced random	modal analysis	equivalent linearlization	[10]
planar string non autonomous	forced periodic	modal analysis	perturbation	[33]
1-D rod hysteretic	traveling waves	perturbation	harmonic balance	[9]
non-planar bending beam	stability of forced, periodic	Galerkin	perturbation	[17]
bending beam	sub-harmonics forced, periodic	modal analysis	Galerkin	[50]
bending beam	free vibration	modal analysis	exact	[53]
bending beam	stability of forced, periodic	finite elements	perturbation	[6]
plate	free vibration	Ritz- Kantrorovitch	finite difference	[18]
plate	forced periodic	Lagrange's equations	equivalent linearization	[31]
plate	forced periodic	exact	multiple scales	[46]
plate on elastic foundation	free vibration	Galerkin	exact	[14]
plate with initial stress	forced periodíc	Galerkin	Galerkin	[12]
plate	free vibration	perturbation	Hamilton's principle	[40]
cylindrical shell	forced periodic	Lagrange's equations	harmonic balance	[15]
2-D plane strain hysteretic	forced random	finite elements	equivalent linearization	[43]

Table 2.1 Summary of Existing Work on the Nonlinear Vibration of Continua given region.

Furthermore, since solutions are sought which are periodic with period $\frac{2\pi}{\omega}$, let $f_i(t)$ be a set of functions which are periodic with a period $\frac{2\pi}{\omega}$. Since the set of functions [$\sin\omega t$, $\cos\omega t$] form a basis for such periodic functions, let

$$f_i(t) = u_i \cos\omega t + v_i \sin\omega t \tag{2.14}$$

Using these properties of Φ_i and f_i , the solution $\mathbf{w}(\mathbf{x}, t)$ will be approximated by $\widehat{\mathbf{w}}(\mathbf{x}, t)$, where

$$\widehat{\mathbf{w}}(\mathbf{x},t) = \sum_{i=1}^{N} \Phi_{i}(\mathbf{x}) f_{i}(t)$$
(2.15)

Since $\widehat{\mathbf{w}}$ is not an exact solution, substitution of $\widehat{\mathbf{w}}$ into the equations of motion leaves a non-vanishing residual, ε , where:

$$\boldsymbol{\varepsilon} = \nabla \cdot \boldsymbol{\tau}^{L}(\widehat{\mathbf{w}}) + \nabla \cdot \boldsymbol{\tau}^{N}(\widehat{\mathbf{w}}, \widehat{\mathbf{w}}) + \widehat{\mathbf{Q}}^{L} + m(\mathbf{x})\widehat{\mathbf{w}} - \mathbf{r}(\mathbf{x})\cos\omega t \qquad (2.16)$$

With $\widehat{\mathbf{w}}$ defined as in equation (2.19), Galerkin's method says that the best solution for u_i and v_i ; i=1, 2, ..., N, is given by the solution to the following set of equations (see reference [27]):

$$\int_{0}^{T} \int_{\mathcal{D}} \varepsilon \cdot \frac{\partial \widehat{\mathbf{w}}}{\partial u_{k}} d\mathbf{x} dt = 0$$

$$T \int_{0}^{T} \int_{\mathcal{D}} \varepsilon \cdot \frac{\partial \widehat{\mathbf{w}}}{\partial v_{k}} d\mathbf{x} dt = 0$$
(2.17)

for k = 1, 2, ..., N.

Inserting equations (2.15) and (2.16) into equations (2.17) and performing the integrations gives:

$$\sum_{j=1}^{N} \left[-\omega^2 M_{ij}^G + K_{ij}^G \right] u_j + \omega \sum_{j=1}^{n} B_{ij}^G v_j + \mathfrak{F}_{Ci}^G = f_i^G$$
(2.18)

$$-\omega \sum_{j=1}^{N} B_{ij}^{G} u_{j} + \sum_{j=1}^{N} \left[-\omega^{2} M_{ij}^{G} + K_{ij}^{G} \right] v_{j} + \Im_{Si}^{G} = 0$$

where

$$M_{ij}^{\mathcal{C}} = \int m(\mathbf{x}) \Phi_i \Phi_j d \mathcal{D}(\mathbf{x})$$

$$\mathcal{O}$$
(2.18a)

$$B_{ij}^{\mathcal{G}} = \int_{\mathfrak{Q}} \dot{\mathbf{Q}}^{L}(\Phi_{j}) \cdot \Phi_{i} \ d \ \mathfrak{Q}(\mathbf{x})$$
(2.18b)

$$K_{ij}^{G} = \int_{\mathcal{D}} [\nabla \cdot \tau^{L}(\Phi_{j})] \cdot \Phi_{i} \ d \ \mathcal{D}(\mathbf{x})$$
(2.18c)

$$\Im \mathcal{E}_{i} = \int [\nabla \cdot \mathbf{C}(\mathbf{u}, \mathbf{v})] \cdot \Phi_{i} d \otimes (\mathbf{x})$$
(2.18d)

$$\mathcal{F}_{S_i}^{\mathcal{G}} = \int_{\mathcal{Q}} [\nabla \cdot \mathbf{S}(\mathbf{u}, \mathbf{v})] \cdot \Phi_i \ d \ \mathcal{Q}(\mathbf{x})$$
(2.18e)

$$\mathbf{C}(\mathbf{u},\mathbf{v}) = \frac{1}{\pi} \int_{0}^{2\pi} \tau^{N}(\mathbf{u},\mathbf{v},\vartheta) \cos\vartheta \,d\vartheta \qquad (2.18f)$$

$$\mathbf{S}(\mathbf{u},\mathbf{v}) = \frac{1}{\pi} \int_{0}^{2\pi} \tau^{N}(\mathbf{u},\mathbf{v},\vartheta) \sin\vartheta \, d\,\vartheta$$
(2.18g)

$$f_i^G = \int \mathbf{r}(\mathbf{x}) \cdot \Phi_i \ d \ \mathcal{D}(\mathbf{x})$$
(2.18h)

$$\vartheta = \omega t$$
 (2.18i)

Recall the properties of the nonlinear system given in section (2.3), particularly those of positive definiteness, self-adjointness and Rayleigh damping. Now suppose that Φ_k ; k=1, 2, ..., N, are the first N sets of eigenfunctions to the linearized problem, or

$$\nabla \cdot \tau^{L}(\Phi_{k}) = \lambda_{k}^{2} m(\mathbf{x}) \Phi_{k}$$
(2.19)

where λ_k^2 represents the corresponding k^{th} eigenvalue. Using (2.19), the orthogonality property of equation (2.8), and the Rayleigh damping definiton, the matrices of equations (2.18a) - (2.18c) become

$$M_{ii}^G = 1$$

$$B_{ii}^{G} = 2\xi_{i}\lambda_{i}$$

$$K_{ii}^{G} = \lambda_{i}^{2}$$

$$M_{ij}^{G} = B_{ij}^{G} = K_{ij}^{G} = 0 \quad i \neq j$$

where ζ_i is the damping coefficient of the *i*th linear mode and $\zeta_i = \frac{1}{2} \left(\frac{1}{\lambda_i} \gamma_1 + \lambda_i \gamma_2 \right).$

All future references to the Galerkin solution for the periodic nonlinear problem will be solutions to the nonlinear algebraic equations of (2.18), (2.18d) -(2.18h) and (2.20). Note that by using the linear eigenfunctions for comparison functions, the linear portion is *uncoupled*, thus leaving only the nonlinear vectors \Im_{c}^{c} and \Im_{s}^{c} to couple the equations (2.18).

(2.20)

2.4.2 Ritz method (periodic solutions)

The Ritz method will proceed from the variational form of the nonlinear problem, whereas the preceding Galerkin method started with the system being described by a differential equation. Let the generalized Lagrangian, L, of the system (2.12) be written as:

$$L = T - U + W \tag{2.21}$$

where T is the kinetic energy, U the potential energy and W the virtual work done on the system by non-conservative forces. The generalized Hamilton's principle [16] says that from time t_1 to time t_2 , a point in the system will follow a path which extremizes the integral of L over time from t_1 to t_2 , or

$$\delta \int_{t_1}^{t_2} L \, dt = 0 \tag{2.22}$$

In the Ritz method, the solution will be approximated by a function

$$\widetilde{\mathbf{w}}(\mathbf{x},t) = \sum_{i=1}^{N} \Phi_{i}(\mathbf{x}) [u_{i} \cos\omega t + v_{i} \sin\omega t]$$
(2.23)

where here, Φ_i ; i=1, 2, ..., N, are admissable functions. Admissable functions are similar to comparison functions except that they need only satisfy the geometric boundary conditions; that is, for the system of equations (2.12), only the following need be satisfied:

$$B^{g}(\Phi_{i}) = 0$$
; $i=1, 2, ..., N$

Surely trial functions which satisfy all of the boundary conditons could be used, but the use of admissable functions enlarges the class of functions from which to choose. In fact, the satisfaction of geometric boundary conditions is not necessarily required to effect a solution when dealing with the variational form of the problem. Discussion of this can be found in references [48] and [34].

The variation in equation (2.22) is accomplished by setting

$$\frac{\partial}{\partial u_k} \int_0^{2\frac{\pi}{\omega}} (T - U + W) dt = 0$$

$$\frac{\partial}{\partial v_k} \int_0^{2\frac{\pi}{\omega}} (T - U + W) dt = 0$$

$$(2.24)$$

The kinetic energy for the system in equations (2.12) is given by

$$T(\widetilde{\mathbf{w}}) = \frac{1}{2} \int m(\mathbf{x}) \cdot \widetilde{\mathbf{w}} \, d \, \mathcal{D}(\mathbf{x}) \tag{2.25}$$

Suppose that the potential energy U can be written as:

$$U = U^{L} + U^{NL}$$
$$= \frac{1}{2} \int \tilde{\tau}^{L} \tilde{\epsilon}^{L} d \quad \vartheta(\mathbf{x}) + U^{NL}$$
(2.26)

where U^L is the potential energy in the linearized problem, $\tilde{\tau}^L$ and $\tilde{\epsilon}^L$ are linear operations of w and U^{NL} is the contribution to the potential energy by τ^N .

Using the linearity of $\tilde{\tau}^L$ and $\tilde{\epsilon}^L$, along with equations (2.23), (2.25) and (2.26), the Ritz equations can be written as:

$$\sum_{j=1}^{N} \left[-\omega^2 \, \mathcal{M}_{ij}^R + \, \mathfrak{X}_{ij}^R \right] u_j = \, \mathfrak{F}_{Ci}^R \tag{2.27}$$

$$\sum_{j=1}^{N} \left[-\omega^2 \, \mathcal{M}_{ij}^{R} + \, \mathcal{K}_{ij}^{R} \right] v_j = \mathcal{F}_{Si}^{R}$$

where

$$\mathcal{M}_{ij}^{R} = \int_{\mathcal{Q}} m(\mathbf{x}) \Phi_{i} \Phi_{j} d \mathcal{Q}(\mathbf{x})$$
(2.27a)

$$\mathfrak{X}_{ij}^{R} = \frac{1}{2} \int_{\mathfrak{Q}} \left[\widetilde{\tau}^{L}(\Phi_{i}) \cdot \widetilde{\varepsilon}^{L}(\Phi_{j}) + \widetilde{\tau}^{L}(\Phi_{j}) \cdot \widetilde{\varepsilon}^{L}(\Phi_{i}) \right] d \mathfrak{Q} (\mathbf{x})$$
(2.27b)

$$\mathfrak{F}_{Ci}^{R} = \frac{1}{\pi} \frac{\partial}{\partial u_{i}} \int_{0}^{2\pi} [W(\widetilde{\mathbf{w}}) - U^{NL}(\widetilde{\mathbf{w}})] d\vartheta$$
(2.27c)

$$\mathcal{F}_{Si}^{R} = \frac{1}{\pi} \frac{\partial}{\partial v_{i}} \int_{0}^{2\pi} [\mathcal{W}(\widetilde{\mathbf{w}}) - U^{NL}(\widetilde{\mathbf{w}})] d\vartheta$$
(2.27d)

$$\vartheta = \omega t \tag{2.27e}$$

If the system investigated is conservative and linear, that is if W and U^{NL} are omitted from (2.27c) and (2.27d), the system of equations reduce to the eigenvalue problem:

$$\sum_{j=1}^{N} \left[-\omega^2 \, \mathcal{M}_{ij}^R + \, \varkappa \, _{ij}^R \right] \, u_j = 0 \tag{2.28}$$

It can be shown that the matrices \mathcal{M}^R and \mathcal{K}^R in (2.27a) and (2.27b) are identical to those one would obtain by directly applying the well-known Rayleigh-Ritz method to the linear, conservative system. Therefore, this method of performing a Ritz approximation to Hamilton's principle can be shown equivalent to a well established method for linear problems.

2.4.3 Spatial discretization and equivalent linearization

Let attention be returned to the general nonlinear system of equations given by (2.1) - (2.3). The first step of this method is to discretize the spatial portion by any acceptable technique, such as via the finite element method, Galerkin's method, or modal expansion in terms of the linearized eigenfunctions. This will generally lead to a set of ordinary differential equations in time of the form:

$$M\ddot{\mathbf{d}} + C(\mathbf{b})\dot{\mathbf{d}} + K(\mathbf{a})\mathbf{d} + \Im(\mathbf{c}, \mathbf{d}, \dot{\mathbf{d}}) = \mathbf{f}(t)$$
(2.29)

where $\mathbf{d} = \mathbf{d}(t)$ are discretized N-vector displacements. *M*, *C* and *K* are the mass, damping and stiffness matrices of size NxN, resulting from the linear operators in (2.1), while $\mathbf{f}(t)$ represents the discretized forcing term. \Im (**b**, **d**, **d**) is an Nvector which is nonlinear in displacements **d** and velocities **d** and originates from the discretization of the nonlinear operator $\nabla \cdot \tau^N$. Recall that **a**, **b** and **c** are stiffness, damping and nonlinearity parameters, respectively, in the original nonlinear system. The explicit form of the matrices and vectors in (2.29) will depend on the method of discretization.

By way of obtaining appproximate solutions to equations (2.29), let an auxiliary set of *linear* equations be given by:

$$M\mathbf{d} + (C+C^{e})\mathbf{d} + (K+K^{e})\mathbf{d} = \mathbf{f}(t)$$
 (2.30)

where C^{e} and K^{e} are matrices independent of time. As will be seen later in this section, these matrices will be adjusted in a prescribed manner such that the difference between the nonlinear equations of (2.29) and the auxiliary, or *equivalent linear*, equations in (2.30) will be minimized.

Let \mathbf{d}^{\bullet} be a vector function of time and a member of a class of functions containing the solutions to the linear, auxiliary equations (2.30). Let Δ represent the difference between the original nonlinear system and the linear auxiliary system both operating on the same function $\mathbf{d}^{\bullet}(t)$, or

$$\Delta = \Im (\mathbf{d}^*, \mathbf{\dot{d}}^*) - C^e \mathbf{\dot{d}}^* - K^e \mathbf{d}^*$$
(2.31)

The central idea of equivalent linearization, as applied to equations (2.29), is to minimize the difference term Δ over all functions \mathbf{d}^* belonging to C. The usual procedure is to first average over time the Euclidean norm of Δ . Then the minimization of Δ is stated as

$$G_{\bullet}(\Delta^{T}\Delta) = minimum \quad \text{for all } \mathbf{d}^{\bullet} \in \mathbb{C}$$
(2.32)

where Δ^T denotes the transpose of Δ and G is a linear, time invariant operation.

The minimization is then accomplished by extremizing $G(\Delta^T \Delta)$ with respect to all elements of C^e and K^e . If C_{ij}^e and K_{ij}^e ; i,j=1, 2, ..., N, represent the elements of C^e and K^e , the extremization is expressed as:

$$\frac{\partial}{\partial C_{ij}^{e}} G\left(\Delta^{T}\Delta\right) = 0 \tag{2.33}$$

$$\frac{\partial}{\partial K_{ij}^{\varrho}} \quad (\Delta^T \Delta) = 0 \tag{2.34}$$

Since $d^{\bullet}(t)$ belongs to the class of functions which satisfy the linear auxiliary equations, there exists an implicit relationship between the solution **d** of auxiliary equations (2.30) and the equivalent linear matrices C^{e} and K^{e} in equations (2.33) and (2.34). These equations are then solved in an iterative fashion for \mathbf{d}, C^{e} and K^{e} .

For the sake of brevity, the above description of the generalized equivalent linearization method for ordinary differential equations is by no means complete. A more thorough treatment of the method can be found in references [45] and [21]. One result, from the investigation reported in [45] on the invertibility of relations (2.33) and (2.34), deserves comment at this point. It was shown that when the dimension of the solution space described by equaled or was greater than 2N, the relations in (2.33) and (2.34) became singular. Therefore, if solutions for C^e and K^e exist, they are not unique. However, it was demonstrated that the non-uniqueness does not affect the quality of the solutions in that all values of C^e and K^e which satisfy the singular relationships do an equally good job of minimizing $G(\Delta^T \Delta)$. It should be noted, however, that in the mechanization of the general method, one encounters the task of solving a singular set of equations. This singularity of equations (2.33) and (2.34) will create numerical difficulties. The ideas of existence and uniqueness of the minimization relationships resulting from the extension of equivalent linearization to partial differential equations will be investigated in Chapter III.

2.5 Remarks

The class of nonlinear partial differential equations that will be investigated has been presented in equations (2.1) - (2.3). Simplifying assumptions of linearizability, positive definiteness, self-adjointness, form of damping to be considered, and symmetry of the nonlinearity have been stated and discussed.

A brief survey of existing techniques for solving nonlinear continua problems that are either purely analytical or numerical-analytical has been presented. Galerkin's method, the Ritz approximation to Hamilton's principle and the method of equivalent linearization applied to the spatially discretized problem have been discussed in further detail. These methods reduce the original nonlinear partial differential equations to a set of of nonlinear *algebraic* equations for which general solution techniques exist. The algebraic equations developed via Galerkin's method (2.18) and the Ritz method (2.27) will be used to obtain solutions to particular problems for comparison with solutions generated by the new equivalent linearization technique to be developed in Chapter III. It should be noted that the striking difference between the Galerkin and Ritz method is that the Ritz method is applied to the *variational* form of the system, while the Galerkin method starts with the problem stated in the *differential* form. The variational statement of the problem permits the usage of admissable functions instead of the comparison functions that are needed for Galerkin's method.

This difference gives the edge to the Ritz method on problems in which satisfying the natural boundary conditions becomes difficult, if not impossible.

III. EQUIVALENT LINEARIZATION FOR CONTINUOUS SYSTEMS

3.1 Introduction

The concept of equivalent linearization was introduced in Chapter II in the application of the method to a set of nonlinear ordinary differential equations. The principle of the method was first presented in reference [4] for the case of a single ordinary differential equation with harmonic excitation. Although this original work derived the equivalent linear relationships through the methods of power balance and harmonic balance, the relationships are now typically developed by the residual minimization procedure discussed in Chapter II. The method was then extended to stochastic equations in references [5] and [8], and a generalization to multi-degree-of-freedom systems has been presented in reference [21]. For the most part, equivalent linearization has been used exclusively for determining periodic solutions and for stationary random solutions. One exception to this is recent work in reference [45] in which the method was applied to transient response of nonlinear single-degree-of-freedom systems.

In this chapter, the idea of equivalent linearization is be extended to the class of nonlinear systems by equations (2.1) - (2.3). This method differs significantly from the discretization-equivalent linearization technique described in section 2.4.3, in that here, the linearization is performed before the spatial discretization. More specifically, the auxiliary system for this method is a set of partial differential equations containing dissipative and stiffness parameters to be determined in such a way that a difference between the original nonlinear system and the linear auxiliary is minimized. The difference term here is not only a function of time but also a function of space. Therefore, its minimization involves complexities not encountered in its discrete analog. This chapter deals with the mechanics of constructing such an equivalent linear solution and the properties of their solutions. The first sections are concerned with the definitions of the linear auxiliary system, system difference terms and the averaging operators to be used on the difference terms. Following this in section 3.5, the minimization of the equation difference terms is discussed. Discretization of the resulting equivalent linear system and properties of the equivalent linear solutions are discussed in sections 3.6 and 3.7.

3.2 Class of Auxiliary Systems

Let the linear auxiliary system for the equivalent linearizaton method be defined as:

$$\nabla \cdot \tau^{L}(\mathbf{a}, \mathbf{w}) + \nabla \cdot \tau^{L}(\alpha, \mathbf{w}) + \tilde{\mathbf{Q}}^{L}(\mathbf{b}, \mathbf{w}) + \tilde{\mathbf{Q}}^{L}(\boldsymbol{\beta}, \mathbf{w}) + m(\mathbf{x}) \mathbf{\ddot{w}} = \mathbf{p}(\mathbf{x}, t)$$
(3.1)

$$B^{g}(\mathbf{w}) = \mathbf{0}$$
(3.2)

$$B^{n}(\mathbf{w}) = \mathbf{0}$$

where $\nabla \cdot \tau^{L}(\mathbf{a}, \mathbf{w})$, $\dot{\mathbf{Q}}^{L}(\mathbf{b}, \mathbf{w})$, $m(\mathbf{x})$, p, B^g and Bⁿ are as defined in section 2.2. $\alpha(\mathbf{x})$ and $\beta(\mathbf{x})$ are the equivalent linear stiffness and damping parameters, respectively, for the auxiliary system.

The form of the equivalent linear damping $\dot{\mathbf{Q}}^{L}(\boldsymbol{\beta}, \mathbf{w})$ is more restrictive than for $\dot{\mathbf{Q}}^{L}(\mathbf{b}, \mathbf{w})$. $\dot{\mathbf{Q}}^{L}(\boldsymbol{\beta}, \mathbf{w})$ is limited to strain related damping; that is, in the Rayleigh damping definition of equation (2.9), $\gamma_{1} = 0$ or

$$\mathbf{Q}^{L}(\boldsymbol{\beta}, \boldsymbol{w}) = \nabla \cdot \boldsymbol{\tau}^{L}(\boldsymbol{\beta}, \boldsymbol{w}) \tag{3.3}$$

It is clear that the form of the linear auxiliary system arises from the replacement of the nonlinear operation $\nabla \cdot \tau^N$ in the nonlinear equation (2.1) by the linear operation $\nabla \cdot \tau^L(\alpha, \mathbf{w}) + \dot{\mathbf{Q}}^L(\beta, \mathbf{w})$. Then if equation (3.1) is to model the response of equation (2.1), $\nabla \cdot \tau^L$ and $\dot{\mathbf{Q}}^L$ must model the response dependent

stiffness and dissipation properties of $\nabla \cdot \tau^N$ in some prescribed way.

3.3 Definition of Equation Differences

Let $\mathbf{w}^* = \mathbf{w}^*(\mathbf{x}, t)$ be a set of functions belonging to a class of functions C to which solutions of (3.1) belong. The difference terms defined in this section are differences between given nonlinear operations of the system (2.1) and given linear operations of the auxiliary (3.1), both operating on the function \mathbf{w}^* . Three difference terms, constituting what are referred to as problem formulations I, II and III, are defined as the following.

Definition - Differential Stress Difference (Formulation I)

Since $\mathbf{w}^{\bullet}(\mathbf{x}, t)$ is not necessarily a solution of either the nonlinear system or of the auxiliary system, substitution of \mathbf{w}^{\bullet} into equations (2.1) and (3.1) will leave non-vanishing residual terms. The difference of these residual terms, Δ^{I} , is written as

$$\Delta^{I} \equiv \nabla \cdot \tau^{N}(\mathbf{c}, \mathbf{w}^{*}, \dot{\mathbf{w}}^{*}) - \nabla \cdot \tau^{L}(\alpha, \mathbf{w}^{*}) - \nabla \cdot \dot{\tau}^{L}(\beta, \mathbf{w}^{*})$$
(3.4)

Definition - Energy Difference (Formulation II)

Let $g(w^{*})$ be a linear spatial operation on w^{*} such that for $w^{*} \neq 0$

$$\int \tau^{L}(\alpha, \mathbf{w}^{\bullet}) \cdot g(\mathbf{w}^{\bullet}) d \quad (\mathbf{x}) = 0 \qquad \text{iff } \alpha = \mathbf{0}$$

Then the difference term for formulation II, Δ^{II} , is defined as:

$$\Delta^{II} \equiv \left[\tau^{N}(\mathbf{c}, \mathbf{w}^{\bullet}, \dot{\mathbf{w}}^{\bullet}) - \tau^{L}(\alpha, \mathbf{w}^{\bullet}) - \dot{\tau}^{L}(\beta, \mathbf{w}^{\bullet})\right] \cdot \mathbf{g}(\mathbf{w}^{\bullet})$$
(3.5)

where the () \cdot () scalar inner product is a defined in section 2.2.

Definition - Stress Difference (Formulation III)

For formulation III, let the system difference term, Δ^{III} , be defined as :
$$\Delta^{III} \equiv \tau^{N}(\mathbf{c}, \mathbf{w}^{\bullet}, \dot{\mathbf{w}}^{\bullet}) - \tau^{L}(\alpha, \mathbf{w}^{\bullet}) - \dot{\tau}^{L}(\beta, \dot{\mathbf{w}}^{\bullet})$$
(3.6)

For the sake of clarity, a few observations should be made on the preceding definitons.

- 1. Δ^{I} for formulation I is directly analogous to the equation difference term for ordinary differential equations (2.31).
- 2. If $\$(w^{\bullet})$ is interpreted as the strain operator and if the system is elastic, then Δ^{II} would represent the difference in the strain energy density of τ^N and τ^L operating on the same function w^{\bullet} . Although inelastic systems are not to be excluded from the analysis, Δ^{II} will still be referred to as an "energy" difference.
- 3. Note that Δ^{I} and Δ^{II} are functionally related to Δ^{III} . Δ^{I} is simply the divergence of the stress difference term Δ^{III} , while Δ^{II} is equivalent to $\Delta^{III} \cdot S$.

3.4 Averaging Operators

In establishing an equivalent linear solution, one is left with the task of minimizing a given difference, Δ , between the nonlinear system and the equivalent linear system. Minimizing on a point by point basis will generally engender too many constraints on the solution to make a solution possible. One alternative is to minimize some measure of the difference: for example, its averaged value over time and space. That precisely is the approach taken here.

For a given difference term, Δ , let its scalar inner product, as defined in section 2.3.2, be given by $\Delta \cdot \Delta$. Then the averaged value of $\Delta \cdot \Delta$ will be given by $G_{tx}(\Delta \cdot \Delta)$, with $G_{tx}(\)$ defined as below

Definition - Averaging Operator

For a given scalar function, $u = u(\mathbf{x}, t)$, let $G_{tx}(u)$ be defined as follows:

$$G_{ix}(u) = G_i(G_x(u))$$

$$= G_x(G_i(u))$$
(3.7)

where

$$G_{\mathbf{x}}(\mathbf{u}) = \int \mathbf{u} \, d \, \vartheta(\mathbf{x}) \tag{3.8}$$

and $G_t(u)$ has the following properties

- 1. Time invariance: $\frac{d}{dt} G_t(u)$
- 2. Linearity: $G_t(u+v) = G_t(u) + G_t(v)$
- 3. Positive definiteness: $G_t(u^2) > 0$ for $u \neq 0$ and $G_t(0) = 0$

3.5 System Difference Minimization

In order to determine the equivalent linear stiffness and damping parameters, $\alpha(\mathbf{x})$ and $\beta(\mathbf{x})$, it is necessary to minimize the averaged system difference between systems (2.1) and (3.1). Let Δ represent one of the system differences defined in section 3.3 and $G_{tx}(\Delta \cdot \Delta)$ be the averaging operation defined in section 3.4. Formally, this minimization can be stated as the extremization of $G_{tx}(\Delta \cdot \Delta)$ over all functions $\alpha(\mathbf{x})$ and $\beta(\mathbf{x})$, or

$$\delta_{tx}(\Delta(\alpha, \beta, \mathbf{w}^{\bullet}) \cdot \Delta(\alpha, \beta, \mathbf{w}^{\bullet})) = 0 \quad \text{for all } \mathbf{w}^{\bullet} \in \mathbb{C}$$
(3.9)

One approach in effecting the extremization in equation (3.9) would be the use of direct analytical variation via calculus of variation. For example, consider the system difference defined for formulation III. Using the property of the inner product operator () \cdot () given in section 2.3.2 and the linearity of

 G_{tx} and τ^{L} , the variation with respect to α can be written as:

$$\delta_{ux,ia} G_{ix}(\Delta^{III} \cdot \Delta^{III}) = 2 G_{ix}(\Delta^{III} \cdot \delta \Delta^{III})$$

$$= -2 \ \mathbb{G}_{tx}([\tau^{N}(\mathbf{w}^{\bullet}) - \tau^{L}(\alpha, \mathbf{w}^{\bullet}) - \dot{\mathbf{w}}^{L}(\beta, \mathbf{w}^{\bullet})] \cdot \tau^{L}(\delta\alpha, \mathbf{w}^{\bullet}))$$

The variation with respect to β can be written in a similar way. Therefore the extremizaton in (3.9) can be written as

$$G_{ix}([\tau^{N}(\mathbf{w}^{*}) - \tau^{L}(\alpha, \mathbf{w}^{*}) - \dot{\tau}^{L}(\beta, \mathbf{w}^{*})] \cdot \tau^{L}(\delta\alpha, \mathbf{w}^{*})) = 0$$
(3.10)

$$G_{tx}([\tau^{N}(\mathbf{w}^{\bullet}) - \tau^{L}(\alpha, \mathbf{w}^{\bullet}) - \dot{\tau}^{L}(\beta, \mathbf{w}^{\bullet})] \cdot \dot{\tau}^{L}(\delta\beta, \mathbf{w}^{\bullet})) = 0$$
(3.11)

Integration by parts over $\mathfrak{D}(\mathbf{x})$ of (3.10) and (3.11) will produce a set of Euler equations in $\boldsymbol{\alpha}$ and $\boldsymbol{\beta}$ with \mathbf{x} being the independent variable. The Euler equations for $\boldsymbol{\alpha}$ and $\boldsymbol{\beta}$ will generally be partial differential equations for which general solutions are not readily available. Therefore the exact minimization has not significantly simplified the solution procedure for the original nonlinear partial differential equations. In what follows, an approximate method for performing the extremization of (3.9) will be explored which will produce algebraic relationships for the minimizing stiffness and damping parameters.

Suppose the equivalent linear parameters $\alpha(\mathbf{x})$ and $\beta(\mathbf{x})$ are approximated by the following functions

$$\boldsymbol{\alpha}(\mathbf{x}) \approx \sum_{i=1}^{M} \varphi_i(\mathbf{x}) \widehat{\boldsymbol{\alpha}}_i$$
(3.12)

$$\boldsymbol{\beta}(\mathbf{x}) \approx \sum_{i=1}^{M} \varphi_i(\mathbf{x}) \hat{\boldsymbol{\beta}}_i$$
(3.13)

where $\hat{\alpha}_i$ and $\hat{\beta}_i$ are unknown constants and $\varphi_i(\mathbf{x})$; i=1, 2, ..., M, are linearly independent functions of \mathbf{x} .

Sufficient conditions for the minimization of $G_{tx}(\Delta \cdot \Delta)$ with respect to the approximations to α and β given in (3.12) and (3.13) are

$$\frac{\partial}{\partial \widehat{\alpha}_k} \quad \widehat{\alpha}_{tr} (\Delta \cdot \Delta) = 0 \tag{3.14}$$

$$\frac{\partial}{\partial \beta_{k}} G_{k}(\Delta \cdot \Delta) = 0$$
(3.15)

The question of whether the $\hat{\alpha}$ and $\hat{\beta}$ that satisfy (3.14) and (3.15) truly minimize $G_{tx}(\Delta \cdot \Delta)$, and not maximizes $G_{tx}(\Delta \cdot \Delta)$, will be examined in a later section.

Let attention be focused on developing the minimizing relations (3.14) and (3.15) for the specific problem formulations. Up to this point, the equivalent linear parameters α and β were assumed to be sets of parameters for the sake of generality. To expedite the developing of the minimizing relationships, let it be assumed that α and β are sets of one parameter each, or simply α and β . Similar relationships to those in the following sections can be established for larger sets of α and β .

3.5.1 Formulation I minimization

For problem formulation I, the differential stress difference Δ^{I} was defined as

$$\Delta^{I} = \nabla \cdot \tau^{N} \langle \mathbf{w}^{*}, \dot{\mathbf{w}}^{*} \rangle - \nabla \cdot \tau^{L} \langle \alpha, \mathbf{w}^{*} \rangle - \nabla \cdot \dot{\tau}^{L} \langle \beta, \mathbf{w}^{*} \rangle$$
(3.16)

Since τ^{L} is a linear operation,

$$\tau^{L}(\widehat{\alpha}, \mathbf{w}^{\bullet}) = \sum_{j=1}^{M} \tau_{j}^{L} \widehat{\alpha}_{j}$$
(3.17)

where τ_j^L is a shorthand notation for $\tau^L(\varphi_j, \mathbf{w}^*)$. Similarly, $\tau^N(\mathbf{w}^*, \mathbf{w}^*)$ will be written as τ^N , with the operation on \mathbf{w}^* and \mathbf{w}^* implied.

Using (3.16) and (3.17) and the linear property of G_{tx} in the first minimizing

relation (3.14) gives:

$$\frac{\partial}{\partial \widehat{\alpha}_{k}} \quad G_{tx}(\Delta^{I} \cdot \Delta^{I}) = 2 \quad G_{tx}(\Delta^{I} \cdot \frac{\partial \Delta^{I}}{\partial \widehat{\alpha}_{k}})$$
$$0 = \sum_{i=1}^{M} \quad G_{tx}([\nabla \cdot \tau^{N} - \nabla \cdot \tau_{i}^{L} \widehat{\alpha}_{i} - \nabla \cdot \tau_{i}^{L} \widehat{\beta}_{i}] \cdot [\nabla \cdot \tau_{k}^{L}]) \quad (3.18)$$

Equation (3.18) can be written in matrix form as:

$$A^{I}\hat{\boldsymbol{\alpha}} + C^{I}\hat{\boldsymbol{\beta}} = \mathbf{G}$$
(3.19)

where

$$A_{ij}^{I} = G_{ix}([\nabla \cdot \tau_{i}^{L}] \cdot [\nabla \cdot \tau_{j}^{L}])$$

$$C_{ij}^{I} = G_{ix}([\nabla \cdot \dot{\tau}_{j}^{L}] \cdot [\nabla \cdot \tau_{i}^{L}])$$

$$G_{i}^{I} = G_{ix}([\nabla \cdot \tau^{N}] \cdot [\nabla \cdot \tau_{i}^{L}])$$

$$(3.20)$$

$$\widehat{\boldsymbol{\alpha}}, \ \widehat{\boldsymbol{\beta}} = (\widehat{\alpha}_1, \ldots, \widehat{\alpha}_M)^T \text{ and } (\widehat{\boldsymbol{\beta}}_1, \ldots, \widehat{\boldsymbol{\beta}}_M)^T$$
 (3.21)

Similiarly, the minimization relation (3.15) becomes

$$D^{I}\hat{\alpha} + B^{I}\hat{\beta} = \mathbf{H}^{I} \tag{3.22}$$

where

$$D_{ij}^{I} = G_{tx}([\nabla \cdot \tau_{i}^{L}] \cdot [\nabla \cdot \dot{\tau}_{i}^{L}])$$

$$B_{ij}^{I} = G_{ix}([\nabla \cdot \dot{\tau}_{i}^{L}] \cdot [\nabla \cdot \dot{\tau}_{j}^{L}])$$

$$H_{i}^{I} = G_{tx}([\nabla \cdot \tau^{N}] \cdot [\nabla \cdot \dot{\tau}_{i}^{L}])$$
(3.23)

Equations (3.19) and (3.22) can simplified further by observing the form of matrices C^{I} and D^{I} along with properties of the operator G_{tx} . First note that

$$C^I = (D^I)^T \tag{3.24}$$

where $(D^I)^T$ is the transpose of the matrix D^I . For any two functions $\mathbf{u}(\mathbf{x},t)$ and $\mathbf{v}(\mathbf{x},t)$ on $\mathcal{D}(\mathbf{x})$, and since $G_{tx}(\cdot)$ is linear and time invariant,

$$\frac{\partial}{\partial t} G_{tx}(\mathbf{u} \cdot \mathbf{v}) = G_{tx}(\dot{\mathbf{u}} \cdot \mathbf{v}) + G_{tx}(\mathbf{u}, \dot{\mathbf{v}})$$

= 0

Therefore

$$G_{tx}(\mathbf{\dot{u}}\cdot\mathbf{v}) = - \quad tx(\mathbf{u}\cdot\mathbf{\dot{v}}) \tag{3.25}$$

Looking at equations (3.20) and (3.23), it is clear that (3.25) implies

$$C^I = -(D^I)^T \tag{3.26}$$

Equations (3.24) and (3.26) together say that

$$C_{ij} = D_{ij} = 0$$
; $i, j = 1, 2, ..., M$ (3.27)

Therefore the equations for the formulation I parameters $\hat{\alpha}$ and $\hat{\beta}$ are uncoupled from each other.

3.5.1 Formulation II minimization

The system difference for formulation II, the scalar Δ^{II} , was defined to be

$$\Delta^{II} = [\tau^N - \tau^L(\alpha) - \dot{\tau}^L(\beta)] \cdot \mathbb{S}$$

Using (3.14) and the shorthand notation of S for $S(\mathbf{w}^*)$, the minimization relations (3.14) and (3.15) can be written as:

$$A^{II}\widehat{\alpha} + C^{II}\widehat{\beta} = \mathbf{G}^{II} \tag{3.28}$$

$$D^{II}\hat{\alpha} + B^{II}\hat{\beta} = \mathbf{H}^{II} \tag{3.29}$$

where

$$A_{ij}^{II} = G_{ix}([\tau_i^L \cdot S][\tau_j^L \cdot S])$$

$$B_{ij}^{II} = G_{ix}([\dot{\tau}_i^L \cdot S][\dot{\tau}_j^L \cdot S])$$

$$C_{ij}^{II} = G_{ix}([\dot{\tau}_j^L \cdot S][\tau_i^L \cdot S])$$

$$D_{ij}^{II} = G_{ix}([\tau_j^L \cdot S][\dot{\tau}_i^L \cdot S])$$

$$G_{i}^{II} = G_{ix}([\tau^N \cdot S][\tau_i^L \cdot S])$$

$$H_{i}^{II} = G_{ix}([\tau^N \cdot S][\dot{\tau}_i^L \cdot S])$$

$$(3.30)$$

As was true for the formulation I minimization relations, the matrix D^{II} is the transpose of C^{II} , or

$$D_{ij}^{II} = C_{ji}^{II} \tag{3.31}$$

However C^{II} and D^{II} are not null matrices, as was the case for formulation I. Consequently, the equations for $\hat{\alpha}$ and $\hat{\beta}$ remain coupled for a general system. As will be seen in section 3.8, the uncoupling of the equations is sufficient condition to guarantee the existence of unique solutions for $\hat{\alpha}$ and $\hat{\beta}$.

There does exist, however, at least one special definiton of the strain operation, S, and of the basis functions, $\varphi_i(\mathbf{x})$, which permit an uncoupling of the minimization equations (3.28) and (3.29). It can be shown that, if $\varphi_i(\mathbf{x})$; i=1, 2, ..., M are such that C^{II} and D^{II} are diagonal matrices and the operation S is proportional to τ^L , C^{II} and D^{II} vanish identically, thereby uncoupling the minimization relations.

3.5.3 Formulation III minimization

The formulation III system difference was given by

$$\Delta = \tau^N - \tau^L(\alpha) - \dot{\tau}^L(\beta) \tag{3.32}$$

In the same manner as for formulations I and II, the minimization relations can be shown to give the following equations for $\hat{\alpha}$ and $\hat{\beta}$:

$$A^{III}\hat{\boldsymbol{\alpha}} + C^{III}\hat{\boldsymbol{\beta}} = \mathbf{G}^{III} \tag{3.33}$$

$$D^{III}\hat{\alpha} + B^{III}\hat{\beta} = \mathbf{H}^{III} \tag{3.34}$$

where

ť

$$A_{ij}^{III} = G_{tx}(\tau_i^L \cdot \tau_j^L)$$

$$B_{ij}^{III} = G_{tx}(\dot{\tau}_i^L \cdot \dot{\tau}_j^L)$$

$$C_{ijI}^{III} = G_{tx}(\dot{\tau}_j^L \cdot \tau_i^L)$$

$$D_{ijI}^{III} = G_{tx}(\tau_j^L \cdot \dot{\tau}_i^L)$$

$$G_{i}^{III} = G_{tx}(\tau^N \cdot \tau_i^L)$$

$$H_{i}^{III} = G_{tx}(\tau^N \cdot \dot{\tau}_i^L)$$

$$(3.35)$$

for i,j=1,2,...,M

Here again (3.35) shows that

$$C^{III} = (D^{III})^T$$
 (3.36)

Equation (3.36) and the time invariance of G_{tx} () implies that

$$C_{ij}^{III} = D_{ij}^{III} = 0$$
; $i, j = 1, 2, ..., M$

which uncouples the equations for $\widehat{\alpha}$ and $\widehat{\beta}$, as was the case with formulation I.

3.6 Examination of the Minimum

In determining the values of the equivalent linear parameters $\hat{\alpha}$ and $\hat{\beta}$ which made the averaged system difference an extreme value, there is no guarantee that the extreme value is truly a minimum and not actually a maximum. It will be shown in this section that the values of $\hat{\alpha}$ and $\hat{\beta}$ established from equations (3.14) and (3.15) do in fact make the averaged system difference term $G_{tx}(\Delta \cdot \Delta)$ a minimum.

Again let Δ represent the general system difference term from either formulation I, II or III, and G_{tx} have the properties specified in section 3.4. Recall the definitions of Δ for the three problem formulations given in section 3.3. All three definitions can be written in the following general form, when the approximations to α and β of (3.12) and (3.13) are considered:

$$\Delta = \mathcal{F}^{N}(\mathbf{w}, \dot{\mathbf{w}}) - \sum_{k=1}^{M} \mathcal{F}^{a}_{k}(\mathbf{w}) \hat{\alpha}_{k} - \sum_{k=1}^{M} \mathcal{F}^{b}_{k}(\dot{\mathbf{w}}) \hat{\beta}_{k} \qquad (3.37)$$

where \mathcal{F}^N , is a nonlinear operation on w and $\dot{\mathbf{w}}$. \mathcal{F}^a_k and \mathcal{F}^b_k ; k=1, 2, ..., M are linear operations on w and $\dot{\mathbf{w}}$.

Then $\Delta \cdot \Delta$ is a scalar, *quadratic* polynomial in the parameters $\hat{\alpha}$ and $\hat{\beta}$. Therefore its mixed partial derivatives with respect to $\hat{\alpha}_k$ and $\hat{\beta}_k$; k=1, 2, ..., M, of order higher than 2 vanish. With $\hat{\alpha}$ and $\hat{\beta}$ satisfying the extremization relationships of (3.14) and (3.15), let another set of parameters $\hat{\alpha}'$ and $\hat{\beta}'$ be defined as:

$$\widehat{\alpha}'_{k} = \widehat{\alpha}_{k} + \delta \widehat{\alpha}_{k}$$

$$\widehat{\beta}'_{k} = \widehat{\beta}_{k} + \delta \widehat{\beta}_{k}$$
(3.38)

Using the shorthand notation of $\Delta' \equiv \Delta(\widehat{\alpha}, \widehat{\beta}')$ and $\Delta \equiv \Delta(\widehat{\alpha}, \widehat{\beta})$, the averaged inner product $G_{tx}(\Delta' \cdot \Delta')$ can be expanded in a Taylor series expansion about $\widehat{\alpha}$

and $\hat{\beta}$ as

$$\begin{aligned} G_{tx}(\Delta' \cdot \Delta') &= G_{tx}(\Delta \cdot \Delta) \\ &+ \sum_{p=1}^{M} \frac{\partial}{\partial \widehat{\alpha}_{p}} \ G_{tx}(\Delta \cdot \Delta) \delta \widehat{\alpha}_{p} + \frac{\partial}{\partial \widehat{\beta}_{p}} \ G_{tx}(\Delta \cdot \Delta) \delta \widehat{\beta}_{p} \\ &+ \frac{1}{2} \sum_{m,p=1}^{M} \frac{\partial}{\partial \widehat{\alpha}_{m}} \frac{G_{tx}(\Delta \cdot \Delta)}{\partial \widehat{\alpha}_{m}} \delta \widehat{\alpha}_{p} + \frac{1}{2} \sum_{m,p=1}^{M} \frac{\partial}{\partial \widehat{\beta}_{m}} \frac{G_{tx}(\Delta \cdot \Delta)}{\partial \widehat{\beta}_{m}} \delta \widehat{\beta}_{p} \\ &+ \sum_{m,p=1}^{M} \frac{\partial}{\partial \widehat{\alpha}_{m}} \frac{G_{tx}(\Delta \cdot \Delta)}{\partial \widehat{\alpha}_{m}} \delta \widehat{\alpha}_{p} \delta \widehat{\alpha}_{m} \delta \widehat{\beta}_{p} \end{aligned}$$
(3.39)

38

But, from (3.14) and (3.15),

$$\frac{\partial}{\partial \hat{\alpha}_{p}} G_{tx}(\Delta \cdot \Delta) = 0$$

$$\frac{\partial}{\partial \hat{\beta}_{p}} G_{tx}(\Delta \cdot \Delta) = 0$$

$$(3.40)$$

The higher order derivatives can be written , with the use of (3.40) as:

 $= 2 \ \mathbb{G}_{tx}(\ \mathbb{F}_m^a \cdot \ \mathbb{F}_p^b)$

$$\frac{\partial^{2}}{\partial \widehat{\alpha}_{m} \partial \widehat{\alpha}_{p}} \quad \mathbb{G}_{tx}(\Delta \cdot \Delta) = 2 \quad \mathbb{G}_{tx} \quad \frac{\partial \Delta}{\partial \widehat{\alpha}_{m}} \cdot \frac{\partial \Delta}{\partial \widehat{\alpha}_{p}}$$

$$= 2 \quad \mathbb{G}_{tx}(\quad \Im_{m}^{a} \cdot \quad \Im_{p}^{a}) \qquad (3.41)$$

$$\frac{\partial^{2}}{\partial \widehat{\beta}_{m} \partial \widehat{\beta}_{p}} \quad \mathbb{G}_{tx}(\Delta \cdot \Delta) = 2 \quad \mathbb{G}_{tx} \quad \frac{\partial \Delta}{\partial \widehat{\beta}_{m}} \cdot \frac{\partial \Delta}{\partial \widehat{\beta}_{p}}$$

$$= 2 \quad \mathbb{G}_{tx}(\quad \Im_{m}^{b} \cdot \quad \Im_{p}^{b}) \qquad (3.42)$$

$$\frac{\partial^{2}}{\partial \widehat{\alpha}_{m} \partial \widehat{\beta}_{p}} \quad \mathbb{G}_{tx}(\Delta \cdot \Delta) = 2 \quad \mathbb{G}_{tx} \quad \frac{\partial \Delta}{\partial \widehat{\alpha}_{m}} \cdot \frac{\partial \Delta}{\partial \widehat{\beta}_{p}}$$

(3.43)

Using equations (3.40) - (3.43) in (3.39) and simplifying through the linearity of C_{tx} gives

$$G_{ix}(\Delta' \cdot \Delta') = G_{ix}(\Delta \cdot \Delta) + G_{ix}(f(\delta \widehat{\alpha}, \delta \widehat{\beta}) \cdot f(\delta \widehat{\alpha}, \delta \widehat{\beta}))$$
(3.44)

where

$$\mathbf{f}(\delta\widehat{\alpha},\,\delta\widehat{\beta}\,)=\sum_{m=1}^{M}\left(\ \Im_{m}^{a}\delta\widehat{\alpha}_{m}\,+\,\,\Im_{m}^{b}\delta\widehat{\beta}_{m}\right)$$

The positive definite properties of G_{tx} and ()-() prescribes that

$$\begin{array}{c} \Omega_{tx}(\mathbf{f}(\delta\hat{\boldsymbol{\alpha}},\delta\hat{\boldsymbol{\beta}})\cdot\mathbf{f}(\delta\hat{\boldsymbol{\alpha}},\delta\hat{\boldsymbol{\beta}})) > 0 \\ \\ \text{for } \mathbf{f}(\delta\hat{\boldsymbol{\alpha}},\delta\hat{\boldsymbol{\beta}}) \neq 0 \end{array} \end{array}$$
(3.45)

and

$$\begin{array}{c}
\Omega_{tx}(\mathbf{f}(\delta\widehat{\boldsymbol{\alpha}},\delta\widehat{\boldsymbol{\beta}})\cdot\mathbf{f}(\delta\widehat{\boldsymbol{\alpha}},\delta\widehat{\boldsymbol{\beta}})) = 0 \\
\text{iff} \quad \mathbf{f}(\delta\widehat{\boldsymbol{\alpha}},\delta\widehat{\boldsymbol{\beta}}) = 0
\end{array}$$
(3.46)

۱

Hence

$$G_{tx}(\Delta(\widehat{\alpha}', \widehat{\beta}') \cdot \Delta(\widehat{\alpha}', \widehat{\beta}')) \ge G_{tx}(\Delta(\widehat{\alpha}, \widehat{\beta}) \cdot \Delta(\widehat{\alpha}, \widehat{\beta}))$$
(3.47)

In words, equation (3.47) says that for parameters $\hat{\alpha}$ and $\hat{\beta}$ that make $G_{tx}(\Delta \cdot \Delta)$ an extremum, the corresponding extreme value is no larger than any other parameters $\hat{\alpha}'$ and $\hat{\beta}'$. Therefore $G_{tx}(\Delta(\hat{\alpha}, \hat{\beta}) \cdot \Delta(\hat{\alpha}, \hat{\beta}))$ is a global minimum. Note, however, that this does not preclude the possibility of more than one set of $\hat{\alpha}$ and $\hat{\beta}$ which minimizes $G_{tx}(\Delta \cdot \Delta)$. If non-unique solutions do exist, then relation (3.46) implies a linear dependence of the operations \Im_m^a and \Im_m^b ; m=1, 2, ..., M, with the form of \Im_m^a and \Im_m^b of course depending on the problem formulation. The existence and uniqueness of the solutions for $\hat{\alpha}$ and

 $\widehat{oldsymbol{eta}}$ will be investigated in a later section for the three problem formulations.

3.7 Spatial Discretization of the Auxiliary Equations

In solving the type of equations given by the linear auxiliary system (3.1), approximate techniques are usually employed. This is especially true when the problem contains inhomogeneous material properties and/or has complex boundary conditions that need to be satisfied. The former will always be true for the equivalent linearization method due to the manner in which the linear parameters are discretized.

The finite element method enjoys widespread usage in solving the types of problems described by (3.1). The underlying principle of the method is the subdivision of the total spatial domain , $\mathfrak{D}(\mathbf{x})$, into subdomains, or elements $\mathfrak{D}^{\mathbf{e}}(\mathbf{x})$. Within each element the displacements are discretized in such a way that they can be written in terms of linearly independent basis, or "shape", functions of the spatial coordinates, or

$$\mathbf{w}(\mathbf{x},t) = N(\Psi)\mathbf{d}(t) \tag{3.48}$$

where N is a matrix of a set of shape functions $\Psi(\mathbf{x})$ and **d** is a set of time dependent nodal coefficients. A Galerkin procedure, with **w** defined as in (3.48), can be then be used to approximate the solution to (3.1). This procedure leads to the following discretized equation

$$\mathcal{M}\ddot{\mathbf{d}}(t) + \mathcal{C}\dot{\mathbf{d}}(t) + \lambda \mathbf{d}(t) = \mathbf{f}(t)$$
(3.49)

where the vector **f** and the matrices \mathcal{M} , \mathcal{C} and \mathcal{X} are integral functionals of the shape functions Ψ and their derivatives.

The striking difference between the equations (3.49) from the finite element method, and an analogous set derived from the general Galerkin formulation, lies in the shape functions Ψ . The basis functions used for the finite element method are simple polynomials which are defined over only one element. With such a compact support, boundary conditions are easier to impose locally, along the edge of an element, rather than globally along a more complicated boundary. The accuracy of the method is increased by a refinement of the subdivisions, and not by the classical Galerkin method of including more complex basis functions. The advantage of this is in the ease of implementation into a computer code; that is, the computer instructions remain nearly the same as the mesh is refined.

In order to determine suitable shape functions for a particular problem, the convergence properties of the finite element method must be considered. Convergence of the method, as the mesh of elements is refined, depends on the continuity of the shape functions used. In particular, the stiffness matrix χ in equation (3.49) will generally be the strain-energy integral for the system, which will involve a number, say m, of derivatives of the shape functions. Then, if the m-1 derivatives are continuous across the elements, the integrand is finite and the strain-energy integral exists. This is a simplified explanation of one criterion of convergence, which is sufficient continuity of the shape functions [54]. The use of shape functions which do not satisfy this continuity requirement does appear in the literature (for example [49]), but attention here will only be given to shape functions which satisfy continuity.

Another requirement for convergence of the finite element method is mathematical *completeness*. From a physical standpoint, completeness implies that the finite element mesh must be capable of modeling a constant strain condition throughout the domain. Completeness can be established for isoparametric elements (see [54]): that is, elements which use the same shape functions on the spatial coordinates as on the displacement field. For the case of non-isoparametric elements, the constant strain condition can be established

41

through an element 'patch test" (reference [48]).

Continuity of the shape functions is also a consideration in the minimization procedure for determining the equivalent linear parameters $\alpha(\mathbf{x})$ and $\beta(\mathbf{x})$. For example, in the formulation III minimization relations of equations (3.33) - (3.35), a typical integration that must be performed is:

$$A_{ij}^{III} = G_t \left(\int \tau^L(\varphi_i, \mathbf{w}) \cdot \tau^L(\varphi_j, \mathbf{w}) d \mathcal{D}(\mathbf{x}) \right)$$
(3.50)

From (3.50), it is seen that τ^L must be continuous in the mean square sense for derivatives of w up to n-1, where n is the number of spatial derivatives of w in τ^L . Generally n will be greater than m, where m has previously been defined as the number of derivatives in the strain energy integral. Therefore, continuity requirements of the displacement field imposed by equivalent linearization tend to be more restrictive than those of the finite element method.

3.8 Existence and Uniqueness of Equivalent Linear Parameters

In section 3.5, a set of relationships were established between the equivalent linear parameters and the solution of the linear auxiliary system such that the difference between given properties of the nonlinear system and the auxiliary system are minimized. The preceding section dealt with the spatial discretization of the auxiliary system, which produces a second set of relations between the equivalent linear solution and the equivalent linear stiffness and damping parameters. Before these ideas are applied to specific problems, some consideration must be given to whether this technique is capable of producing reasonable solutions. Therefore this section deals with the question of whether the minimization technique yields equations for which solutions exist and, if solutions exist, whether they are unique. The equations for the equivalent linear parameters $\hat{\alpha}$ and $\hat{\beta}$ which minimized the system differences for problem formulations I. II and III can be written in the general form (see equations (3.19), (3.22), (3.28), (3.29), (3.33) and (3.34)):

$$\begin{bmatrix} A & C \\ C^T & B \end{bmatrix} \begin{pmatrix} \widehat{\alpha} \\ \widehat{\beta} \end{pmatrix} = \begin{pmatrix} G \\ H \end{pmatrix}$$
(3.51)

where the matrices A, B and C and the vectors G and H depend on the problem formulation and are functions of the auxiliary equation solutions.

As shown in section 3.5.1 - 3.5.3, the matrix C always vanishes for formulations I and III. Therefore, for the case of uncoupled equations, the existence of unique solutions for $\hat{\alpha}$ and $\hat{\beta}$ depend on the invertibility of matrices A and B. The question of whether a matrix is non-singular can be answered by showing that it is sign definite; that is, if all the eigenvalues of a matrix tend away from zero, then its determinant will never vanish.

The following claims will show that the matrices A and B for all three problem formulations are positive definite according to the definition:

Definition - Matrix Positive Definiteness

A matrix T of size $M \ge M$ will said to be positive definite if for an M - vector r

$$\mathbf{r}^{T}T\mathbf{r} > 0 \quad ; \text{ if } \mathbf{r} \neq \mathbf{0} \tag{3.52}$$
$$\mathbf{r}^{T}T\mathbf{r} = 0 \quad ; \text{ iff } \mathbf{r} = \mathbf{0}$$

where \mathbf{r}^{T} is the transpose of \mathbf{r} .

Claim - Positive Definiteness of A^{I} and B^{I}

For matrices A^{I} and B^{I} defined in equations (3.20) and (3.23) with $\mathbf{w} \neq \mathbf{0}$, $\mathbf{\dot{w}} \neq \mathbf{0}$ for all \mathbf{x} and t, A^{I} and B^{I} are positive definite. Proof

Recall that for formulation I, A^{I} and B^{I} were derived to be

$$A_{ij}^{I} = G_{tx}([\nabla \cdot \tau^{L}(\varphi_{i}, \mathbf{w})] \cdot [\nabla \cdot \tau^{L}(\varphi_{j}, \mathbf{w})])$$
$$B_{ij}^{I} = G_{tx}([\nabla \cdot \dot{\tau}^{L}(\varphi_{i}, \mathbf{w})] \cdot [\nabla \cdot \dot{\tau}^{L}(\varphi_{j}, \mathbf{w})])$$

Then

$$\mathbf{r}^{T} A^{I} \mathbf{r} = \sum_{i,j=1}^{M} r_{i} \quad \mathbb{G}_{tx} \left(\left[\nabla \cdot \tau^{L}(\varphi_{i}, \mathbf{w}) \right] \cdot \left[\nabla \cdot \tau^{L}(\varphi_{j}, \mathbf{w}) \right] \right) r_{j}$$
$$= \quad \mathbb{G}_{tx} \left(\left[\nabla \cdot \tau^{L}(\widehat{r}, \mathbf{w}) \right] \cdot \left[\nabla \cdot \tau^{L}(\widehat{r}, \mathbf{w}) \right] \right)$$
(3.53)

where $\hat{\tau} = \sum_{i=1}^{M} \varphi_i(\mathbf{x}) r_i$. Note that the linearity of G_{tx} and τ^L were used in obtaining (3.53). Taking into account the positive definitenes of G_{tx} and the inner product operation, equation (3.53) says that:

$$\mathbf{r}^{T}A^{I}\mathbf{r} \ge 0$$
(3.54)
$$\mathbf{r}^{T}A^{I}\mathbf{r} = 0 \quad \text{iff} \quad \nabla \cdot \tau^{L}(\hat{\boldsymbol{r}}, \mathbf{w}) = 0$$

Forming the inner product of \mathbf{w} and $\nabla \cdot \tau^L$, integrating over $\mathfrak{D}(\mathbf{x})$, and using the last equation in (3.54) gives

$$\int \mathbf{w} \cdot \left[\nabla \cdot \tau^{L}(\hat{r}, \mathbf{w}) \right] d \mathcal{D}(\mathbf{x}) = 0$$
(3.55)

Since $\nabla \cdot \tau^{L}(\hat{r}, \mathbf{w})$ is positive definite according to section 2.3.2, equation (3.55) can be satisfied for $\mathbf{w} \neq \mathbf{0}$ iff

$$\hat{r} = \sum_{i=1}^{M} \varphi_i(\mathbf{x}) r_i = 0$$
(3.56)

The basis functions $\varphi_i(\mathbf{x})$; i=1, 2, ..., M, are linearly independent. Hence

 $\mathbf{r} = [r_1, r_2, \cdots, r_M]^T = \mathbf{0}$, and therefore A^I is positive definite.

Now looking at B^{I} ,

$$\mathbf{r}^{I}B^{I}\mathbf{r} = \mathbf{G}_{tx}([\nabla \cdot \dot{\tau}^{L}(\hat{r}, \mathbf{w})] \cdot [\nabla \cdot \dot{\tau}^{L}(\hat{r}, \mathbf{w})])$$

and by the same arguments as with A^{I} ,

$$\mathbf{r}^{T}B^{I}\mathbf{r} \ge 0$$

$$\mathbf{r}^{T}B^{I}\mathbf{r} = 0 \quad \text{iff} \quad \nabla \cdot \dot{\tau}^{L}(\hat{\tau}, \mathbf{w}) = \mathbf{0}$$

$$(3.57)$$

 τ^{L} is a linear operation. Therefore $\dot{\tau}^{L}(\hat{r}, \mathbf{w}) = \tau^{L}(\hat{r}, \dot{\mathbf{w}})$, and with (3.57)

$$\int \dot{\mathbf{w}} \cdot \left[\nabla \cdot \tau^{L}(\hat{r}, \dot{\mathbf{w}}) \right] d \quad \mathcal{Q}(\mathbf{x}) = 0$$
(3.58)

Again for positive definite $\nabla \cdot \tau^L$, $\dot{\mathbf{w}} \neq 0$ and linearly independent φ_i , $\mathbf{r} \equiv 0$ if (3.57) holds, and B^I is positive definite. Q.E.D.

Claim

The matrices A^{II} and B^{II} developed for formulation II are positive definite.

Claim

The matrices A^{III} and B^{III} developed for formulation III are positive definite.

The proofs for the preceding two claims are quite similar to the proof for formulation I and therefore are not included here.

In conclusion, solutions for the formulation I and III equivalent parameters always exist for non-zero displacements and are unique. For the special case described in section 3.5.2 in which the equations for $\hat{\alpha}$ and $\hat{\beta}$ uncouple, unique solutions for formulation II parameters exist. Unfortunately, attempts to show existence and uniqueness of the general formulation II minimization solutions have not been successful.

IV. SPECIAL CASES OF THE GENERAL APPROACH

Chapter III dealt with the development of the general equivalent linearization approach for nonlinear continuous systems. The relationships between the equivalent linear nodal parameters, $\hat{\alpha}$ and $\hat{\beta}$ and the displacements w which belong to the class of functions C were established. Describing C are the basis functions which are solutions to the linear auxiliary system given by equations (3.1) and (3.2). In this chapter, the equivalent linear parameter relationships are specialized for two types of excitation, g(t), for the nonlinear system of equations (2.1) written in the form

$$\nabla \cdot \tau^{L}(\mathbf{a}) + \nabla \cdot \tau^{N}(\mathbf{c}) + \dot{\mathbf{Q}}^{L}(\mathbf{b}) + m(\mathbf{x}) \ddot{\mathbf{w}} = \mathbf{p}(\mathbf{x})g(t)$$
(4.1)

The linear auxiliary equations are written as

$$\nabla \cdot \tau^{L}(\mathbf{a}) + \nabla \cdot \tau^{L}(\alpha) + \dot{\mathbf{Q}}^{L}(\mathbf{b}) + \nabla \cdot \dot{\tau}^{L}(\beta) + m(\mathbf{x}) \ddot{\mathbf{w}} = \mathbf{p}(\mathbf{x})g(t)$$
(4.2)

where, as before, the functional dependence of τ^L , τ^N and \mathbf{Q}^L on \mathbf{w} and $\mathbf{\dot{w}}$ is understood. For the excitation g(t) in (4.1) and (4.2), attention is focused on the cases of monofrequency harmonic and stationary Gaussian excitation.

Interest in harmonic response often arises when rotating machinery serves as the excitation of the structural element, or when adjoining structural elements filter the excitation to a single frequency input. In addition to this, the forced vibration tests of structural elements to determine stiffness and energy dissipation properties frequently use harmonic excitation. Therefore, steadystate response of a nonlinear system has significant relevance to engineering problems.

The random nature of excitations due to earthquakes and acoustical noise is also of engineering importance. If the analyst considers a large number of independent records of the random process where no one record's contribution is more significant than the others, the central limit theorem says that process can be considered *normal*, or Gaussian. If the average and second moments of these records can be assumed to be good substitutes for the actual records of the process, then the use of a Gaussian excitation would appear to be a good approach to these engineering problems.

For simplicity of notation, the displacements $\mathbf{w}(\mathbf{x}, t)$ will be assumed to be simply a scalar function $w(\mathbf{x}, t)$. The examples to be considered in the next chapter will fit into this category. Therefore use of relations developed can be applied to the examples in a straight forward manner. Extension of the results of this chapter can be made to problems requiring a vector representation of displacements but only at the expense of notation.

4.1 Steady-State Harmonic Response

Consider the case where $g(t) = \cos \omega t$ in the nonlinear equation and auxiliary linear equation in (4.1) and (4.2). The nonlinearity, τ^N , of equation (4.1) has been assumed to be symmetric in accordance to the definition of section 2.3.4. Therefore, if the linear auxiliary system of equations (4.2) is to model the nonlinear system, the steady-state, linear solution can be expanded in the first terms of its Fourier series as

$$w(\mathbf{x},t) = U(\mathbf{x})\cos\omega t + V(\mathbf{x})\sin\omega t$$
(4.3)

Hence, the set of basis functions for steady-state harmonic response is $[\cos \omega t, \sin \omega t]$.

In discretizing the auxiliary system, the functions $U(\mathbf{x})$ and $V(\mathbf{x})$ from equation (4.3) will be approximated by

$$U(\mathbf{x}) \approx \sum_{i=1}^{N} \psi_i(\mathbf{x}) u_i \tag{4.4}$$

$$V(\mathbf{x}) \approx \sum_{i=1}^{N} \psi_i(\mathbf{x}) v_i \tag{4.5}$$

where ψ_i is a set of N spatial basis functions and u_i and v_i are undetermined coefficients for i=1,2,...,N. Then the discretization can be carried out through the finite element method, Galerkin's method, modal analysis, etc. Equating the coefficients of the sine and cosine functions will give a set of matrix algebraic equations for **u** and **v** in terms of the equivalent linear parameters \hat{a} and $\hat{\beta}$ as:

$$\left[-\omega^2 \,\mathcal{M} + \mathcal{K}\left(\widehat{\alpha}\right)\right]\mathbf{u} + \omega \,\,\mathbb{C}\left(\widehat{\beta}\right)\mathbf{v} = \mathbf{f} \tag{4.6}$$

$$-\omega \ \mathbb{C}(\widehat{\beta})\mathbf{u} + \left[-\omega^2 \ \mathcal{M} + \ \mathcal{X}(\widehat{\alpha})\right]\mathbf{v} = \mathbf{0}$$
(4.7)

where \mathcal{M} , C and \mathcal{X} are the mass, damping and stiffness matrices, respectively. f is the discretized excitation term. The form of the matrices and vector will depend on the method used for discretization. Equations (4.6) and (4.7) can then be solved for the 'in-phase'' and 'quadrature'' (w.r.t. the excitation) solutions u and v, respectively, once the stiffness and damping parameters $\hat{\alpha}$ and $\hat{\beta}$ are known.

The system difference minimization procedure produces a set of relations for $\hat{\alpha}$ and $\hat{\beta}$ in terms of the auxiliary equation solution w, as were developed in section 3.5. These equations were written in terms of the general temporal averaging operator G_t . Since the steady-state solution for the auxiliary system (4.2) with g(t) harmonic is periodic with period

$$T = \frac{2\pi}{\omega},\tag{4.8}$$

a logical choice of G_t which satisfies the properties of G_t in section 3.4 is

$$G_t(\cdot) = \int_0^T (\cdot) dt$$
(4.9)

Using this definition of G₂, the minimization relations

$$\frac{\partial}{\partial \hat{\alpha}_{k}} \quad G_{tx}(\Delta \cdot \Delta) = 0$$

$$\frac{\partial}{\partial \hat{\beta}_{k}} \quad G_{tx}(\Delta \cdot \Delta) = 0$$

$$(4.10)$$

for k=1,2,...,M, will be specialized for steady-state harmonic analysis for the three problem formulations.

Recall first that the equations (4.10) could be written in the general matrix equation form of:

$$\begin{bmatrix} A & C \\ C^T & B \end{bmatrix} \left\{ \hat{\beta} \right\} = \left\{ \begin{array}{c} \mathbf{G} \\ \mathbf{H} \end{array} \right\}$$
(4.11)

where the form of matrices A, B and C and vectors **G** and **H** depend on the problem formulation. In fact, for formulations I and III, the matrix C vanishes. These matrices will be discussed in the following sections.

4.1.1 Formulation I - steady-state harmonic response

Equations (3.20) and (3.23) give the form of A^I , B^I , G^I and H^I for formulation I. Since τ^L is a linear operation and with w defined in (4.3), the linear stress operator becomes:

$$\tau_i^L(w) = \tau_i^L(U)\cos\omega t + \tau_i^L(V)\sin\omega t$$
(4.12)

where $\tau_i^L(w) = \tau^L(\varphi_i, w)$

Using (4.9), (4.12), the definition of Ω_x and the orthogonality of $\sin \omega t$ and $\cos \omega t$ over one period of oscillation gives

$$A_{ij}^{I} = G_{x}(G_{t}([\nabla \cdot \tau_{i}^{L}(w)] \cdot [\nabla \cdot \tau_{j}^{L}(w)]))$$

$$= \frac{\pi}{\omega} \int \left\{ \left[\nabla \cdot \tau_i^L(U) \right] \cdot \left[\nabla \cdot \tau_j^L(U) \right] + \left[\nabla \cdot \tau_i^L(V) \right] \cdot \left[\nabla \cdot \tau_j^L(V) \right] \right\} d \mathcal{D}(\mathbf{x})$$

$$= \frac{\pi}{\omega} \widehat{A} \frac{1}{ij} \qquad (4.13)$$

$$B_{ij}^{I} = C_{\mathbf{x}} \left(C_{t} \left(\left[\nabla \cdot \tau_{t}^{I}(\dot{w}) \right] \cdot \left[\nabla \cdot \tau_{j}^{I}(\dot{w}) \right] \right) \right) \right)$$

$$= \pi \omega \int \left\{ \left[\nabla \cdot \tau_{t}^{I}(U) \right] \cdot \left[\nabla \cdot \tau_{j}^{I}(U) \right] + \left[\nabla \cdot \tau_{t}^{I}(V) \right] \cdot \left[\nabla \cdot \tau_{j}^{I}(V) \right] \right\} d \, \, \&(\mathbf{x})$$

$$= \pi \omega \widehat{A} \frac{1}{ij} \qquad (4.14)$$

$$C_{i}^{I} = C_{\mathbf{x}} \left(C_{t} \left(\left[\nabla \cdot \tau^{N}(w, \dot{w}) \right] \cdot \left[\nabla \cdot \tau_{t}^{I}(w) \right] \right) \right) \right)$$

$$= \frac{\pi}{\omega} \int \left\{ \left[\nabla \cdot \mathbf{C}^{(1)}(U, V) \right] \cdot \left[\nabla \cdot \tau_{t}^{I}(U) \right] + \left[\nabla \cdot \mathbf{S}^{(1)}(U, V) \right] \cdot \left[\nabla \cdot \tau_{t}^{I}(V) \right] \right\} d \, \, \&(\mathbf{x})$$

$$= \frac{\pi}{\omega} \widehat{C} \frac{1}{i} \qquad (4.15)$$

$$H_{i}^{I} = C_{\mathbf{x}} \left(C_{t} \left(\left[\nabla \cdot \tau^{N}(w, \dot{w}) \right] \cdot \left[\nabla \cdot \tau_{t}^{I}(\dot{w}) \right] \right) \right)$$

$$= \pi \int \left\{ \left[\nabla \cdot \mathbf{C}^{(1)}(U, V) \right] \cdot \left[\nabla \cdot \tau_{t}^{I}(w) \right] - \left[\nabla \cdot \mathbf{S}^{(1)}(U, V) \right] \cdot \left[\nabla \cdot \tau_{t}^{I}(U) \right] \right\} d \, \, \&(\mathbf{x})$$

$$=\pi \hat{H} \, (4.16)$$

where

$$\mathbf{C}^{(1)}(U,V) = \frac{1}{\pi} \int_{0}^{2\pi} \tau^{N}(U\cos\vartheta, V\sin\vartheta) \,\cos\vartheta \,d\vartheta \tag{4.17}$$

$$\mathbf{S}^{(1)}(U,V) = \frac{1}{\pi} \int_{0}^{2\pi} \tau^{N}(U\cos\vartheta, V\sin\vartheta) \sin\vartheta d\vartheta$$
(4.18)

$$\vartheta = \omega t$$

The formulation l equations for $\widehat{\alpha}$ and $\widehat{\beta}$ can then be written as

50

$$\widehat{A}^{I}\widehat{\alpha} = \widehat{G}^{I} \tag{4.19}$$

$$\widehat{A}^{I}\widehat{\beta} = \frac{1}{\omega}\widehat{H}^{I} \tag{4.20}$$

A few observations are pertinent to the above expressions for $\hat{\alpha}$ and $\hat{\beta}$:

- 1. The solutions $\hat{\alpha}$ and $\hat{\beta}$ depend on the first Fourier coefficients of the nonlinearity τ^N as seen by equations (4.17) and (4.18). A number of other approximate methods including Galerkin's method (see equations (2.18f) and (2.18g)) also require the evaluation of such coefficients.
- 2. In solving for $\hat{\alpha}$ and $\hat{\beta}$, only one matrix \hat{A}^{I} needs to be computed and inverted. Thus a saving of computation effort has appeared for steady-state harmonic analysis.
- 3. With \hat{A}^{I} and \hat{H}^{I} not being explicit functions of ω , equation (4.20) shows that the effect of the equivalent linear damping, $\hat{\beta}$, will diminish with higher frequencies.

4.1.2 Formulation II - steady-state harmonic response

For the steady-state solutions of w given by (4.3), the stress and strain operations can be written as

$$\tau_i^L(w) = \tau_i^L(U)\cos\omega t + \tau_i^L(V)\sin\omega t \tag{4.21}$$

$$\mathbf{g}(w) = \mathbf{g}(U)\cos\omega t + \mathbf{g}(V)\sin\omega t \tag{4.22}$$

where $\tau_i^L(w) = \tau^L(\varphi_i, w)$. With the above relations, the general matrices and vectors for the formulaton II minimization take on the following special form for steady-state solutions:

$$A_{ij}^{II} = \int [\tau_i^L(U) \cdot \mathbb{S}(V) + \tau_i^L(V) \cdot \mathbb{S}(U)] [\tau_j^L(U) \cdot \mathbb{S}(V) + \tau_j^L(V) \cdot \mathbb{S}(U)]$$

+
$$[\tau_i^L(U) \cdot \mathbf{S}(U)][3\tau_j^L(U) \cdot \mathbf{S}(U) + \tau_j^L(V) \cdot \mathbf{S}(V)]$$

+
$$[\tau_i^L(V) \cdot \mathbf{S}(V)][\tau_j^L(U) \cdot \mathbf{S}(U) + 3\tau_j^L(V) \cdot \mathbf{S}(V)] d \mathcal{D}(\mathbf{x})$$

 $C_{ij}^{II} = \omega \int \left\{ - \left[\tau_i^L(U) \cdot \$\left(U\right) - \tau_i^L(V) \cdot \$\left(V\right) \right] \left[\tau_j^L(U) \cdot \$\left(V\right) + \tau_j^L(V) \cdot \$\left(U\right) \right] \right\}$

$$- \left[\tau_i^L(U) : \mathbf{S}(V)\right] \left[\tau_j^L(U) \cdot \mathbf{S}(U) + 3\tau_j^L(V) \cdot \mathbf{S}(V)\right]$$

+
$$[\tau_i^L(V) \cdot S(U)][3\tau_j^L(U) \cdot S(U) + \tau_j^L(V) \cdot S(V)]$$
 d $\mathcal{D}(\mathbf{x})$ (4.23b)

(4.23a)

$$D_{ij}^{II} = C_{ji}^{II} \tag{4.23c}$$

$$B_{ij}^{II} = \omega^2 \int \left\{ [\tau_i^L(U) \cdot \mathfrak{S}(U) - \tau_i^L(V) \cdot \mathfrak{S}(V)] [\tau_j^L(U) \cdot \mathfrak{S}(U) - \tau_j^L(V) \cdot \mathfrak{S}(V)] \right\}$$

+
$$[\tau_i^L(U) \cdot \mathbf{s}(V)][3\tau_j^L(U) \cdot \mathbf{s}(V) - \tau_j^L(V) \cdot \mathbf{s}(U)]$$

$$-\left[\tau_{i}^{L}(V)\cdot \mathbf{S}(U)\right]\left[\tau_{i}^{L}(U)\cdot \mathbf{S}(V)-\tau_{i}^{L}(V)\cdot \mathbf{S}(U)\right]\right]d\,\mathcal{Q}(\mathbf{x})$$

$$(4.23d)$$

$$\mathbf{C}^{(3)} = \frac{1}{\pi} \int_{0}^{2\pi} \tau^{N}(U, V, \vartheta) \cos 3\vartheta d\vartheta$$

$$\mathbf{S}^{(3)} = \frac{1}{\pi} \int_{0}^{2\pi} \tau^{N}(U, V, \vartheta) \sin 3\vartheta d\vartheta$$
(4.23h)

$$\mathbf{C}^{(3)} = \frac{1}{\pi} \int_{0}^{2\pi} \tau^{N}(U, V, \vartheta) \cos 3\vartheta d\,\vartheta \qquad (4.23g)$$
$$\mathbf{S}^{(3)} = \frac{1}{\pi} \int_{0}^{2\pi} \tau^{N}(U, V, \vartheta) \sin 3\vartheta d\,\vartheta \qquad (4.23h)$$

$$\mathbf{C}^{(3)} = \frac{1}{\pi} \int_{0}^{2\pi} \tau^{N}(U, V, \vartheta) \cos 3\vartheta d\,\vartheta \tag{4.23g}$$

(4.23f)

 $+ [\mathbf{C}^{(3)} \cdot \mathbb{S}(U) + \mathbf{S}^{(3)} \cdot \mathbb{S}(V)] [\tau_i^{\mathcal{L}}(U) \cdot \mathbb{S}(V) + \tau_i^{\mathcal{L}}(V) \cdot \mathbb{S}(U)]$

 $- \left[\mathbf{S}^{(1)} \cdot \mathbf{S}(V)\right] \left[3\tau_i^L(U) \cdot \mathbf{S}(V) - \tau_i^L(V) \cdot \mathbf{S}(U)\right] \ d\,\mathcal{D}(\mathbf{x})$

and $C^{(1)}$ and $S^{(1)}$ are as defined in (4.17) and (4.18).

where

+ $[(-\mathbf{C}^{(1)} + \mathbf{C}^{(3)}) \cdot \mathbb{S}(V) - (\mathbf{S}^{(1)} + \mathbf{S}^{(3)}) \cdot \mathbb{S}(U)][\tau_i^L(U) \cdot \mathbb{S}(U) - \tau_i^L(V) \cdot \mathbb{S}(V)]$

$$H_i^{II} = \omega \int -[\mathbb{C}^{(1)} \cdot \mathbb{S}(U)][\tau_i^L(U) \cdot \mathbb{S}(V) - 3\tau_i^L(V) \cdot \mathbb{S}(U)]$$

+
$$[\mathbf{S}^{(1)} \cdot \mathbf{S}(V)][\tau_i^{\mathcal{L}}(U) \cdot \mathbf{S}(U) + 3\tau_i^{\mathcal{L}}(V) \cdot \mathbf{S}(V)] d \mathcal{L}(\mathbf{x})$$
 (4.23e)

+
$$[\mathbf{C}^{(3)} \cdot \mathfrak{S}(U) + \mathbf{S}^{(3)} \cdot \mathfrak{S}(V)][\tau_i^{\perp}(U) \cdot \mathfrak{S}(U) - \tau_i^{\perp}(V) \cdot \mathfrak{S}(V)]$$

+
$$[(C^{(1)} + C^{(3)}) \cdot S(V) + (S^{(1)} + S^{(3)}) \cdot S(U)][\tau_i^{\perp}(U) \cdot S(V) + \tau_i^{\perp}(V) \cdot S(U)]$$

$$C_{i}^{II} = \int \left[\mathbf{C}^{(1)} \cdot \mathbf{S}(U) \right] \left[3\tau_{i}^{L}(U) \cdot \mathbf{S}(U) + \tau_{i}^{L}(V) \cdot \mathbf{S}(V) \right]$$

first Fourier coefficients $C^{(1)}$ and $S^{(1)}$, as did formulation I, but also the third coefficients $C^{(3)}$ and $S^{(3)}$. It appears then that this formulation extracts more information from the nonlinearity than formulation I.

4.1.3 Formulation III - steady-state harmonic response

Looking back at the general minimization relations for formulation I, (3.20) and (3.23), and for formulation III, (3.30), one can see that the matrices and vectors A^{III} , B^{III} , G^{III} and H^{III} are the same as those for formulation I if the $\nabla \cdot \tau^L$ operation is replaced by τ^L . Using this equivalence, the minimization relations for steady-state harmonic response with formulation III can be written directly using those of formulation I, equations (4.13) - (4.20). That is

$$\widehat{A}^{III}\widehat{a} = \widehat{G}^{III} \tag{4.24}$$

$$\widehat{A}^{III}\widehat{\beta} = \frac{1}{\omega}\widehat{H}^{III}$$
(4.25)

where

$$\widehat{A}_{ij}^{III} = \int [\tau_i^L(U) \cdot \tau_j^L(U) + \tau_i^L(V) \cdot \tau_j^L(V)] d \quad (\mathbf{x})$$

$$\widehat{G}_i^{III} = \int [\mathbf{C}^{(1)} \cdot \tau_i^L(U) + \mathbf{S}^{(1)} \cdot \tau_i^L(V)] d \quad (\mathbf{x}) \qquad (4.26)$$

$$\widehat{H}_i^{III} = \int [\mathbf{C}^{(1)} \cdot \tau_i^L(V) - \mathbf{S}^{(1)} \cdot \tau_i^L(U)] d \quad (\mathbf{x})$$

with $C^{(1)}$ and $S^{(1)}$ as defined in (4.17) and (4.18).

4.2 Stationary Random Response

For the case of g(t) in equation (4.1) being a stationary, Gaussian excitation, stationary solutions to (4.1) with zero mean value will be sought. Since the replacement auxiliary system is linear and the input is Gaussian, it is well known that the response w will also be Gaussian.

Because $w(\mathbf{x}, t)$ is a random process, the temporal averaging operator for this type of analysis must be a stochastic average. If the process is also *ergodic* the stochastic operator may be written as an ensemble average, or

$$G_t(G_x(\Delta \cdot \Delta)) = E[G_x(\Delta \cdot \Delta)]$$
(4.27)

where $E[\cdot]$ is the mathematical expectation operator.

Before the minimizaton relations are developed for stationary random response, the solution will be expanded in terms of deterministic spatial basis functions ψ_k and stochastic coefficients $d_k(t)$; k=1, 2, ..., N, as

$$w(\mathbf{x},t) = \sum_{k=1}^{N} \psi_k(\mathbf{x}) d_k(t)$$
(4.28)

If L and M represent linear operators on w, then, with (4.28), the following can be written:

$$E[\mathbf{L}(w)] = \sum_{k=1}^{N} \mathbf{L}(\psi_k) E[d_k]$$

$$E[\mathbf{L}(w) \cdot \mathbf{M}(w)] = \sum_{k,l=1}^{N} \mathbf{L}(\psi_k) \cdot \mathbf{M}(\psi_l) E[d_k d_l]$$

$$(4.29)$$

Recall that w has a zero mean in time. Therefore

$$E[w] = \sum_{k=1}^{N} \psi_k E[d_k] = 0$$

and since the ψ_k are linearly independent

$$E[d_k] = 0$$
; $k = 1, 2, ..., N$ (4.30)

The zero mean property of **d** along with the properties of stationarity and normality permit further statements to be made about certain averaged values of **d**. It can be shown that for a stationary random vector $\mathbf{d}(t)$ which is differentiable

$$E[\mathbf{d}\mathbf{d}^{T}] = -E[\mathbf{d}\mathbf{d}^{T}]$$

$$E[\mathbf{d}\mathbf{d}^{T}] = E[\mathbf{d}\mathbf{d}^{T}] = -E[\mathbf{d}\mathbf{d}^{T}]$$

$$(4.31)$$

The normality of **d** with (4.30) can be used to show that for a function q = q(y) (reference [1]):

$$E[\mathbf{y}q(\mathbf{y})] = E[\mathbf{y}\mathbf{y}^T]E[\nabla_{\mathbf{y}}q(\mathbf{y})]$$
(4.32)

where $\nabla_{\mathbf{y}} = \left[\frac{\partial}{\partial y_1}, \frac{\partial}{\partial y_2}, \cdots, \frac{\partial}{\partial y_N}\right]^T$ These properties can also be used to show that the higher order moments of **d** can be expressed in terms of the second order moments [30]. In particular,

$$E[d_1 d_2 d_3 d_4] = E[d_1 d_2] E[d_3 d_4] + E[d_1 d_3] E[d_2 d_4]$$
(4.33)
+ $E[d_1 d_4] E[d_2 d_3]$

Using the discretized form of $w(\mathbf{x},t)$ given by (4.28), the linear auxiliary equation (4.2) can be spatially discretized via a number of methods to produce a set of stochastic differential equations for **d** in the form of

$$\mathfrak{M}\mathbf{d} + \mathbb{C}(\boldsymbol{\beta})\mathbf{d} + \mathfrak{X}(\boldsymbol{\alpha})\mathbf{d} = \mathbf{f}\boldsymbol{g}(t) \tag{4.34}$$

where as before the matrices \mathcal{M} , C and \mathcal{X} and the vector **f** depend on the discretization method.

The relationships (4.31) for a stationary random vector permits the discretized auxiliary equation to be written in terms of its correlation matrices as (see reference [51]):

$$AX + XA^{T} = B$$

where

$$A = \begin{bmatrix} 0 & -I \\ m^{-1} & \kappa(\alpha) & m^{-1} & C(\beta) \end{bmatrix}$$
$$X = \begin{bmatrix} E[dd^{T}] & E[dd^{T}] \\ E[dd^{T}] & E[dd^{T}] \end{bmatrix}$$

$$B = \begin{bmatrix} 0 & E[df^{T}] \mathcal{M}^{-T} \\ \mathcal{M}^{-1}E[fd^{T}] & \mathcal{M}^{-1}E[fd^{T}] + E[df^{T}] \mathcal{M}^{-T} \end{bmatrix}$$

where 0 and I are the null and identity matrices, respectively.

If g(t) is white noise, i.e. having a constant spectral density, the matrices $E[\mathbf{fd}^T]$ and $E[\mathbf{fd}^T]$ on the right hand side of (4.35) can be written as:

$$E[\mathbf{fd}^{T}] = 0 \tag{4.36}$$
$$E[\mathbf{fd}^{T}] = \pi \ \mathcal{M}^{-1} W$$

where W is the spectral density of g(t).

Therefore the stationary random solutions to the linear auxiliary equations with Gaussian excitation can be written in terms of the correlation matrices $E[\mathbf{dd}^T]$, $E[\mathbf{dd}^T]$ and $E[\mathbf{dd}^T]$ where these matrices are related to the equivalent linear parameters α and β through (4.35). The minimization relationships will now be expressed in terms of these correlation matrices.

(4.35)

4.2.1 Formulation I - stationary random response

The coefficient matrix, A^{I} , for the equivalent linear stiffness parameters $\hat{\alpha}$ were given in equation (3.21) as

$$A_{ij}^{I} = \mathbb{G}_{tx}([\nabla \cdot \tau_{i}^{L}(w)] \cdot [\nabla \cdot \tau_{j}^{L}(w)])$$

where, again $\tau_i^L(w) = \tau^L(\varphi_i, w)$ and $\varphi_i(\mathbf{x})$ are the basis functions for α and β . With $w(\mathbf{x}, t)$ spatially discretized as

$$w(\mathbf{x},t) = \sum_{k=1}^{N} \psi_k(\mathbf{x}) d_k(t)$$

the divergence of the linear stress operator τ^L can be written as

$$\nabla \cdot \tau_i^L(w) = \nabla \cdot \sum_{k=1}^N \tau_i^L(\psi_k) d_k$$

Then for stationary random response $G_t() = E[]$ and

$$A_{ij}^{I} = \sum_{k,l=1}^{N} E_{kl} \int [\nabla \cdot \tau_{i}^{L}(\psi_{k})] \cdot [\nabla \cdot \tau_{j}^{L}(\psi_{l})] d \quad \mathcal{D}(\mathbf{x})$$

$$(4.37)$$

where $E_{kl} \equiv E[d_k d_l]$

Similarly the coefficient matrix for the damping parameters $\hat{\beta}$, B^{I} , along with the right hand side vectors G^{I} and H^{I} for both $\hat{\alpha}$ and $\hat{\beta}$ are:

$$B_{ij}^{l} = \sum_{k,l=1}^{N} E_{kl} \int [\nabla \cdot \tau_{i}^{L}(\psi_{k})] \cdot [\nabla \cdot \tau_{j}^{L}(\psi_{l})]$$
(4.38)

$$G_i^{I} = \sum_{k=1}^{N} \int E\left[(\nabla \cdot \tau^N(w, \dot{w})) \cdot (\nabla \cdot \tau_i^L(\psi_k)) d_k \right] d \, \mathcal{D}(\mathbf{x})$$
(4.39)

$$H_{i}^{I} = \sum_{k=1}^{N} \int E\left[\left(\nabla \cdot \tau^{N}(w, \dot{w})\right) \cdot \left(\nabla \cdot \tau_{i}^{L}(\psi_{k})\right) \dot{d}_{k}\right] d \mathcal{Q}(\mathbf{x})$$

$$(4.40)$$

where, for example, $E_{kl} \equiv E[d_k d_l]$.

From the property given in equation (4.32) for a Gaussian vector **d** with a zero mean

$$E\left[\left(\nabla \cdot \tau^{N}(w,\dot{w})\right) \cdot \left(\nabla \cdot \tau^{L}_{i}(\psi_{k})\right)d_{k}\right] = \sum_{l=1}^{N} E\left[\frac{\partial}{\partial d_{l}}\left(\nabla \cdot \tau^{N}(\psi, \mathbf{d}, \dot{\mathbf{d}})\right)\right] \cdot \left(\nabla \cdot \tau^{L}_{i}(\psi_{k})\right) E_{kl}$$

Therefore (4.39) can be written as

$$G_{i}^{I} = \sum_{k,l=1}^{N} E_{kl} \int E\left[\frac{\partial}{\partial d_{l}} (\nabla \cdot \tau^{N}(\psi, \mathbf{d}, \dot{\mathbf{d}}))\right] \cdot (\nabla \cdot \tau_{i}^{L}(\psi_{k})) d \quad \mathcal{D}(\mathbf{x})$$
(4.41)

and in a similar way (4.40) becomes

$$H_{i}^{I} = \sum_{k,l=1}^{N} E_{kl} \int E\left[\frac{\partial}{\partial d_{l}} (\nabla \cdot \tau^{N}(\psi, \mathbf{d}, \mathbf{d}))\right] \cdot (\nabla \cdot \tau_{i}^{L}(\psi_{k})) d \, \mathcal{D}(\mathbf{x})$$

$$(4.42)$$

Generally the nonlinearity $\nabla \cdot \tau^N$ can be written in a power series of **d** and **d** as

$$\nabla \cdot \tau^{N} = \sum_{\substack{m_1 m_2 \cdots \\ n_1 n_3 \cdots}}^{M} \mathcal{F}_{m_1 m_2 \cdots n_1 n_2} \cdots \langle \psi \rangle d_{m_1} d_{m_2} \cdots d_{n_1} d_{n_2} \cdots$$

which says that $E\left[\frac{\partial}{\partial d_l}(\nabla \cdot \tau^N)\right]$ and $E\left[\frac{\partial}{\partial \dot{d}_l}(\nabla \cdot \tau^N)\right]$ can be expressed in terms of the higher order moments of the jointly distributed Gaussian vectors **d** and **d**. From reference [30], it can be shown that such higher order moments of Gaussian processes can always be expressed in terms of second order moments, for example equation (4.33) of this chapter. Hence the formulation I minimization relationships for stationary, Gaussian response can be written in terms of the correlation matrices for the response of the linear auxiliary solutions.

4.2.2 Formulation II - stationary random response

Using a similar specialization as performed in the last section, the formulation II minimization for stationary random response can be written as:

$$A^{II}\widehat{\alpha} + C^{II}\widehat{\beta} = G^{II}$$

$$D^{II}\widehat{\alpha} + B^{II}\widehat{\beta} = H^{II}$$

where

$$\begin{split} A_{ij}^{II} &= \sum_{k,l,m,n=1}^{N} \Lambda_{klmn}^{ij} [E_{kl}E_{mn} + E_{km}E_{ln} + E_{kn}E_{lm}] \\ C_{ij}^{II} &= \sum_{k,l,m,n=1}^{N} \Lambda_{klmn}^{ij} [E_{kl}E_{mn} + E_{km}]E_{ln} + E_{kn}E_{lm}] \\ D_{ij}^{II} &= C_{ji}^{II} \\ B_{ij}^{II} &= \sum_{k,l,m,n=1}^{N} \Lambda_{klmn}^{ij} [E_{kl}E_{mn} + E_{km}E_{ln} + E_{kn}E_{kn}] \\ C_{i}^{II} &= \sum_{k,l,m,n=1}^{N} [E_{kl}E_{mn} + E_{km}E_{ln} + E_{kn}E_{lm}] \\ \int [E[\frac{\partial}{\partial d_l}\tau^N] \cdot \$[\psi_k]][\tau_i^I(\psi_m) \cdot [\$(\psi_n)]d \ \$(\mathbf{x}) \\ H_i^{II} &= \sum_{k,l,m,n=1}^{N} [E_{kl}E_{mn} + E_{km}E_{ln} + E_{kn}E_{lm}] \\ \int [E[\frac{\partial}{\partial d_l}\tau^N] \cdot \$[\psi_k]][\tau_i^I(\psi_m) \cdot [\$(\psi_n)]d \ \$(\mathbf{x}) \end{split}$$

$$\Lambda_{klmn}^{ij} = \int [\tau_i^L(\psi_k), \, \mathbf{S}(\psi_l)] [\tau_j^L(\psi_m); \, \mathbf{S}(\psi_l)] \, d \, \mathcal{Q}(\mathbf{x})$$

4.2.3 Formulation III - stationary random response

In section 4.1.3, the formulation III minimization equations for steady- state harmonic response were obtained from the corresponding formulation I relations by replacing $\nabla \cdot \tau^N$ and $\nabla \cdot \tau^L$ by τ^N and τ^L , respectively. Again this similarity between the two formulations is used to obtain the formulation III relations.

Replacing $\nabla \cdot \tau^N$ and $\nabla \cdot \tau^L$ in equations (4.37), (4.38), (4.41) and (4.42) by τ^N and τ^L gives

$$A^{III}\hat{\boldsymbol{\alpha}} = G^{III} \tag{4.43}$$

$$B^{III}\hat{\beta} = H^{III} \tag{4.44}$$

where

$$A_{ij}^{III} = \sum_{k,l=1}^{N} E_{kl} \int [\tau_i^L(\psi_k) \cdot \tau_j^L(\psi_l)] d \, \mathcal{S}(\mathbf{x})$$

$$B_{ij}^{III} = \sum_{k,l=1}^{N} E_{kl} \int [\tau_i^L(\psi_k) \cdot \tau_j^L(\psi_l)] d \, \mathcal{S}(\mathbf{x})$$

$$G_i^{III} = \sum_{k,l=1}^{N} E_{kl} \cdot \int E \left[\frac{\partial}{\partial d_l} \tau^N(\psi, \mathbf{d}, \mathbf{d})\right] \cdot \tau_i^L(\psi_k) d \, \mathcal{S}(\mathbf{x})$$

$$H_i^{III} = \sum_{k,l=1}^{N} E_{kl} \int E \left[\frac{\partial}{\partial d_l} \tau^N(\psi, \mathbf{d}, \mathbf{d})\right] \cdot \tau_i^L(\psi_k) d \, \mathcal{S}(\mathbf{x})$$

Remarks made in section 4.2.1 are herein applicable.

.

4.3 Discussion of Solution Procedures

This chapter has dealt with the development of two sets of equations describing the response of a nonlinear dynamic system via the equivalent linearization technique for the two special cases of steady-state periodic solutions and stationary random response. The first set of these equations arise from the minimization of a given system difference with respect to the equivalent linear stiffness and damping parameters α and β , respectively. The second set results from the spatial discretization of the auxiliary linear system. These two sets of equations are interrelated through α and β and the linear displacements w. This suggests that such equations will have to be solved in an iterative fashion. Techniques for solving different types of vibration problems described by the equations developed in this chapter will now be discussed.

4.3.1 Free vibration response

Consider the case of a nonlinear, conservative system excited only by prescribed initial displacements. Since the system is conservative, the response $W(\mathbf{x},t)$ will continue to oscillate at a frequency, say ω . Then the linear auxiliary response will be governed by the singular matrix equation

$$\left[-\omega^2 \ \mathcal{H} + \ \mathcal{K}(a, \hat{\alpha})\right] \mathbf{u} = \mathbf{0} \tag{4.46}$$

which is constrained by the minimization equations

$$A(\mathbf{u})\widehat{\boldsymbol{\alpha}} = \mathbf{G}(\mathbf{u}) \tag{4.47}$$

where

•
$$w(\mathbf{x},t) = (\sum_{i=1}^{N} \psi_k(\mathbf{x}) u_k) \cos \omega t$$

• \mathcal{M} , $\mathbb X$ are the mass and stiffness matrices, respectively, for the discretized auxiliary equation

- a is the stiffness parameter of the linearized equation
- $\alpha(\mathbf{x}) = \sum_{i=1}^{N} \varphi_i(\mathbf{x}) \widehat{\alpha}_k$ is the equivalent linear stiffness parameter
- $A(\mathbf{u})$ and $\mathbf{G}(\mathbf{u})$ depend on the problem formulation used

If the constraint between \hat{a} and \mathbf{u} in equation (4.47) is not imposed, equation (4.48) is simply a linear eigenvalue problem in ω and \mathbf{u} , and therefore the frequency of vibration is *independent* of the amplitude of the response. However the inclusion of this constraint introduces the well known dependence of frequency on the amplitude of vibration. To determine this dependence, an iterative scheme will be introduced.

Let ω_0 be the lowest value of ω which satisfies (4.46) with $\hat{a} = 0$ (i.e. the linearized problem) and e_0 the corresponding eigenvector. Let c be a measure of the amplitude of response, and define the first approximation of **u** be equal to

$$\mathbf{u}_1 = c \, \mathbf{e}_0 \, . \tag{4.48}$$

Using (4.48), the first approximation to \hat{a} can be obtained from (4.47)

$$\widehat{\alpha}_1 = A^{-1}(\mathbf{u}_1)\mathbf{G}(\mathbf{u}_1) \quad . \tag{4.49}$$

Substituting (4.49) into (4.46) gives the next approximation to ω :

 $\left[-\omega_{1}^{2} \mathcal{M} + \mathcal{K}\left(\mathbf{a}, \widehat{\boldsymbol{\alpha}}_{1}\right)\right] \mathbf{e}_{1} = \mathbf{0}$

This iteration continues with

$$\mathbf{u}_k = c \, \mathbf{e}_{k-1} \tag{4.50}$$

until the nonlinear frequency on the k^{th} step, ω_k , converges to prescribed accuracy or the mode shape \mathbf{e}_k satisfies some convergence criterion, where

$$\left[-\omega_{k}^{2} \mathcal{M} + \mathcal{K}\left(a, \widehat{\alpha}_{k}\right)\right] \mathbf{e}_{k} = \mathbf{0}$$

$$(4.51)$$

$$\widehat{\boldsymbol{\alpha}}_{k} = A^{-1}(\mathbf{u}_{k})\mathbf{G}(\mathbf{u}_{k}) \tag{4.52}$$

This method can be extended to higher modes of nonlinear vibration. As in the case of the first mode, the complete eigenvalue problem of (4.51) need not be solved. A number of approximate eigenvalue/eigenvector techniques allow for the extraction of the first few modes without having to solve for the higher ones.

4.3.2 Forced response to harmonic excitation

As described in section (4.1), the equivalent linear response $w(\mathbf{x},t)$ is given by the solutions to

$$\begin{bmatrix} \left[-\omega^{2} \, \mathcal{H} + h\left(\mathbf{a}, \widehat{\boldsymbol{\alpha}}\right)\right] & \left[\omega \, \mathbb{C}\left(b, \widehat{\boldsymbol{\beta}}\right)\right] \\ \left[-\omega \, \mathbb{C}\left(b, \widehat{\boldsymbol{\beta}}\right)\right] & \left[-\omega^{2} \, \mathcal{H} + h\left(\mathbf{a}, \widehat{\boldsymbol{\alpha}}\right)\right] \end{bmatrix} \begin{pmatrix} \mathbf{u} \\ \mathbf{v} \end{pmatrix} = \begin{pmatrix} \mathbf{f} \\ \mathbf{0} \end{pmatrix}$$
(4.53)
and constrained by

$$\begin{bmatrix} A(\mathbf{u}, \mathbf{v}) & C(\mathbf{u}, \mathbf{v}) \\ C^{T}(\mathbf{u}, \mathbf{v}) & B(\mathbf{u}, \mathbf{v}) \end{bmatrix} \begin{cases} \widehat{\alpha} \\ \widehat{\beta} \end{cases} = \begin{cases} G \\ H \end{cases}$$
(4.54)
(4.54)

where $w(\mathbf{x},t) = \psi^T \mathbf{u} \cos \omega t + \psi^T \mathbf{v} \sin \omega t$. The rest of the terms in (4.53) and (4.54) are as defined before.

A "secant" method" method might appear to be a logical way to solve equations (4.53) and (4.54). That is, for a given value of ω and an initial guess for **u** and **v**, $\hat{\alpha}$ and $\hat{\beta}$ are determined from (4.54). This approximation is then used to obtain a next approximation for **u** and **v** in (4.53) and so on until **u** and **v** converge according to a prescribed criterion. It has been found, however that such a scheme is unstable, especially for the case of small damping. In this case, the coefficient matrix for **u** and **v** in (4.53) can become nearly singular for ω near a natural frequency of the linearized response.

Equations (4.53) and (4.54) together are nonlinear in $\mathbf{u}, \mathbf{v}, \hat{\boldsymbol{\alpha}}$ and $\hat{\boldsymbol{\beta}}$. Hence
the well established Newton-Raphson method for nonlinear algebraic equations can be used to obtain a nonlinear frequency response curve. In the case where good initial guesses for $\mathbf{u}, \mathbf{v}, \hat{\boldsymbol{\alpha}}$ and $\hat{\boldsymbol{\beta}}$ are difficult to obtain, a hybrid technique such as the Levenberg/Marquardt method (see references [38] and [39]) might be used. This method, which combines features of both Newton's method and the method of steepest descent, has convergence properties less dependent on the initial guess.

If u and v are N - vectors and $\hat{\alpha}$ and $\hat{\beta}$ are M - vectors, solution via the above two methods will require the solution of a $2(N+M) \ge 2(N+M)$ system of equations a number of times for only one point on the response curve. For large problems, such a number of inversions may prove to be prohibitively expensive in computing a nonlinear frequency spectra. It is often the case that only the resonant response of say the first nonlinear mode of response is desired for a given applied load. A new method will now be presented for establishing a relationship between applied load in the forced problem and the response of the system under free vibration. This method will be restricted to systems with conservative nonlinearities.

In equation (4.53), let the excitation vector f on the right hand side be written as

$$\mathbf{f} = p_s \,\mathfrak{F} \tag{4.55}$$

where p_e can be thought of as a spatially constant applied pressure. Since (4.53) is a linear system, resonance occurs when the response is nearly 90° out of phase with the excitation (i.e. $\mathbf{u} \approx \mathbf{0}$) provided the frequencies are wellseparated and the damping is small. With $\hat{\boldsymbol{\beta}} = \mathbf{0}$, for a conservative nonlinearity in the original system, equations (4.53) and (4.54) reduce to

65

$$\omega C (b) \mathbf{v} = p_e \ \mathcal{F} \tag{4.56}$$

$$\left[-\omega^2 \mathcal{M} + \mathcal{K}(\mathbf{a}, \hat{\boldsymbol{\alpha}})\right] \mathbf{v} = \mathbf{0}$$
(4.57)

$$A(\mathbf{v})\hat{\boldsymbol{\alpha}} = \mathbf{G}(\mathbf{v}) \tag{4.58}$$

But equations (4.57) and (4.58) are simply the equivalent linear free vibration problem of (4.46) and (4.47). Premultiplying (4.56) by \mathbf{v}^T and solving for p_e gives

$$p_{s} = \omega \frac{\mathbf{v}^{T} C \mathbf{v}}{\mathbf{v}^{T} \mathbf{\mathcal{F}}}$$
(4.59)

Equation (4.59) can be thought of as a system power balance relationship at resonance. Here the input power $p_e(\omega \mathbf{v}^T)$ is balanced by the dissipated power $(\omega \mathbf{v}^T) C(\omega \mathbf{v})$.

Consequently, the damped resonant response can be obtained once the free vibration response is known. The steps for such a method are as follows: First the free vibration is solved for \mathbf{v} using (4.57) and (4.58) through the method in section 4.3.1. Then the applied load required for the system to resonate at ω with displacement \mathbf{v} and damping given by C can be found by using equation (4.59). This will be considerably more cost efficient than solving equations (4.53) and (4.54) directly.

4.3.3 Forced response to stationary random excitation

The equivalent linear equations for stationary random response to Gaussian excitation were written in the form of equation (4.35) and the minimization relations developed in sections 4.2.1 - 4.2.3. These equations can be solved iteratively by a secant method such as the one described in section 4.3.2. The auxiliary system equation of (4.35) is the familiar Liapunov matrix equation which often arises in stability theory of linear systems. References [25] and [35] discuss some computationally efficient algorithms for solving Liapunov's equation.

V. EXAMPLE STUDIES

5.1 Response of a One-Dimensional Yielding Continuum

Many tall structures can be modeled by a continuous shear beam where the base of the beam is excited by a prescribed function of time. Such a system is shown in figure 5.1.1. For large amplitude of response, the need may arise for a mathematical model which includes nonlinearities due to geometric considerations and/or the behavior of the material.

Consider the case of nonlinear material behavior in the beam. Let the spatial domain $\mathfrak{O}(x)$ of the beam be $x \in [0, 1]$ and the boundary $\partial \mathfrak{O}(x)$ be at x=0 and x=1. If the bending effects are negligible compared to the effects of shearing strains, the equation of motion and boundary conditions for the harmonically excited beam of figure 5.1.1 may be written as

$$-\ddot{w} + \frac{\partial}{\partial x} \left[a(x) \frac{\partial w}{\partial x} \right] + \frac{\partial}{\partial x} \left[\tau^{N}(w, \dot{w}) \right] = r \cos \omega t$$
(5.1.1)

$$w(0,t) = 0 \tag{5.1.2}$$

$$\frac{\partial w}{\partial x}(1,t) + \tau^{N}(w(1,t), \dot{w}(1,t)) = 0$$
(5.1.3)

in which w = w(x,t) is the deflection of the beam from its unstrained equilibrium position at a distance x from the base. τ^N is a nonlinear stress function/functional of the strain and possibly strain rate, a(x) is the linear stiffness and r is the amplitude of the excitation. The boundary conditons in equations (5.1.2) and (5.1.3) are those of a beam fixed at the base and stress free at the top.

The linear auxiliary equation to be used in solving (5.1.1) - (5.1.3) is written in the following form:

67



Figure 5.1.1 Shear Structure with Harmonic Base Excitation

$$-\ddot{w} + \frac{\partial}{\partial x} \left[a(x) \frac{\partial w}{\partial x} \right] + \frac{\partial}{\partial x} \left[\alpha(x) \frac{\partial w}{\partial x} \right] + \frac{\partial}{\partial x} \left[\beta(x) \frac{\partial \dot{w}}{\partial x} \right] = r \cos \omega t \qquad (5.1.4)$$

where $\alpha(x)$ and $\beta(x)$ are the equivalent linear stiffness and damping parameters, respectively. The other terms in (5.1.4) are as defined for (5.1.1) and the boundary conditions of (5.1.2) and (5.1.3) also apply to (5.1.4).

It should be noted that equations (5.1.1) and (5.1.4) are simply special cases of the nonlinear and linear auxiliary systems of equation (2.1) and (3.1). The positive definiteness and self-adjointness of $\frac{\partial \tau^L}{\partial x} = \frac{\partial}{\partial x} [a(x) \frac{\partial w}{\partial x}]$ can be shown using the boundary conditions stated in (5.2) and (5.3). The stress operations, τ^L and τ^N , and the displacements are scalar quantities, which implies that the dot product defined in Chapter II is simply a scalar multiplication.

The equivalent linearization approach developed in Chapters III and IV can be applied directly to this problem. The use of problem formulation III for this nonlinear system will be detailed in the following sections.

5.1.1 Minimization relations and discretization of the auxiliary equations

The relationships for $\alpha(x)$ and $\beta(x)$ which minimize the system difference term for steady state harmonic solutions in formulation III were developed in section 4.1.3 and are given by equations (4.24) - (4.26). Recall that for the onedimensional continuum model of equation (5.1.1), the linear stress operation $\tau^{L}(\varphi_{i}, w)$ is given by

$$\tau^{L}(\varphi_{i},w) = \varphi_{i}(x)\frac{\partial w}{\partial x}$$
(5.1.5)

where φ_i ; i=1, 2, ..., M, are the linear independent basis functions for the equivalent linear parameters $\alpha(x)$ and $\beta(x)$, as specified in equations (3.12) and (3.13).

Let the spatial domain of the beam be subdivided into "elements"

 $\mathfrak{D}_e(x): x \in (\frac{e-1}{M}, \frac{e}{M})$ for e=1, 2, ..., M. Furthermore let the basis functions be defined as

$$\varphi_{e}(x) = \begin{cases} 1 & \text{if } x \in \mathcal{D}_{e} \\ 0 & \text{if } x \notin \mathcal{D}_{e} \end{cases}$$

(5.1.6)

Therefore the basis functions are defined in such a way that $\alpha(x)$ and $\beta(x)$ are constant within each element but discontinuous across the element boundaries.

Substituting (5.1.5) and (5.1.6) into (4.24) - (4.26) gives the following explicit relations for the elemental equivalent linear parameters of formulation III in terms of the equivalent linear solution w(x,t):

$$\widehat{\alpha}_{i} = \frac{\int [C(U,V)U_{,x} + S(U,V)V_{,x}] dx}{\int [U_{,x}^{2} + V_{,x}^{2}] dx}$$
(5.1.7)

$$\widehat{\beta}_{i} = \frac{\int [C(U,V)V_{,x} - S(U,V)U_{,x}] dx}{\omega \int [U_{,x}^{2} + V_{,x}^{2}] dx}$$
(5.1.8)

where

$$w(x,t) = U(x)\cos\omega t + V(x)\sin\omega t$$

$$U_{,x}, V_{,x} = \frac{\partial U}{\partial x}, \frac{\partial V}{\partial x}$$

$$C(U,V) = \frac{1}{\pi} \int_0^{2\pi} \tau^N(U,V,\vartheta) \cos\vartheta d\vartheta$$

$$S(U,V) = \frac{1}{\pi} \int_0^{2\pi} \tau^N(U,V,\vartheta) \sin\vartheta d\vartheta$$

 $\vartheta = \omega t$

Let $\psi_i(x)$ and λ_i^2 ; i=1, 2, ... be the eigenfunctions and eigenvalues of the linearized form of equation (5.1.1). The auxiliary linear system will now be discretized in terms of the N eigenfunctions; that is

$$U(x) = \sum_{i=1}^{N} \psi_i(x) u_i = \psi^T \mathbf{u}$$
(5.1.9)

$$V(x) = \sum_{i=1}^{N} \psi_i(x) v_i = \psi^T \mathbf{v}$$
 (5.1.10)

Substituting (5.1.9) and (5.1.10) into (5.1.4), premultiplying by $\psi(x)$, integrating over $\vartheta(x)$ using orthogonality properties of $\psi(x)$, and equating coefficients of $\cos \omega t$ and $\sin \omega t$ gives the modally discretized form of the linear auxiliary equation as

$$\mathbf{A}\mathbf{u} + \mathbf{A}\mathbf{u} + \mathbf{B}\mathbf{v} = r\mathbf{f} \tag{5.1.11}$$

$$\mathbf{A}\mathbf{v} + \mathbf{A}\mathbf{v} - \mathbf{B}\mathbf{u} = \mathbf{0} \tag{5.1.12}$$

where

$$\Lambda = [\Lambda_i] \tag{5.1.13}$$

$$\Lambda_i = \lambda_i^2 - \omega^2 \tag{5.1.14}$$

$$A = \sum_{k=1}^{M} \widehat{\alpha}_{k} \int_{\mathcal{D}_{k}} \Psi \Psi_{xx}^{T} dx \qquad (5.1.15)$$

$$B = \omega \sum_{k=1}^{M} \widehat{\beta}_{k} \int_{\mathcal{D}_{k}} \psi \psi_{,xx}^{T} dx \qquad (5.1.16)$$

$$\mathbf{f} = \int \boldsymbol{\psi} d \, \boldsymbol{\vartheta} \, (\mathbf{x}) \tag{5.1.17}$$

$$\Psi_{xx} = \left[\frac{\partial^2 \psi_1}{\partial x^2}, \cdots, \frac{\partial^2 \psi_N}{\partial x^2}\right]^T$$
(5.1.18)

It will be assumed that the linearized form of equation (5.1.1) has spatially constant stiffness properties, i.e. $a \neq a(x)$. Then its eigenvalues and normalized eigenfunctions take on the form:

$$\lambda_{i} = \frac{\pi}{2} (2i-1)\sqrt{a}$$

$$\psi_{i}(x) = \sqrt{2} \sin \frac{\pi}{2} (2i-1)x$$
(5.1.19)

for i=1,2,... The spatial basis functions, ψ , used in all further analysis will be written as in(5.1.19) unless otherwise stated.

The solution of the equivalent linearized problem is the solution to the set of equations given in (5.1.7), (5.1.8), (5.1.11) and (5.1.12). These can be solved once the nonlinear stress functional τ^N is specified. In the next section, a form of τ^N will be introduced to model a yielding material behavior in the one-dimensional beam.

5.1.2 Specification of the yielding model

In reference [24], a model for yielding behavior in one dimension has been presented which consists of an infinite collection of ideal elastic-plastic elements with continuous distributed yield levels. The same author, in reference [23], extended this yielding model to cyclic loading for a yielding level distribution which is constant for strains up to a level, say $W_{,x}^{u}$ and zero for strains levels above $W_{,x}^{u}$ where

$$W_{x}^{u} = \frac{1}{\sqrt{3\mu}}$$
(5.1.20)

and μ is a constant specifying the nonlinearity of the material. The resulting cyclic stress functional, as given in [23], is an integral relationship in terms of the strain level $W_{,x}$. This stress-strain relationship is shown in figure 5.1.2 for $\mu = 0.2$.

A solution of the nonlinear vibration problem using the above yielding model has been obtained in reference [23] by the application of Galerkin's method. In developing a solution, the same Fourier coefficients C(U,V) and S(U,V) as in the



equivalent linear parameter relations in (5.1.7) and (5.1.8) of this chapter were evaluated as

$$C(U_{,x}, V_{,x}) = -\frac{15}{16} \mu \left[U_{,x}^2 + V_{,x}^2 \right]^{\frac{3}{2}}$$

$$S(U_{,x}, V_{,x}) = -\frac{2}{\pi} \mu \left[U_{,x}^2 + V_{,x}^2 \right]^{\frac{3}{2}}$$
(5.1.21)

for $W_{,x} \leq W_{,x}^{u}$ and where $W_{,x} = \left[U_{,x}^{2} + V_{,x}^{2}\right]^{\frac{1}{2}}$. The relationships for C and S become more complicated when $W_{,x} > W_{,x}^{u}$. Therfore, as in reference [23], only strain levels satisfying $W_{,x} \leq W_{,x}^{u}$ will be considered herein.

Substituting (5.1.21) into (5.1.7) and (5.1.8) with the use of the expansions of U(x) and V(x) in (5.1.9) and (5.1.10) give the following relations for the discretized equivalent linear parameters:

$$\widehat{\alpha}_{i} = \frac{-\mu \sum_{m=1}^{N} \left(\frac{15}{16} u_{m} + \frac{2}{\pi} v_{m}\right) \, h_{m}^{(i)}}{\varsigma^{(i)}}$$

$$\widehat{\beta}_{i} = \frac{-\frac{\mu}{\omega} \sum_{m=1}^{N} \left(\frac{15}{16} v_{m} - \frac{2}{\pi} u_{m}\right) \, h_{m}^{(i)}}{\varsigma^{(i)}}$$
(5.1.22)

where

$$\mathcal{H}_{m}^{(i)} = \sum_{k,l=1}^{N} (u_{k}u_{l} + v_{k}v_{l})^{\frac{3}{2}} \int_{\mathcal{D}_{i}} (\psi_{k,x}\psi_{l,x})^{\frac{3}{2}} \psi_{m,x} dx$$

$$\mathcal{L}^{(i)} = \sum_{k,l=1}^{N} (u_k u_l + v_k v_l) \int_{\mathfrak{O}_i} \psi_{k,x} \psi_{l,x} \, dx$$

$$\psi_{k,x}=\frac{\partial\psi_k}{\partial x}$$

5.1.3 Convergence of equivalent linear parameters

Recall from Chapter III that the equations for the discretized equivalent parameters $\hat{\alpha}$ and $\hat{\beta}$ were established by an approximation to the exact minimization in equation (3.9). The question may arise as to how well the approximations

$$\alpha(x) \approx \sum_{i=1}^{M} \varphi_i(x) \widehat{\alpha}_i$$
 (5.1.24)

$$\beta(x) \approx \sum_{i=1}^{M} \varphi_i(x) \hat{\beta}_i$$
 (5.1.25)

compare with the values of α and β that satisfy (3.9). In this section, it will be shown for the basis functions φ_i ; i=1, 2, ..., M, defined in (5.1.6) how the approximate values for α and β approach their exact counterparts as $M \rightarrow \infty$. For the special yielding nonlinearity specified in the last section, the convergence rate will also be given.

The formulation III stiffness parameters $\hat{\alpha}_i$ for the one-dimensional problem of this chapter were shown to be (equation (5.7)):

$$\widehat{\alpha}_{i} = \frac{\int f(x) dx}{\int g(x) dx}$$
(5.1.26)

where

$$f(x) = C(U,V)U_{,x} + S(U,V)V_{,x}$$
$$g(x) = U_{,x}^{2} + V_{,x}^{2}$$

If the minimization of (3.9) is carried out *exactly* for the formulation III equation difference, it can be shown that $\alpha(x)$ must satisfy the relationship:

$$\alpha(x) = \frac{f(x)}{g(x)}$$
(5.1.27)

Recall that the displacement functions U(x) and V(x) were expanded in terms of the basis functions $\psi_i(x)$; i=1, 2, ..., N, as shown in equations (5.1.9) and (5.1.10). Suppose that the functions $\psi_i(x)$ are such that f(x) and g(x) are continuous except at say the boundaries of $\vartheta_i(x)$, where they may be discontinuous but finite. These types of discontinuities may arise if, for example, the functions $\psi_i(x)$ are finite element shape functions and f(x) and g(x) are differential functions of $\psi_i(x)$. Let x_i ; i=1, 2, ..., M, be the equally spaced coordinates of the possible discontinuities. Then α at the discontinuities will be defined to be

$$\alpha(x_i) = \frac{1}{2} \left[\alpha(x_i^{-}) + \alpha(x_i^{+}) \right]$$
(5.1.28)

where $x_i^- = \lim_{\varepsilon \to 0} (x_i - \varepsilon)$ and $x_i^+ = \lim_{\varepsilon \to 0} (x_i + \varepsilon)$.

Since f(x) and g(x) are continuous on $x \in (x_i, x_{i+1})$, the mean value theorem can be applied to $\hat{\alpha}_i$ in (5.1.26) to show that

$$\lim_{h \to 0} \hat{\alpha}_i = \frac{f(x_i^+)}{g(x_i^+)}$$
(5.1.29)

and

$$\lim_{h \to 0} \hat{\alpha}_{i-1} = \frac{f(x_i^-)}{g(x_i^-)}$$
(5.1.30)

where $h = x_{i+1} - x_i = x_i - x_{i-1} = \frac{1}{M}$. Therefore, from (5.1.28) - (5.1.30)

$$\lim_{h \to 0} \frac{1}{2} \left[\widehat{\alpha}_{i} + \widehat{\alpha}_{i-1} \right] = \frac{1}{2} \left[\frac{f(x_{i}^{+})}{g(x_{i}^{+})} + \frac{f(x_{i}^{-})}{g(x_{i}^{-})} \right]$$
$$= \alpha(x_{i})$$
(5.1.31)

Equation (5.1.31) says that the approximate equivalent linear stiffness

parameters $\hat{\alpha}$ converge as $h \to 0$ to their exact counterpart α in an "average" sense across a discontinuity of the functions f(x) and g(x). If no discontinuities exist, it is clear that $f(x_i^+) = f(x_i^-)$ and $g(x_i^+) = g(x_i^-)$, implying that $\hat{\alpha}_i$ converges exactly to $\alpha(x_i)$ or

$$\lim_{h\to 0} \widehat{\alpha}_i = \alpha(x_i)$$

Similar statements can be made about the convergence of \widehat{eta}_i to $eta(x_i)$

Consider now the equivalent linear parameter relations (5.1.22) and (5.1.23) for the yielding nonlinearity, with U(x) and V(x) expanded in terms of only one mode of the linearized problem. That is, from (5.1.9), (5.1.10) and (5.1.19), U and V are written as

$$U(x) = \left[\sqrt{2}\sin\frac{\pi}{2}x\right]u_1$$

$$V(x) = \left[\sqrt{2}\sin\frac{\pi}{2}x\right]v_1$$

The equivalent linear parameters then reduce to

$$\widehat{\alpha}_{i} = c_{a} \frac{\int \cos^{4} \frac{\pi}{2} x \, dx}{\int \cos^{2} \frac{\pi}{2} x \, dx}$$

$$\widehat{\beta}_{i} = c_{b} \frac{\int \cos^{4} \frac{\pi}{2} x \, dx}{\int \cos^{2} \frac{\pi}{2} x \, dx}$$
(5.1.32)
$$(5.1.33)$$

where

$$c_{a} = -2\mu \left(\frac{\pi}{2}\right)^{2} \left(u_{1}^{2} + v_{1}^{2}\right)^{\frac{1}{2}} \left[\frac{15}{16}u_{1} + \frac{2}{\pi}v_{1}\right]$$

$$c_{b} = -2\frac{\mu}{\omega} \left(\frac{\pi}{2}\right)^{2} \left(u_{1}^{2} + v_{1}^{2}\right)^{\frac{1}{2}} \left[\frac{15}{16}v_{1} - \frac{2}{\pi}u_{1}\right]$$

The exact minimization of the formulation III equation difference with respect to α and β gives:

$$a(x) = c_a \cos^2 \frac{\pi}{2} x \tag{5.1.34}$$

$$\beta(x) = c_b \cos^2 \frac{\pi}{2} x \tag{5.1.35}$$

Let e be defined as the mean square value of the error between $\alpha(x)$ and its approximation $\alpha(x) \approx \sum_{i=1}^{M} \varphi_i(x) \hat{\alpha}_i$, or

$$e = \int_0^1 [\alpha(x) - \sum_{i=1}^M \varphi_i(x) \widehat{\alpha}_i]^2 dx$$

 $= c_a^2 E_M$

where
$$E_{\mathbf{M}} = \sum_{i=1}^{\mathbf{M}} \int_{0}^{1} \left[\cos^{2}\frac{\pi}{2}x - \frac{\int_{0}^{1} \cos^{4}\frac{\pi}{2}x dx}{\int_{0}^{1} \cos^{2}\frac{\pi}{2}x dx}\right]^{2} dx$$

The rate of convergence of $\sum_{i=1}^{M} \varphi_i(x) \widehat{\alpha}_i \to \alpha$ will be determined by investigating

how $E_M \to 0$ as $M \to \infty$. Figure 5.1.3 shows the relationship of $\log(\frac{1}{M})$ vs. $\log E_M$. For large values of M, the slope of this curve is about 2; therefore, the error decreases at a nearly quadratic rate as $\frac{1}{M} \to 0$.

From (5.1.33) and (5.1.35) it is clear that the convergence of $\sum_{i=1}^{M} \varphi_i(x) \hat{\beta}_i \to \beta(x)$ as $M \to \infty$ is governed by the same error function E_M . Therefore the approximate equivalent linear damping parameters also converge at a quadratic rate.



Figure 5.1.3 Convergence Rate for Equivalent Linear Parameters

5.1.4 Numerical results

In this section, a one term expansion on w(x,t) will be used to calculate the response of the yielding shear beam by means of the equivalent linearization technique and Galerkin's method. The equivalent linearization equations to be solved are equations (5.1.11), (5.1.12), (5.1.22) and (5.1.23) with N = 1 while the Galerkin equations can be found in reference [22].

Figure 5.1.4 shows the base shear strain ratio $\frac{W_{,x}}{W_{,x}^{u}}$ obtained from the equivalent linearization method with M = 10 and Galerkin's method in terms of the dimensionless excitation frequency Ω , where

$$\Omega = \frac{\omega}{\lambda_1} \tag{5.1.36}$$

These response curves are shown on a somewhat expanded scale in order that the characteristics of the two types of solutions can be seen in detail. As can be seen, with M = 10 the equivalent linearization technique predicts a slightly lower level of response and lower resonant frequency than Galerkin's method. This difference is more pronounced for larger levels of excitation R where

$$R \equiv \frac{r}{\lambda_1^2} \tag{5.1.37}$$

Figure 5.1.5 shows the effect of mesh subdivision for the equivalent linear parameters on the peak response of the beam. In the last section, it was shown that as $M \to \infty$, the equivalent linear parameters obtained by the approximate minimization technique converged to their counterparts obtained by the exact minimization technique. An interesting result is that when the exact linear parameters are used in the one term expansion of the auxiliary linear equations, the equivalent linear solutions are exactly the same as the one term Galerkin solution. This is displayed in figure 5.1.5 by the coincidence of the







 $M \rightarrow \infty$ equivalent linearization curve with the Galerkin response. Therefore, for this special case, there appears to be an equivalence between the two methods.

5.2 Steady-State Harmonic Response of a Nonlinear Plate

A typical assumption used in the vibration of thin plates governed by the Poisson-Kirchoff theory is that the amplitude of response is small in comparison to the thickness of the plate. This assumption permits the use of a linear equation of motion for describing the response of the plate. Whenever the deflection becomes of the same order of magnitude as the thickness, geometric nonlinearities arise from the coupling of the membrane and bending stresses. In this case a linear model of the response of the plate would not be appropriate.

The nonlinear theory used in this example is based on the so-called Berger approximation to the coupled nonlinear Von Karman equations for static analysis. Berger's analysis, which originally appeared in reference [2], is essentially based upon neglecting the strain energy due to the second invariant of the middle surface. This assumption permits the nonlinear equations for the inplane and transverse displacements to become uncoupled. The work presented in reference [36] extended the Berger approximation to the dynamic equations of motion.

5.2.1 Equations of motion

The nonlinear equations of motion can be derived through the application of the generalized Hamilton's principle (see section 2.4.2) which states that

$$\delta \int_{t_1}^{t_2} (T + W - U) dt = 0$$
 (5.2.1)

where T, U, and W are the kinetic energy, potential energy and non-conservative virtual work of the system, respectively. Let the spatial domain of the plate be defined for rectangular coordinates $\mathbf{x} = (x_1, x_2)$ on $\mathfrak{L}(\mathbf{x})$ and the boundary $\partial \mathfrak{L}(\mathbf{x})$. Then if the in-plane kinetic energy is negligible compared to that due to transverse motion, the kinetic energy can be written as:

$$T = \frac{1}{2} \int \rho \dot{w}^2 d \quad \mathfrak{Q}(\mathbf{x})$$
(5.2.2)

and the work done by an externally applied, harmonic load is

$$W = -\left[\int p(\mathbf{x})w \, d \, \vartheta(\mathbf{x})\right] \cos \omega t \tag{5.2.3}$$

where $w = w(\mathbf{x}, t)$ is the transverse displacement of the plate for which $\dot{w} = \frac{\partial w}{\partial t}$, ρ is the mass per unit area and $p(\mathbf{x})$ is the distributed applied load. From reference [2], the potential energy of the plate which neglects the second invariant of strain energy can be written for a non-homogeneous elastic material as:

$$U = \frac{1}{2} \int_{\mathcal{L}} \left[M_{\gamma\delta}(D, w) w_{\gamma\delta} + \frac{12D}{h^2} e^2 \right] d \mathcal{D}(\mathbf{x})$$
(5.2.4)

where the summation convention is implied on γ , $\delta = 1, 2$ and

$$M_{11} = D \ (w_{11} + \nu w_{22})$$

 $M_{12} = D (1-\nu) w_{12} = M_{21}$

$$M_{22} = D \left(\nu w_{11} + w_{22} \right)$$

$$e = u_{1} + v_{2} + \frac{1}{2}w_{a}w_{a}$$

= the first invariant of the strains at the midplane surface

u, v = in-plane displacements of the plate

in the x_1 and x_2 directions, respectively

 ν = Poisson's ratio for the material

h =thickness of the plate

$$D = \frac{Eh^3}{12(1-\nu^2)} = \text{flexural rigidity of the plate}$$

E =modulus of elasticity for the material

$$w_{\gamma} = \frac{\partial w}{\partial x_{\gamma}}$$

Using the energy relations (5.2.2) - (5.2.4) in the variational equation (5.2.1)and integrating (5.2.1) by parts produces the following set of equations of motion for the plate:

$$\left(\frac{12D}{h^2}e\right)_{,\gamma} = 0$$
 for $\gamma = 1, 2$ (5.2.5)

$$M_{\gamma\delta,\gamma\delta}(D,w) - \left(\frac{12D}{h^2}ew_{\gamma}\right)_{\gamma} + \rho\ddot{w} = p(\mathbf{x})\cos\omega t$$
(5.2.6)

Equation (5.2.5) states that the quantity $\frac{12D}{h^2}e$ is not a function of the spatial coordinates. Therefore, the integration of $\frac{12D}{h^2}e$ over the domain $\mathfrak{L}(\mathbf{x})$ gives:

$$\int_{\mathcal{Q}} \frac{12D}{h^2} e \, d \quad \mathcal{Q}(\mathbf{x}) = \frac{12D}{h^2} e \int_{\mathcal{Q}} d \quad \mathcal{Q}(\mathbf{x})$$
$$= \frac{12D}{h^2} e \left(Area \right)$$

ог

$$\frac{12D}{h^2}e = \frac{N(w)}{(Area)}$$
(5.2.8)

where Area is the surface area of the plate and

$$N(w) = \int_{\mathcal{D}} \frac{12D}{h^2} e \ d \ \mathcal{D}(\mathbf{x})$$
(5.2.9)

It should be noted that if $D \neq D(\mathbf{x})$, $h \neq h(\mathbf{x})$, and if the inplane displacements vanish at the boundaries of the plate, N(w) becomes a functional involving only the transverse displacements, or

$$N(w) = \frac{12D}{h^2} \int e \ d \ \mathcal{D}(\mathbf{x})$$
$$= \frac{6D}{h^2} \int w_{\gamma} w_{\gamma} \ d \ \mathcal{D}(\mathbf{x})$$
(5.2.9a)

Substitution of (5.2.8) with (5.2.9a) into equation of motion (5.2.6) gives

$$M_{\gamma\delta,\gamma\delta}(D,w) - \mu \left[\int_{\mathcal{Q}} w_{\gamma}w_{\gamma} d \mathcal{Q}(\mathbf{x}) \right] w_{\gamma\gamma} + \rho \dot{w} = p(\mathbf{x}) \cos \omega t \qquad (5.2.10)$$

where $\mu \equiv \frac{6D}{(Area)h^2}$.

Equation (5.2.10) is the dynamic analog to the uncoupled form of Berger's equations discussed in reference [2]. The damped response of the plate will be given by the solution to (5.2.10) with the addition of an absolute, or 'mass proportional', damping term $bw(\mathbf{x},t)$ where $b = b(\mathbf{x})$.

5.2.2 Equivalent linear system

The linear auxiliary equation of motion which serves as a replacement for the damped form of the nonlinear equation of motion (5.2.10) is defined to be:

$$M_{\gamma\delta,\gamma\delta}(D,w) + M_{\gamma\delta,\gamma\delta}(\alpha,w) + b\dot{w} + \rho\dot{w} = p\cos\omega t$$
(5.2.11)

where $\alpha = \alpha(\mathbf{x})$ is the equivalent linear flexural rigidity of the plate and all other terms are as defined before.

Referring to the notation used in chapters II and III, it can be seen by inspection that the stress operators for the nonlinear and linear auxiliary plate equations are are vector operations. Specifically τ^N and τ^L can be written as

$$\tau^{N} = (\tau_{1}^{N}, \tau_{2}^{N})^{T}$$

$$\tau^{L} = (\tau_{1}^{L}, \tau_{2}^{L})^{T}$$
(5.2.12)

where

$$\tau_{\gamma}^{N} = -\mu w_{,\gamma} \int w_{,\delta} w_{,\delta} d \mathcal{D}(\mathbf{x})$$

$$\tau_{\gamma}^{L} = M_{\gamma\delta,\delta}$$
(5.2.13)

for $\gamma = 1, 2$.

Recall that problem formulation II defined in Chapter III required the use of a strain-type function g(w). For the analysis of the plate response using formulation II. g(w) will be a vector operation defined as

$$\mathbf{S} = (S_1, S_2)^T \tag{5.2.14}$$

where

$$s_{\gamma}(w) = w_{\gamma}$$
 (5.2.15)

The nonlinearity in the equation of motion (5.2.10) influences the stiffness of the plate and not the energy dissipation. It should therefore be expected that the inclusion of the equivalent linear damping parameter β will not be necessary. By not including β in the minimization relations, the approximation for the equivalent linear flexural rigidity α

$$\alpha(\mathbf{x}) \approx \sum_{i=1}^{\underline{M}} \varphi_i(\mathbf{x}) \widehat{\alpha}_i$$

must satisfy the following minimization relationship:

$$A\,\hat{\alpha} = \mathbf{G} \tag{5.2.16}$$

where $\hat{\alpha} = (\hat{\alpha}_1, \hat{\alpha}_2, \dots, \hat{\alpha}_M)^T$. The form of the matrix A and vector **G** will depend on the problem formulation used and are given in section 4.1 for the

steady-state response of the general system. It should be observed that in section 3.8 the matrix A was shown to be invertible for all three problem formulations, and henceforth unique solution for $\hat{\alpha}$ and $\hat{\beta}$ always exist.

Let attention now be focused on developing the specific form of A and G in (5.2.16) for problem formulations I, II and III. The linear operations τ^L , $\nabla \cdot \tau^L$ and from (5.2.13) and (5.2.15) are expressed as:

$$\tau_{\gamma}^{L}(\varphi_{i},w) = M_{\gamma a,\delta}^{(i)}(U) \cos \omega t + M_{\gamma a,\delta}^{(i)}(V) \sin \omega t$$
(5.2.17)

$$\nabla \cdot \tau^{L}(\varphi_{i}, w) = M_{\gamma \delta, \gamma \delta}^{(i)}(U) \cos \omega t + M_{\gamma \delta, \gamma \delta}^{(i)}(V) \sin \omega t$$
(5.2.18)

$$S_{\gamma}(w) = U_{\gamma} \cos \omega t + V_{\gamma} \sin \omega t \qquad (5.2.19)$$

where

$$w(\mathbf{x},t) = U(\mathbf{x})\cos\omega t + V(\mathbf{x})\sin\omega t$$
(5.2.20)

$$M_{\gamma\delta}^{(i)}(U) \equiv M_{\gamma\delta}(\varphi_i, U) \tag{5.2.21}$$

The time integrals $C_{\gamma}^{(1)}$, $C_{\gamma}^{(3)}$, $S_{\gamma}^{(1)}$ and $S_{\gamma}^{(3)}$; $\gamma = 1, 2$, defined in equations (4.17), (4.18), (4.23g) and (4.23h), respectively, can be written for the plate problem as:

$$C_{\gamma}^{(1)}(U,V) = -\frac{\mu}{4} \left[3 \, \mathcal{N}(U,U) + \, \mathcal{N}(V,V) \right] U_{,\gamma} + 2 \, \mathcal{N}(U,V) V_{,\gamma} \tag{5.2.22}$$

$$S_{\gamma}^{(1)}(U,V) = -\frac{\mu}{4} 2 \, \mathcal{N}(U,V)U_{,\gamma} + [\,\mathcal{N}(U,U) + 3 \, \mathcal{N}(V,V)]V_{,\gamma}$$
(5.2.23)

$$C_{\gamma}^{(3)}(U,V) = -\frac{\mu}{4} \left[\eta(U,U) - \eta(V,V) \right] U_{,\gamma} - 2 \eta(U,V) V_{,\gamma}$$
(5.2.24)

$$S_{\gamma}^{(3)}(U,V) = -\frac{\mu}{4} 2 \, \mathcal{N}(U,V)U_{,\gamma} + [\,\mathcal{N}(U,U) - \,\mathcal{N}(V,V)]V_{,\gamma}$$
(5.2.25)

where

$$\gamma_{\iota}(U,V) = \int U_{\cdot\delta} V_{\cdot\delta} d \quad \mathcal{D}(\mathbf{x})$$
(5.2.26)

The preceding relations, (5.2.17) - (5.2.26) are then used to derive the matrix A and vector G for the three problem formulations, as shown below. As before, repeated Greek subscripts will imply summation on indices from 1 to 2.

Formulation I

$$A_{ij}^{I} = \int_{\mathcal{D}} \left[M_{\gamma \alpha, \gamma \delta}^{(i)}(U) M_{\xi \eta, \xi \eta}^{(j)}(U) + M_{\gamma \alpha, \gamma \delta}^{(i)}(V) M_{\xi \eta, \xi \eta}^{(j)}(V) \right] d \quad \mathcal{D}(\mathbf{x})$$
(5.2.27)

$$G_{i}^{I} = -\frac{\mu}{4} \Big\{ \begin{bmatrix} 3 & \eta(U,U) + \eta(U,V) \end{bmatrix} \int M_{\gamma \delta,\gamma \delta}^{(i)}(U) U_{i \notin \delta} d \mathcal{D}(\mathbf{x})$$
(5.2.28)

+ 2
$$\mathcal{N}(U,V) \int M_{\gamma \sigma, \gamma \sigma}^{(i)}(U) V_{\xi \xi} + M_{\gamma \sigma, \gamma \sigma}^{(i)}(V) U_{\xi \xi} d \mathcal{D}(\mathbf{x})$$

+ [
$$\mathcal{N}(U,U)$$
 + 3 $\mathcal{N}(V,V)$] $\int M_{\gamma \alpha \gamma \delta}^{(i)}(V)V_{i \notin d} \mathcal{D}(\mathbf{x})$

Formulation II

$$A_{ij}^{II} = \int_{\mathcal{Q}} \left\{ \left[M_{\gamma \dot{\alpha}, \delta}^{(i)}(U) V_{,\gamma} + M_{\gamma \dot{\alpha}, \delta}^{(i)}(V) U_{,\gamma} \right] \left[M_{\xi \eta, \eta}^{(j)}(U) U_{,\xi} + M_{\xi \eta, \eta}^{(j)}(V) V_{,\xi} \right] + \left[M_{\gamma \dot{\alpha}, \delta}^{(i)}(U) U_{,\gamma} \right] \left[3M_{\xi \eta, \eta}^{(j)}(U) U_{,\xi} + M_{\xi \eta, \eta}^{(j)}(V) V_{,\xi} \right] \right]$$
(5.2.29)

+
$$\left[M_{\mathfrak{H},\mathfrak{h}}^{(i)}\delta(V)V_{,\gamma}\right]\left[M_{\mathfrak{H},\mathfrak{h}}^{(j)}(U)U_{,\gamma}+3M_{\mathfrak{H},\mathfrak{h}}^{(j)}(V)V_{,\gamma}\right]d \mathcal{D}(\mathbf{x})$$

$$G_{\ell}^{II} = -\frac{\mu}{4} \int_{\mathcal{Q}} \left\{ \left[(3 \ \eta(U,U) + \eta(V,V) \right] U_{\ell} U_{\ell} + 2 \ \eta(U,V) V_{\ell} U_{\ell} \right]$$
(5.2.30)

$$\left[3M_{\gamma\delta,\delta}^{(i)}(U)U_{,\gamma}+M_{\gamma\delta,\delta}^{(i)}(V)V_{,\gamma}\right]$$

+
$$[2(\mathcal{N}(U,U) - \mathcal{N}(V,V))U_{\xi}V_{\xi} - 4 \mathcal{N}(U,V)U_{\xi}U_{\xi}][M_{\gamma a}^{(i)}\delta(U)V_{\gamma} - M_{\gamma a}^{(i)}\delta(V)U_{\gamma}]$$

+
$$\left[\mathcal{N}(U,U) - \mathcal{N}(V,V) \right] \left[U_{\xi}U_{\xi} + V_{\xi}V_{\xi} \right] \left[M_{\gamma \alpha \delta}(U)U_{\gamma} - M_{\gamma \alpha \delta}(V)V_{\gamma} \right]$$

+
$$[2 \mathcal{N}(U,V)U_{\xi}V_{\xi} + (\mathcal{N}(U,U) + 3\mathcal{N}(V,V))V_{\xi}V_{\xi}]$$

$$\left[M_{\gamma \alpha \delta}^{(i)}(U)U_{\gamma} + 3M_{\gamma \alpha \delta}^{(i)}(V)V_{\gamma}\right] d \mathcal{D}(\mathbf{x})$$

Formulation III

$$A_{ij}^{III} = \int_{\mathcal{Q}} \left[M_{\gamma \alpha \delta}^{(i)}(U) M_{\gamma \eta, \eta}^{(j)}(U) + M_{\gamma \alpha \delta}^{(i)}(V) M_{\gamma \eta, \eta}^{(j)}(V) \right] d \mathcal{Q}(\mathbf{x})$$

$$G_{i}^{III} = -\frac{\mu}{4} \left\{ \left[3 \, \gamma \left(U, U \right) + \gamma \left(V, V \right) \right] \int_{\mathcal{Q}} M_{\gamma \delta, \delta}^{(i)}(U) U_{,\gamma} d \mathcal{Q}(\mathbf{x}) \right] \right\}$$
(5.2.32)

$$+ 2 \eta (U, V) \int [M_{\gamma \alpha \delta}^{(i)}(U) V_{,\gamma} + M_{\gamma \alpha \delta}^{(i)}(V) U_{,\gamma}] d \mathcal{D}(\mathbf{x})$$

+
$$\left[\begin{array}{c} \eta(U,U) + 3 \ \eta(V,V) \right] \int M_{\gamma \alpha \delta}^{(i)}(V) V_{,\gamma} d \ \mathcal{D}(\mathbf{x}) \right\}$$

5.2.3 Finite element analysis of the auxiliary equations

The finite element method was briefly discussed in Chapter III. A more complete discussion of the mechanics of constructing the finite element equations can be found in a number of finite element textbooks. Only the results of the discretization will be presented here.

Recall from equation (5.2.11) that the auxiliary equation for the thin plate problem was chosen to be:

$$M_{\gamma\delta,\gamma\delta}(D,w) + M_{\gamma\delta,\gamma\delta}(\alpha,w) + b\dot{w} + \rho\dot{w} = p\cos\omega t \qquad (5.2.33)$$

where repeated Greek subscripts imply summation of the indices from 1 to 2 and with all other terms and notation are as before. If the displacements w are written as

$$w(\mathbf{x},t) = \sum_{j=1}^{N} \psi_j(\mathbf{x}) d_j(t) , \qquad (5.2.34)$$

then the finite element equations can be written as:

$$\mathfrak{M}\mathbf{d}(t) + \mathcal{C}\mathbf{d}(t) + \mathcal{K}\mathbf{d}(t) = \mathfrak{F}\cos\omega t \qquad (5.2.35)$$

where

$$\mathcal{H}_{ij} = \int \rho \psi_i \psi_j \, d \, \mathcal{Q} \left(\mathbf{x} \right) \tag{5.2.36}$$

$$C_{ij} = \int b \,\psi_i \psi_j \, d \,\mathcal{Q} \,(\mathbf{x}) \tag{5.2.37}$$

$$\mathcal{H}_{ij} = \int \mathcal{B}_i R \ \mathcal{B}_j \, d \ \mathcal{Q} \ (\mathbf{x}) \tag{5.2.38}$$

$$\mathcal{F}_{i} = \int p(\mathbf{x})\psi_{i} d \mathcal{Q}(\mathbf{x})$$
(5.2.39)

$$\mathcal{B}_{i} = (\psi_{i,11}, \psi_{i,22}, \psi_{i,12})^{T}$$
(5.2.40)

$$R = \begin{bmatrix} (D+\alpha) & \nu & 0 \\ \nu & (D+\alpha) & 0 \\ 0 & 0 & 2(1-\nu)(D+\alpha) \end{bmatrix}$$
(5.2.41)

and $\psi_i(\mathbf{x})$ and $d_i(t)$; i=1, 2, ..., N, being the shape functions and nodal "displacements", respectively.

The domain of the plate is to be divided into N_{el} , four-node quadrilateral elements with each element having nodal points as $\mathbf{x} = \mathbf{x}^a$; a=1, 2, 3, 4. Consider a transformation from a set of local coordinates $\boldsymbol{\xi} = (\xi_1, \xi_2)$ to the global set \mathbf{x} given by

$$\mathbf{x}(\xi) = \sum_{a=1}^{4} N_{a}(\xi) \mathbf{x}^{a}$$
(5.2.42)

where

$$N_{a}(\xi) = \frac{1}{4} (1 + \xi_{1}^{a} \xi_{1}) (1 + \xi_{2}^{a} \xi_{2})$$

and the local coordinates of the nodes ξ^a ; a=1, 2, 3, 4, are as shown in figure 5.2.1. The shape functions for the displacement w will now be defined in terms of the local coordinates, with the understanding that the transformation of the shape functions to global coordinates is accomplished through (5.2.42).

Recall from section 3.7 that one sufficient condition for convergence of the finite element method is the continuity of the m-1 derivatives of the shape functions, where m is the highest order derivative in the strain energy integral (5.2.38). From (5.2.40) it is clear that m=2 for Kirchoff plates, and consequently the shape functions must have continuous *first* derivatives across the element boundaries. A set of shape functions developed in reference [3] constructed from Hermite cubic functions satisfies this requirement. Using these shape functions, the unknown parameters at each node are w, $\frac{\partial w}{\partial x_1}$, $\frac{\partial w}{\partial x_2}$, $\frac{\partial^2 w}{\partial x_1 \partial x_2}$ which leads to a total of 16 degrees-of-freedom/element. Let $\hat{\mathbf{d}}_{\mathbf{a}}$: a=1, 2, 3, 4 be defined as

$$\widehat{\mathbf{d}}_{\mathbf{a}} = (w, \frac{\partial w}{\partial x_1}, \frac{\partial w}{\partial x_2}, \frac{\partial^2 w}{\partial x_1 \partial x_2})^T$$
 at $\xi = \xi^{\mathbf{a}}$

Then the set of nodal parameters d_e ; $e=1, 2, ..., N_{el}$ will be ordered as

$$\mathbf{d}_{\mathbf{e}}^{T} = (\widehat{\mathbf{d}}_{1}, \widehat{\mathbf{d}}_{2}, \widehat{\mathbf{d}}_{3}, \widehat{\mathbf{d}}_{4})^{T}$$

With this, the shape functions in local coordinates from reference [3] are as shown in table 5.2.1.



Figure 5.2.1 Domain of Plate Element in Local Coordinates

a	ψ_{a}
1	$\widehat{\psi}_1(\xi_1) \ \widehat{\psi}_1(\xi_2)$
2	$\widehat{\psi}_3(\xi_1) \ \widehat{\psi}_1(\xi_2)$
3	$\widehat{\psi}_1(\xi_1)$ $\widehat{\psi}_3(\xi_2)$
4	$\widehat{\psi}_3(\xi_1) \ \widehat{\psi}_3(\xi_2)$
5	$\widehat{\psi}_2(\xi_1) \ \widehat{\psi}_1(\xi_2)$
6	$\widehat{\psi}_4(\xi_1) \ \widehat{\psi}_1(\xi_2)$
7	$\widehat{\psi}_2(\xi_1) \ \widehat{\psi}_3(\xi_2)$
8	$\widehat{\psi}_4(\xi_1) \ \widehat{\psi}_3(\xi_2)$
9	$\widehat{\psi}_2(\xi_1) \ \widehat{\psi}_2(\xi_2)$
10	$\widehat{\psi}_4(\xi_1)\widehat{\psi}_2(\xi_2)$
11	$\widehat{\psi}_2(\xi_1)\widehat{\psi}_4(\xi_2)$
12	$\widehat{\psi}_4(\xi_1)\widehat{\psi}_4(\xi_2)$
13	$\hat{\psi}_1(\xi_1) \hat{\psi}_2(\xi_2)$
14	$\widehat{\psi}_3({arepsilon}_1)\widehat{\psi}_2({arepsilon}_2)$
15	$\widehat{\psi}_1(\xi_1)\widehat{\psi}_4(\xi_2)$
16	$\widehat{\psi}_3(\xi_1)\widehat{\psi}_4(\xi_2)$

where

 $\hat{\psi}_{1}(z) = (z-1)^{2} (2+z)/4$ $\hat{\psi}_{2}(z) = (z+1)^{2} (2-z)/4$ $\hat{\psi}_{3}(z) = h_{z} (z+1) (z-1)^{2}/8$ $\hat{\psi}_{4}(z) = h_{z} (z+1)^{2} (z-1)/8$



The finite element discretization of plates governed by the Kirchoff theory has an inherent difficulty in modeling curved boundaries. As mentioned in reference [19], such modeling of a simply-supported, curved boundary boundary will impose a "clamped" boundary condition as the mesh is refined. Consequently, analysis will be restricted to plates having straight edges and , in particular, only rectangular elements will be used.

5.2.4 Convergence considerations for equivalent linear parameters

For the previous example of the one-dimensional shear member, the convergence of the equivalent linear parameters with mesh refinement was readily established. In this example of the nonlinear response of thin plates, complexities arise which do not permit such a simple analysis. Recall from the last section that the finite element analysis of the auxiliary linear equation was performed using bi-cubic shape functions in which their first derivatives were continuous across the element boundaries. But from the minimization relations of formulation III, for example, in equations (5.2.16), (5.2.31) and (5.2.32), it can be seen that continuity of the second derivatives of the displacements are required. Otherwise the terms containing three spatial derivatives such as $M_{7k\delta}^{(i)}$ will possess singularities at the element boundaries which are not square integrable. The technique used herein to avoid this difficulty is to compute the integrals within each separate element, and then simply add together the results, thus ignoring the contribution of the interelement discontinuities.

Using this type of construction of the minimization equations, one should be concerned with how the omission of the singularities on the element boundaries affects the solution. This is also a concern in the finite element area when "nonconforming" elements are used (reference [48]). In this latter case, a "patch test" of elements can be used to determine the completeness convergence criterion of the element. If the element passes such a test, it is then concluded that the discontinuities in the 'non-conforming" elements do not contribute to the strain energy. For the minimization equations, it is not clear whether such a test would be applicable, and therefore the convergence of such a scheme has not been resolved.

In spite of these uncertaintities, the minimization relations will be formed at the element level, as mentioned above, for all three problem formulations. Investigation of how these considerations affect the convergence of the method will be deferred to further investigations.

5.2.5 Numerical results

This section will deal with the symmetric, steady-state, harmonic response of a simply supported rectangular plate governed by the nonlinear theory introduced in section 5.2.1. The numerical results from the equivalent linearization technique will be compared with those from Galerkin's method described in section 2.4.1.

Let the domain of the rectangular plate $\mathfrak{D}(\mathbf{x})$ be defined as

$$\mathfrak{V}(\mathbf{x}) = (-a \leq x_1 \leq a, \ -b \leq x_2 \leq b)$$

The simply supported boundary conditions are therefore given by:

$$w(-a, x_{2}, t) = w(a, x_{2}, t) = 0$$

$$w(x_{1}, -b, t) = w(x_{1}, b, t) = 0$$

$$\frac{\partial^{2} w}{\partial x_{1}^{2}}(-a, x_{2}, t) = \frac{\partial^{2} w}{\partial x_{1}^{2}}(a, x_{2}, t) = 0$$

$$\frac{\partial^{2} w}{\partial x_{2}^{2}}(x_{1}, -b, t) = \frac{\partial^{2} w}{\partial x_{2}^{2}}(x_{1}, b, t) = 0$$
(5.2.43)

That is, the displacements and tangential moments must vanish on the boundary of the plate. If the material properties are constant throughout the plate, the symmetric eigenfunctions $\Phi_k(\mathbf{x})$ and corresponding eigenvalues λ_k for the linearized problem with the boundary conditions of (5.2.43) are

$$\Phi_k(\mathbf{x}) = \cos[\pi(2k-1)\frac{x_1}{2a}] \cos[\pi(2k-1)\frac{x_2}{2b}]$$
(5.2.44)

$$\lambda_{k} = \frac{\pi^{2}}{4} (2k-1)^{2} \left[\frac{1}{a^{2}} + \frac{1}{b^{2}}\right] \sqrt{\frac{D}{\rho}}$$
(5.2.45)

Let the applied load to the plate be due to a spatially constant pressure $\boldsymbol{P}_{\rm C}$, and define the following dimensionless parameters

$$\Omega \equiv \frac{\omega}{\lambda_1}$$

$$\overline{\lambda}_i \equiv \frac{\lambda_i}{\lambda_1} = (2i-1)^2$$

$$\overline{w} \equiv \frac{w}{h}$$

If the nonlinear plate solution is expanded in terms of the first N eigenfunctions of the linearized problem, the non-dimensinal form of the general Galerkin equations (2.18) is:

$$\begin{bmatrix} \Lambda & \Gamma \\ -\Gamma & \Lambda \end{bmatrix} \left\{ \overline{\mathbf{v}} \\ \overline{\mathbf{v}} \\ \mathbf{v} \\$$

whe

$$\Lambda_{ij} = \begin{cases} \overline{\lambda_i^2} - \Omega^2 & \text{if } i = j \\ 0 & \text{if } i \neq j \end{cases}$$

$$\begin{split} \Gamma_{ij} &= \begin{cases} 2\zeta_i \bar{\lambda}_i \Omega & \text{if } i = j \\ 0 & \text{if } i \neq j \end{cases} \\ f_i^G &= \frac{3072(1-\nu^2)}{\pi^6(2i-1)^2} \bar{P} \\ \bar{P} &= \frac{P_0 b^4}{Eh^4[1+(\frac{b}{a})^2]^2} \\ \bar{\mathcal{B}}_{Ci}^G &= \frac{3}{8} \bar{\lambda}_i \sum_{l=1}^N \bar{\lambda}_i [(3\bar{u}_l^Q + \bar{v}_l^Q) \bar{u}_i + \bar{u}_l \bar{v}_l \bar{u}_l] \\ \bar{\mathcal{B}}_{Si}^G &= -\frac{3}{8} \bar{\lambda}_i \sum_{l=1}^N \bar{\lambda}_l [2\bar{u}_l \bar{v}_l \bar{v}_i + (\bar{u}_l^R + 3\bar{v}_l^2) \bar{v}_i] \\ \bar{w} &= \sum_{i=1}^N \Phi_i [\bar{u}_i \cos\omega t + \bar{v}_i \sin\omega t] \end{split}$$

and ζ_i is the damping ratio in the i^{th} linear mode. E and ν are the elastic modulus and Poisson's ratio of the material, respectively.

Therefore, the steady-state Galerkin response is governed by a set of nonlinear algebraic equations in the 'in-phase" and "quadrature" (with respect to the forcing term) components of displacement, $\overline{\mathbf{u}}$ and $\overline{\mathbf{v}}$, respectively. It is interesting to note that for the equations written in the normalized form of (5.2.46) that the plate dimensions a, b and h enter in only through the forcing term. Consequently, the normalized free vibration response of the plate can be presented independent of its aspect ratio or thickness.

5.2.5.1 Free vibration response

If the undamped, unloaded nonlinear plate is given an initial prescribed displacement, it should be expected that the frequency of response will depend on the amplitude of the resulting motion. This contrasts with the linear theory in

99

which the frequency is independent of the amplitude of response.

For free vibration response that can be modeled sufficiently well by the first mode of the linearized problem, the normalized frequency of response, Ω , given by Galerkin's method takes on the rather simple form of

$$\Omega = \left[1 + \frac{9}{8}\bar{u}_1^2\right]^{\frac{1}{2}}$$
(5.2.47)

where $\overline{u_1}$ is the normalized amplitude at the center of the plate. This one-mode Galerkin response is shown in figure 5.2.2.

The equivalent linear free vibration response is governed by the free vibration form of the discretized auxiliary equation (5.2.35) and the minimization relations of (5.2.27) - (5.2.32) for the three problem formulations. Solutions to these equations have been effected through the use of the matrix iteration method detailed in section 4.3.1. Results for the the first nonlinear mode of vibration are presented in figure 5.2.2 for a finite element mesh with 9 degreesof-freedom (DOF's) and in figure 5.2.3 for a 49 DOF mesh.

In order to determine the validity of the equivalent linear solutions, a direct numerical integration procedure was applied to the discretized form of the original nonlinear equation (5.2.10). The discretization was accomplished by using the finite element bi-cubic shape functions in section 5.2.3, which produces a set of nonlinear, ordinary differential equations in terms of the nodal displacements. With prescribed initial displacements, these differential equations were solved by a Runge-Kutta time integration algorithm. The frequency-amplitude results are given in figures 5.2.2 and 5.2.3 for finite element meshes with 9 and 49 DOF's.


Figure 5.2.2 Free Vibration Response of Simply Supported Rectangular Plate (for 9 DOF finite element mesh)



Figure 5.2.3 Free Vibration Response of Simply Supported Rectangular Plate (for 49 DOF finite element mesh)

5.2.5.2 Forced response

The forced response to the one mode Galerkin equations are given by

$$[(1 - \Omega^2)^2 \bar{A}_1 + \frac{9}{8} \bar{A}_1^3]^2 + [2\zeta_1 \Omega \bar{A}_1]^2 = f_1^2$$
(5.2.48)

where

$$\bar{A}_1 = [\bar{u}_1^{e} + \bar{v}_1^{e}]^{\frac{1}{2}}$$

$$f_1 = 48(\frac{2}{\pi})^6(1-\nu^2)\overline{P}$$

and the normalized pressure, \overline{P} is as given following (5.2.46). A plot of Ω versus \overline{A} in (5.2.48) is given in figure (5.2.4) for $\overline{P} = \frac{1}{4}$ and $\frac{3}{4}$. Also in figure (5.2.4) is the formulation III forced response.

5.2.6 Discussion

From figure 5.2.2, it can be seen that solutions from both Galerkin's method and formulation III (stress minimization) of equivalent linearization compare well with the results of direct integration, even for a somewhat "crude" finite element mesh having 9 DOF's. The difference between formulation III and numerical integration becomes somewhat more pronounced at larger amplitudes of vibration. The other two problem formulations, I and II (differential stress and energy difference minimization, respectively), on the other hand, predict a "stiffer" response. That is, for a given amplitude, the frequency of oscillation given by the two methods is significantly higher than that given by numerical integration.

The results from using a more refined finite element mesh are as presented in figure 5.2.3. The difference between formulation III equivalent linearization and numerical integration solutions are less pronounced at larger amplitudes than in figure 5.2.2. Formulation II solutions were not significantly affected by





the change in finite element meshes.

On the other hand, the formulation I approach to the equivalent linearization method behaves rather erratically for the 49 DOF model. The results show that the equivalent linear parameters become small, which in effect makes the free vibration results nearly linear, as can be seen in figure 5.2.3. The reason for such an erratic response could lie in the shape functions used in the finite element model. Recall from section 5.2.2 that the integrands in the equations for the formulation I equivalent linear stiffness parameters are singular at the element boundaries for the bi-cubic shape functions. These singularities are more severe than those encountered by the other two formulations. Therefore the poor performance of formulation I is not particularly surprising.

In conclusion, formulation III appears to be the most attractive of the three approaches to equivalent linearization. The results from formulation III agree with those form direct integration and compare well with free and forced solutions of Galerkin's method. The fact that the one term Galerkin expansion of this simple example produced simpler expressions for the solutions than that given for equivalent linearization is noteworthy. In the next example, however, it can be seen that the simpler Galerkin approach (and the similar Ritz method) can not handle a problem with somewhat more complicated boundary conditions as well as the finite element-equivalent linearization technique.

5.3 Steady-State Harmonic Response of a Nonlinear Plate with a Hole

In this section, the nonlinear response of a plate with somewhat more complex boundary conditions will be investigated. The simply supported rectangular plate of the last section will again be dealt with, but here the effect of a rectangular cutout centered in the plate will be of interest.

The geometry of the problem is as shown in figure 5.3.1. Let $\partial \mathcal{D}_{O}(\mathbf{x})$ be the outer boundary of the plate and $\partial \mathcal{D}_{I}(\mathbf{x})$ be on the perimeter of the cutout. Simply supported conditions are to be imposed on $\partial \mathcal{D}_{O}$, and on $\partial \mathcal{D}_{I}$, moment and shear free conditions exist. It should be noted, however, that in order for the nonlinear theory to be applicable to this problem, the *inplane* displacements of the plate must be constrained at the hole. Therefore a membrane stress will exist on $\partial \mathcal{D}_{I}$ although $\partial \mathcal{D}_{I}$ is free of bending stresses.

5.3.1 Ritz method

For the continuous plate (that is, a plate without cutouts) a Galerkin method has been used in the analysis. This was possible because of the availability of trial functions which satisfied all of the boundary conditions, i.e. comparison functions. However, the presence of the hole, and its associated boundary conditions, create difficulties in finding usuable comparison functions. As can be recalled from section 2.4.1, the class of trial functions that can be used with the Ritz method need only satisfy the geometric boundary conditions. Thus the field of candidates for use with the Ritz method is larger than that which can be used with Galerkin's method.

The use of the linear mode shapes of the continuous plate for analysis of a plate with a cutout is suggested in reference [29]. Reference [26] used a set of trial functions which modeled the appropriate singularities within the hole. For a square plate, the trial functions used in [26] are



Figure 5.3.1 Geometry of Rectangular Plate with Centered Rectangular Cutout

$$\Phi_1(\mathbf{x}) = f(\mathbf{x}) \tag{5.3.1}$$

$$\Phi_2(\mathbf{x}) = f(\mathbf{x}) \frac{x_1^2 + x_2^2}{a^2}$$
(5.3.2)

$$\Phi_{3}(\mathbf{x}) = f(\mathbf{x})\log\left[\frac{\sqrt{x_{1}^{2} + x_{2}^{2}}}{a}\right]$$
(5.3.3)

where

$$f(\mathbf{x}) = \left[1 - \left(\frac{x_1}{a}\right)^2\right] \left[1 - \left(\frac{x_2}{a}\right)^2\right]$$
(5.3.4)

Both types of trial functions will be used in later analysis.

Substitution of the energy and virtual work expressions for the undamped, nonlinear plate into the Ritz equations of section 2.4.2 gives the following normalized relations

$$\left[-\omega^2 \,\mathfrak{M}_{\mathrm{c}} + \mathfrak{K}\right] \mathbf{u} + \,\mathfrak{F}\left(\mathbf{u}\right) = \mathbf{f} \tag{5.3.5}$$

where

$$\mathcal{M}_{ij} = \int_{\mathfrak{D}} \Phi_{i} \Phi_{j} d \, \mathfrak{D}(\mathbf{x})$$

$$\mathcal{M}_{ij} = \frac{D}{\rho} \int_{\mathfrak{D}} [(1-\nu)\Phi_{i,j\delta}\Phi_{j,\gamma\delta} + \nu\Phi_{i,\gamma\gamma}\Phi_{j,\delta\delta}] d \, \mathfrak{D}(\mathbf{x})$$

$$\mathcal{R}_{i} = \frac{9}{8} \frac{D}{\rho h^{2}(ab-cd)} \sum_{j,k,l=1}^{N} [\int_{\mathfrak{D}} \Phi_{i,\gamma}\Phi_{j,\gamma}d \, \mathfrak{D}(\mathbf{x})] [\int_{\mathfrak{D}} \Phi_{k,\delta}\Phi_{l,\delta}d \, \mathfrak{D}(\mathbf{x})] u_{j}u_{k}u_{l}$$

$$f_{i} = \frac{P_{0}}{\rho} \int_{\mathfrak{D}} \Phi_{i} d \, \mathfrak{D}(\mathbf{x})$$

$$\Phi_{i,\gamma} = \frac{\partial\Phi_{i}}{\partial x_{\gamma}}$$

$$\int (\cdot) d \, \mathfrak{D}(\mathbf{x}) = \int_{-a}^{a} \int_{-b}^{b} (\cdot) dx_{1}dx_{2} - \int_{-c}^{c} \int_{-d}^{d} (\cdot) dx_{1}dx_{2}$$

and repeated Greek subscripts imply summation of indices from 1 to 2.

5.3.2 Fundamental natural frequency

For low levels of amplitude, the response of the nonlinear plate should be expected to approach that of the linearized response. Before investigating the nonlinear, free vibration of the plate with a hole, its linearized response should be understood. The satisfaction of the free boundaries at the hole creates difficulties in obtaining exact solutions for the linearized natural frequencies. In this section, the fundamental frequency of a square plate governed by the linear theory will be investigated by three methods: (1) Rayleigh-Ritz method with the mode shapes for the plate without a hole, (2) Rayleigh-Ritz method with the singular trial functions of (5.3.1) - (5.3.4) and (3) the finite element element method.

Recall from section 2.4.2 that the Ritz method reduces to the Rayleigh-Ritz eigenvalue formulation, equation (2.28), for a linear conservative system. The mass and linearized stiffness matrices, \mathcal{M} and \mathcal{K} , for the Ritz method are given in equation (5.3.5). Substitution of the first eigenvalue and eigenfunction of the continuous plate into the expressions for \mathcal{M} and \mathcal{K} give the following relationship for the non-dimensional natural frequency, $\overline{\lambda}$,

$$\bar{\lambda}^{2} = 1 + \frac{2(1-\nu)[F(\frac{c}{a},\frac{d}{b}) - 1](\frac{a}{b})^{2}}{[1 + (\frac{a}{b})^{2}]^{2}}$$
(5.3.6)

where

$$\overline{\lambda} = \frac{\lambda_H}{\lambda_{NH}}$$

$$F = \frac{1 - \left[\frac{c}{a} - \frac{1}{\pi}\sin\pi\frac{c}{a}\right]\left[\frac{d}{b} - \frac{1}{\pi}\sin\pi\frac{d}{b}\right]}{1 - \left[\frac{c}{a} + \frac{1}{\pi}\sin\pi\frac{c}{a}\right]\left[\frac{d}{b} + \frac{1}{\pi}\sin\pi\frac{d}{b}\right]}$$

and λ_H and λ_{NH} are the fundamental natural frequencies of the linear plate with and without a hole, respectively. The use of the three singular trial functions of (5.3.1) - (5.3.4) in the Rayleigh-Ritz equations produces a 3x3 eigenvalue problem. The solution for the lowest eigenvalue for a square plate with a square cutout is shown in figure 5.3.2 for the full range of hole sizes. Equations (5.3.6) with a = b and c = d is also plotted in figure 5.3.2. The fundamental eigenvalue for the finite element formulation of the linear plate has also been determined using typical element meshes of figure 5.3.3 are also shown in figure 5.3.2.

It can be seen from figure 5.3.2 that the Rayleigh-Ritz frequencies are always as large as those obtained via the finite element method. In fact, for holewidth-to-plate-width ratios less than ≈ 0.34 , the finite element frequencies are actually lower than those of the plate without a hole. Reference [48] provides a proof that eigenvalues from the finite element method are an upper bound for the actual eigenvalues. Consequently, it should be assumed from figure 5.3.2 that the finite element frequencies are better estimates of the actual frequencies than those of either of the two Rayleigh-Ritz formulations.

5.3.3 Nonlinear free vibration response

Substitution of the first eigenfunction of the linear, continuous plate from equation (5.2.44) into the Ritz equations (5.3.5) for a square plate with a square cutout gives the following amplitude-frequency relationship:

$$\Omega = \left[\bar{\lambda}^2 + \frac{9}{8}\bar{\mu}\bar{u}^2\right]^{\frac{1}{2}}$$
(5.3.7)

where



Figure 5.3.2 Linear Fundamental Natural Frequency of Simply Supported Square Plate with Square Hole for v = 0.3





c/a = 0.2

c/a = 0.4



c/a = 0.6



c/a = 0.8



$$\Omega = \frac{\omega}{\lambda_{NH}}$$

$$\overline{u} = \frac{w_{\max}}{h}$$

$$\overline{\mu} = \frac{\left[1 - (\frac{c}{a})^2 + \frac{1}{\pi^2} \sin^2 \pi \frac{c}{a}\right]^2}{\left[1 - (\frac{c}{a})^2\right] \left[1 - (\frac{c}{a} + \frac{1}{\pi} \sin \pi \frac{c}{a})^2\right] \cos^2 \frac{\pi}{2} \frac{c}{a}}$$

and w_{max} is the maximum displacement in the plate. $\overline{\lambda}$ is the Rayleigh-Ritz fundamental frequency of the plate given in (5.3.6) for a = b and c = d. If c = 0, i.e. no hole, both $\overline{\lambda}$ in (5.3.6) and $\overline{\mu}$ in (5.3.7) are equal to one, and the amplitude frequency relationship (5.3.7) reduces to that derived by Galerkin's method, equation (5.2.47). This is consistent with the fact that the Ritz method is equivalent to Galerkin's method when comparison functions are used (reference [34]).

In the same manner as for the continuous plate, the free vibration response using the formulation III equivalent linearization technique has been determined for a range of moderately sized holes. The results are presented in figure 5.3.4 along with the Ritz relationship (5.3.7). Also shown in figure 5.3.4 is the free vibration response fo a hole size of $\frac{c}{a} = 0.5$ obtained by numerical integration technique described in section 5.2.5.1.

The effect of hole size on the nonlinear resonant frequency shift of the plate can be seen by looking at a given level of strain, say $\max(\frac{\partial w}{\partial x})$, rather than for a given level of displacement, as before. Recall that the nonlinear stiffness in the plate is related to the total midplane stretch, which is in itself a functional of the strain and not the displacement. In figure 5.3.5, the difference in the linear and nonlinear resonant frequencies, $\Delta\Omega$ is plotted versus hole size for



Free Vibration Response of Simply Supported Square Plate with Square Hole for y = 0.3Figure 5.3.4

114



Figure 5.3.5 Effect of Hole Size on Resonant Frequency Shift for Simply Supported Square Plate with Square Hole, v = 0.3

two normalized levels of strain , $\overline{u}_{,x}$, where

$$\Delta \Omega = \frac{\omega - \lambda_H}{\lambda_H}$$

$$\overline{u_x} = \frac{a}{h} \max(\frac{\partial w}{\partial x})$$

and λ_H , as before, is the linear, fundamental natural frequency of the plate without a hole.

5.3.4 Nonlinear forced response

As with the analysis of the continous plate, the nonlinear response of the cutout plate to a temporally harmonic and spatially constant pressure load will be investigated. Here the emphasis will be on establishing a relation between the size of the hole and the resonant response experienced by the plate.

In section 4.3.2, a method was introduced to determine the relationship between applied pressure and the resonant amplitude of response of a lightly damped system via the equivalent linearization technique. The method established a rather simple relationship, equation (4.59), between the applied load and resonant response, once the free vibration "backbone" curve had been calculated.

Results of the analysis are shown in figure 5.3.6 for a damping ratio of $\zeta = 0.03$ and for a range of moderately sized holes. If the linear theory for the vibration of plates were used, a linear relationship would exist between the applied load and the resonant amplitude. This linear relation is also shown in figure 5.3.6 and, as consistent with the usual hypothesis implied by the use of the linear theory, is tangent to the nonlinear curve at small amplitudes.



Figure 5.3.6 Effect of Applied Pressure on Resonant Response of Simply Supported Square Plate with Square Hole, v = 0.3, $\xi = 0.03$

5.3.5 Discussion

As shown in section 5.3.2, the finite element method provided a means of obtaining a better estimate of the eigenvalues than by the Rayleigh-Ritz approach. In particular, the finite element method was able to model an initial decrease in the fundamental frequency with an increase in hole size. This phenomena is qualitatively similar to that experienced by a circular, simply supported plate with a concentric circular hole, as reported in [26], where, for hole-width-to-plate-width ratios up to 0.4, the fundamental frequency was less that that of the same plate without a hole. The inability of the Ritz procedure to accurately determine the linear eigenvalues will introduce an artificial frequency shift when considering nonlinear analysis for even small amplitudes of response.

In section 5.3.3, the nonlinear free vibration response has been reported in two different ways. The physically appealing maximum displacement-frequency response has been shown in figure 5.3.4 while in figure 5.3.5, a mathematically appealing maximum strain amplitude-frequency response has been presented. It can be seen from figure 5.3.4 that the equivalent linearization predicted a frequency shift in displacements due to the nonlinearity for hole sizes of c/a = 0.2and 0.3 to be less than the frequency shift for the plate without a hole. For all hole sizes, the nonlinear frequency shift predicted for displacement response by the Ritz approach is always larger than that given by equivalent linearization.

In figure 5.3.5, the frequency shift in the maximum strain response has been normalized by the fundamental natural frequency of the cutout plate. Here both methods show that the frequency shift monotonically increases up to a maximum value with increasing hole size, and then decreases. The difference between the two methods is that the Ritz frequency shift for the cutout plate is

118

always larger than that for the continuous plate, whereas with the equivalent linearization method, the frequency shift for large holes is less than for the continuous plate. Hence the two methods predict significantly different type of behavior in the vibration of a plate with a hole.

Figure 5.3.6 shows that for small holes, say c/a = 0.2, the nonlinear forced response remains nearly linear for larger amplitudes than for the same plate without a hole. However, for hole sizes of at least c/a = 0.5, the need for non-linear analysis becomes more apparent, even for lower amplitudes of response.

VI. CONCLUDING REMARKS

The analysis presented in this dissertation has been focused on the type of nonlinear dynamic system introduced in Chapter II. The system described there was one for which the limiting behavior for low level response can be represented by a linear model. This linear model was prescribed to be one of which the restoring stress operation was positive definite and self-adjoint.

The linearizability of the nonlinear system has been the foundation of the new continuum equivalent linearization method presented in Chapter III. That is, it was felt that the behavior of the nonlinear system could be modeled sufficiently well by a replacement, auxiliary linear system. The auxiliary system was determined three ways by the minimization of differences between three distinct quantities of the nonlinear and auxiliary linear systems. These difference quantities were:

- 1. differential stress difference (formulation I)
- 2. energy difference (formulation II)
- 3. stress difference (formulation III)

At this point, the new continuum equivalent equivalent linearization method became significantly different from its well-established equivalent linearization counterpart for discrete systems. Since the auxiliary system was defined at the continuum level, the system difference was not only a function of time, as with the discrete method, but also a function of the spatial coordinates. Hence the equivalent linear parameters were expanded in terms of trial functions (of the spatial coordinates), and the minimization between the system difference terms was performed with respect to the equivalent linear parameters.

It was shown in section 3.8 that, through the use of the positive definiteness

and self-adjointness of the stress operation in the auxiliary system, the minimization relationships for the equivalent linear stiffness and damping parameters were always invertible for formulations I and III. For formulation II, the invertibility was established for special systems. This contrasts with the minimization relations for discrete equivalent linearization. As pointed out in section 2.4.3, if the nonlinear system is discretized first followed by linearization of the resulting ordinary differential equations, the invertibility of the minimization relations is guaranteed for only a small number of degrees-of-freedom. Therefore, an advantage has been established for the continuum approach to the equivalent linearization over its discrete counterpart.

The above minimization procedure delineates one set of relations between the equivalent linear parameters and solutions to the auxiliary equations. Another set of relationships arise from the spatial discretization of the auxiliary equations. The finite element method to be used in the spatial discretization has been described in section 3.7 in terms of its convergence requirements. Of importance was the fact that continuity requirements on the displacement shape functions are generally more restrictive for the minimization relations than for the discretization of the auxiliary equations. Therefore consideration of the linearization technique must be made in choosing the displacement shape functions.

In Chapter IV, the general equivalent linear relationships from Chapter III were applied to the specific analysis of steady-state response to harmonic input and stationary response to stochastic excitation. For harmonic input, the minimization relations have been written in terms of the in-phase and quadrature components of the auxiliary equation solutions, whereas for random response the relations are written in terms of the covariance matrix of the discretized auxiliary system.

121

The continuum equivalent linearization technique was applied to the steadystate, harmonic analysis of three specific types of structural elements governed by nonlinear models. The first, the one-dimensional shear beam, was included to clarify the concepts and mechanics of one formulation of the method. The second and third examples, both pertaining to the nonlinear vibration of thin plates, were used to compare the results from the three problem formulations to solutions from existing methods and to demonstrate the applicability of the new method to problems to which standard techniques are not well suited.

In section 5.1, the response of a one-dimensional shear beam composed of a hysteretic material was investigated. Convergence of the formulation III approach was established. Single mode solutions were obtained for the equivalent linearization method and compared with results obtained from Galerkin's method. It was found that, in the limit as the mesh for the equivalent linear parameters is further subdivided, the formulation III solutions were precisely those of Galerkin's method. For finite subdivisons of the mesh, it was seen that the peak response underestimated the Galerkin solution.

The second example, found in section 5.2, was the vibration of a thin plate in which nonlinear coupling exists between the membrane and bending stresses. Use of the Poisson-Kirchhoff theory of thin plates dictated that the displacement shape functions have continuous first derivatives. Therefore Hermite bicubic shape functions were chosen. However, the minimization relations in section 5.2.2 required the use of displacements that had continuous second derivatives (formulations II and III) and third derivatives (formulation I). In lieu of using displacement fields with higher levels of continuity, it was decided that the resulting inter-element singularities would be discarded in the formation of the minimization equations. Results from section 5.2 showed that the problem formulation I did not perform well in the analysis of nonlinear plates, and that the poor performance could possibly be attributed to the previously mentioned singularities in the minimization relations. The formulation III results compared well with the numerical integration solutions while formulation II consistently predicted a stiffer response. Formulaion III was therefore chosen to be the best approach for further investigation.

Section 5.3 dealt with the further examination of the nonlinear plate from section 5.2, whereas in section 5.3 the plate was permitted to have a concentrically located cutout. The Ritz procedure described in section 2.4.2 was not capable of satisfying the stress free boundary conditions at the hole. This resulted in an artificial frequency shift in the nonlinear response.

A finite element linear eigenvalue analysis of the cutout plate showed that for a range of small hole sizes, the fundamental natural frequency is less than that of the plate with no cutout. The use of the finite element-equivalent linearization technique predicted a similar, more flexible response of the nonlinear displacements for small holes than for the same plate without a hole.

In conclusion, it is felt that the continuum approach to equivalent linearization holds promise for use in the nonlinear analysis of structural members. The formulation of the method is such that the method can easily be incorporated into existing finite element computer codes. The construction of the method has been such that an understanding of the solution properties has been possible in this dissertation. Most importantly, the stress difference formulation has been successfully applied to structural elements in which both stiffness and energy dissipation properties are response dependent.

REFERENCES

- Atalik, T. S. and S. Utku, 'Stochastic Linearization of Multi-Degree-of-Freedom Non-Linear Systems", Earthquake Engineering and Structural Dynamics, 4, (1976), pp. 411-420
- [2] Berger, H. M., "A New Approach to the Analysis of Large Deflections of Plates", Journal of Applied Mechanics, 22, (1955), pp. 465-472
- [3] Bogner, F. K., R. L. Fox and L. A. Schmidt, 'The Generation of Interelement-Compatible Stiffness and Mass Matrices by the Use of Interpolation Formulae", Proceedings of the Conference on Matrix Methods in Structural Mechanics, Air Force Institute of Technology, Wright Patterson A. F. Base, Ohio, (October 1965)
- [4] Bogoliubov, N. N. and Y. A. Mitroplosky, Asymptotic Methods in the Theory of Non-Linear Oscillations, second edition, Gordon and Breach: New York, 1961
- [5] Booton, R. C., 'The Analysis of Non-Linear Control Systems with Random Inputs', in Proceedings of the Symposium on Non-linear Circuit Analysis, Vol. 2, 1953
- [6] Busby, H. R. Jr. and V. I. Weingarten, 'Non-Linear Response of a Beam to Periodic Loading", International Journal of Non-Linear Mechanics, 7, (1972), pp. 289-303
- [7] Caughey, T. K., 'Non-linear Theory of Random Vibrations', Advances in Applied Mechanics, 11, (1971), pp. 209-253
- [8] Caughey, T. K., 'Equivalent Linearization Techniques', Journal of the Acoustical Society of America, 35, (1963), pp 1706-1711
- [9] Caughey, T. K., 'Forced Oscillations of a Semi-Infinite Rod Exhibiting Weak Bilinear Hysteresis", Journal of Applied Mechanics, 27, (1960), pp. 644-648
- [10] Caughey, T. K., 'Response of a Non-Linear String to Random Loading', Journal of Applied Mechanics, 26, (1959), pp. 341-344
- [11] Eisley, J. G., 'Nonlinear Deformation of Elastic Beams, Rings and Strings', Applied Mechanics Surveys, (1966), pp. 285-290
- [12] Eisley, J. G., 'Non-Linear Vibration of Beams and Rectangular Plates', Zeitschrift fur Angewandte Mathematik und Physik, 15, (1964), pp. 167-175
- [13] Eisley, J. G. and J. A. Bennett, 'Stability of Large Amplitude Forced Motion of a Simply Supported Beam", International Journal of Non-Linear Mechanics, 5, (1970), pp. 645-657
- [14] Gajendar, N., 'Large Amplitude Vibrations of Plates on an Elastic Foundation", International Journal of Non-Linear Mechanics", 2, (1967), pp. 163-172
- [15] Ginsberg, J. H., 'Large Amplitude Forced Vibrations of Simply Supported Thin Cylindrical Shells', Journal of Applied Mechanics, (1973), pp. 471-477
- [16] Goldstein, H., Classical Mechanics, Addison-Wesley, (1965)
- [17] Ho, C-. H., R. A. Scott and J. G. Eisley, 'Non-Planar, Non-Linear Oscillations of a Beam - I. Forced Motion", International Journal of Non-Linear Mechanics, 10, (1975), pp. 113-127

- [18] Huang, C-. L., 'Non-Linear Oscillations of a Non-Uniform Fixed Circular Plate", International Journal of Non-Linear Mechanics, 7, (1972), pp. 557-569
- [19] Hughes, T. J. R., M. Cohen and M. Haroun, 'Reduced and Selective Integration Techniques in the Finite Element Analysis of Plates', Nuclear Engineering and Design, 46, (1978), pp.203-222
- [20] Iwan, W. D., "Application of Nonlinear Analysis Techniques", in Applied Mechanics in Earthquake Engineering, W. D. Iwan, ed., ASME Symposium, AMD vol. 8, (1974), pp. 135-161
- [21] Iwan, W. D., "A Generalization of the Concept of Equivalent Linearization", International Journal of Non-Linear Mechanics, 8, (1973), pp. 279-287
- [22] Iwan, W. D., 'Steady-State Response of a Yielding Shear Structure', ASCE Journal of Engineering Mechanics, 6, (1970), pp. 1209-1228
- [23] Iwan, W. D., "On the Steady-State Response of a One-Dimensional Yielding Continuum", Journal of Applied Mechanics, 37, (1970), pp. 720-727
- [24] Iwan, W. D., 'On a Class of Models for the Yielding Behavior of Continuous and Composite Systems", Journal of Applied Mechanics, 34, (1967), pp. 612-617
- [25] Jameson, A., 'Solution of the Equation AX + XB = C by the Inversion of an MxM or an NxN Matrix', SIAM Journal of Applied Mechanics, 16, (1968), pp. 1020-1023
- [26] Joga Rao, C. V. and G. Pickett, 'Vibrations of Plates of Irregular Shapes and Plates with Holes', Journal of the Aeronautical Society, 13, (1961), pp. 83-87
- [27] Kantorovitch, L. V. and V. I. Krylov, Approximate Methods of Higher Analysis, P. Nordhoff Ltd., Groningen, 1958
- [28] Kisliakov, S. D., "On the Non-Linear Dynamic Stability Problem for Thin Elastic Plates", International Journal of Non-Linear Mechanics, 11, 1976, pp. 219-228
- [29] Kristiansen, U. and W. Soedel, 'Fundamental Frequencies of Cutout Square Plate', Journal of Engineering for Industry, 93, (1971), pp. 343-345
- [30] Lin, Y. K., Probablistic Theory of Structural Dynamics, Huntington, N. Y.: Kreiger, 1976
- [31] Lin, Y. K., 'Response of a Non-Linear Flat Panel to Periodic and Randomly-Varying Loadings", Journal of the Aerospace Sciences, 29, (1962), pp. 1029-1034
- [32] McDonald, P. H. Jr., 'Non-Linear Dynamic Coupling in a Beam Vibration', Journal of Applied Mechanics, 22, (1955), pp. 573-578
- [33] McQuery, C.E. and L. G. Clark, 'Periodic Oscillations of a Class of Non-Autononous Non-Linear Elastic Continua", International Journal of Non-Linear Mechanics, 2, (1967), pp. 331-342
- [34] Meirovitch, L., Analytical Methods in Vibrations, New York, N. Y. : Macmillan, 1967
- [35] Muller, P. C., 'Solution of the Matrix Equations AX + XB = -Q and $S^{T}X + XS = -Q$ ", SIAM Journal of Applied Mathematics, 18, (1970), pp.

682-687

- [36] Nash, W. A. and J. R. Modeer, 'Certain Approximate Analyses of Non-Linear Behavior of Plates and Shallow Shells', Proceedings of the Symposium on the Theory of Thin Elastic Shells, Delft, The Netherlands, August 1959
- [37] Nayfeh, A. H., D. T. Mook and S. Sridhar, "Non-Linear Analysis of the Forced Response of Structural Elements", Journal of the Acoustical Society of America, 55, (1974), pp. 281-291
- [38] Powell, M. J. D., "A Hybrid Method for Nonlinear Equations", in Numerical Methods for Nonlinear Algebraic Equations, (1970), pp. 87-114
- [39] Powell, M. J. D., 'A Fortran Subroutine for Solving Systems of Nonlinear Algebraic Equations", Numerical Methods for Nonlinear Algebraic Equations, (1970), pp. 115-161
- [40] Rehfield, L. W., 'Nonlinear Free Vibrations of Elastic Structures', International Journal of Solids and Structures, 9, (1973), pp. 581-590
- [41] Sathyamoorthy, M. and K. A. V. Pandalai, 'Large Amplitude Vibrations of Certain Deformable Bodies. Part I - Membranes and Rings', Journal of the Aeronautical Society of India, 24, (1972), pp. 409-414
- [42] Sathyamoorthy, M. and K. A. V. Pandalai, 'Large Amplitude Vibrations of Certain Deformable Bodies. Part II - Plates and Shells', Journal of the Aeronautical Society of India, 25, (1973), pp. 1-10
- [43] Singh, M. P. and T. P Khatua, "Stochastic Seismic Stability Prediction of Earth Dams", Earthquake Engineering and Soil Dynamics, II, (1978), pp. 875-889
- [44] Somerset, J. H. and R. M. Evan-Iwanowski, "Influence of Non-Linear Inertia on the Parametric Response of Rectangular Plates", Interational Journal of Non-Linear Mechanics, 2, (1967), pp. 217-232
- [45] Spanos, P. -T. D. and W. D. Iwan, "On the Existence and Uniqueness of Solutions Generated by Equivalent Linearization", International Journal of Non-Linear Mechanics, 13, (1978), pp. 71-78
- [46] Sridhar, S., D. T. Mook and A. H. Neyfeh, 'Non-Linear Resonances in the Forced Responses of Plates, Part I : Symmetric Response of Circular Plates', Journal of Sound and Vibration, 41, (1975), pp. 359-373
- [47] Srinivasan, A. V., 'Non-Linear Vibrations of Beams and Plates', International Journal of Non-Linear Mechanics, 7, (1966), pp. 179-191
- [48] Strang, G. and G. J. Fix, An Analysis of the Finite Element Method, Englewood Cliffs, N. J.: Prentice Hall, 1973
- [49] Taylor, R. L., P. J. Beresford and E. L. Wilson, "A Non-Conforming Element for Stress Analysis", International Journal of Numerical Methods in Engineering, 10, (1976), pp. 1211-1219
- [50] Van Dooren, R. and R. Booc, 'Two Mode Sub-Harmonic and Harmonic Vibrations of a Non-Linear Beam Forced by a Two Mode Harmonic Load', International Journal of Non-Linear Mechanics, 10, (1975), pp. 271-280.
- [51] Wah, T., 'Large Amplitude Flexural Vibration of Rectangular Plates', International Journal of Mechanical Science, 5, (1963), pp. 425-438

- [52] Weinberger, H. F., A First Course in Partial Differential Equations, New York. N. Y. : John Wiley and Sons, 1965
- [53] Woinowsky-Kreiger, S., 'The Effect of an Axial Force on the Vibration of Hinged Bars", Journal of Applied Mechanics, 17, (1950), pp. 35-36
- [54] Zienkiewicz, O. C., The Finite Element Method, London : McGraw-Hill, third edition, 1977

CALIFORNIA INSTITUTE OF TECHNOLOGY

Reports Published

by

Earthquake Engineering Research Laboratory Dynamics Laboratory Disaster Research Center

Note: Numbers in parenthesis are Accession Numbers assigned by the National Technical Information Service; these reports may be ordered from the National Technical Information Service, 5285 Port Royal Road, Springfield, Virginia, 22161. Accession numbers should be quoted on orders for reports (PB --- ---). Reports without this information either have not been submitted to NTIS or the information was not available at the time of printing. An N/A in parenthesis indicates that the report is no longer available at Caltech.

- Alford, J.L., G.W. Housner and R.R. Martel, "Spectrum Analysis of Strong-Motion Earthquakes," 1951. (Revised August 1964). (N/A)
- 2. Housner, G.W., "Intensity of Ground Motion During Strong Earthquakes," 1952. (N/A)
- 3. Hudson, D.E., J.L. Alford and G.W. Housner, "Response of a Structure to an Explosive Generated Ground Shock," 1952. (N/A)
- 4. Housner, G.W., "Analysis of the Taft Accelerogram of the Earthquake of 21 July 1952." (N/A)
- 5. Housner, G.W., "A Dislocation Theory of Earthquakes," 1953. (N/A)
- 6. Caughey, T.K., and D.E. Hudson, "An Electric Analog Type Response Spectrum," 1954.(N/A)
- Hudson, D.E., and G.W. Housner, "Vibration Tests of a Steel-Frame Building," 1954. (N/A)
- 8. Housner, G.W., "Earthquake Pressures on Fluid Containers," 1954. (N/A)
- 9. Hudson, D.E., "The Wilmot Survey Type Strong-Motion Earthquake Recorder," 1958. (N/A)

- 10. Hudson, D.E., and W.D. Iwan, "The Wilmot Survey Type Strong-Motion Earthquake Recorder, Part II," 1960. (N/A)
- 11. Caughey, T.K., D.E. Hudson, and R.V. Powell, "The CIT Mark II Electric Analog Type Response Spectrum Analyzer for Earthquake Excitation Studies," 1960. (N/A)
- 12. Keightley, W.O, G.W. Housner and D.E. Hudson, "Vibration Tests of the Encino Dam Intake Tower," 1961. (N/A)
- 13. Merchant, Howard Carl, "Mode Superposition Methods Applied to Linear Mechanical Systems Under Earthquake Type Excitation," 1961. (N/A)
- 14. Iwan, Wilfred D., "The Dynamic Response of Bilinear Hysteretic Systems," 1961. (N/A)
- 15. Hudson, D.E., "A New Vibration Exciter for Dynamic Test of Full-Scale Structures," 1961. (N/A)
- 16. Hudson, D.E., "Synchronized Vibration Generators for Dynamic Tests of Full-Scale Structures," 1962. (N/A)
- 17. Jennings, Paul C., "Velocity Spectra of the Mexican Earthquakes of 11 May and 19 May 1962," 1962. (N/A)
- 18. Jennings, Paul C., "Response of Simple Yielding Structures to Earthquake Excitation," 1963. (N/A)
- 19. Keightley, Willard O., "Vibration Tests of Structures," 1963. (N/A)
- 20. Caughey, T.K. and M.E.J. O'Kelly, "General Theory of Vibration of Damped Linear Dynamic Systems," 1963. (N/A)
- 21. O'Kelly, M.E.J., "Vibration of Viscously Damped Linear Dynamic Systems," 1964. (N/A)
- 22. Nielsen, N. Norby, "Dynamic Response of Multistory Buildings," 1964. (N/A)
- 23. Tso, Wai Keung, "Dynamics of Thin-Walled Beams of Open Section," 1964. (N/A)
- 24. Keightley, Willard O., "A Dynamic Investigation of Bouquet Canyon Dam," 1964. (N/A)
- 25. Malhotra, R.K., "Free and Forced Oscillations of a Class of Self-Excited Oscillators," 1964.

- 26. Hanson, Robert D., "Post-Elastic Response of Mild Steel Structures," 1965.
- 27. Masri, Sami F., "Analytical and Experimental Studies of Impact Dampers," 1965.
- 28. Hanson, Robert D., "Static and Dynamic Tests of a Full-Scale Steel-Frame Structure," 1965.
- 29. Cronin, Donald L., "Response of Linear, Viscous Damped Systems to Excitations Having Time-Varying Frequency," 1965.
- 30. Hu, Paul Yu-fei, "Analytical and Experimental Studies of Random Vibration," 1965.
- 31. Crede, Charles E., "Research on Failure of Equipment when Subject to Vibration," 1965.
- 32. Lutes, Loren D., "Numerical Response Characteristics of a Uniform Beam Carrying One Discrete Load," 1965. (N/A)
- 33. Rocke, Richard D., "Transmission Matrices and Lumped Parameter Models for Continuous Systems," 1966. (N/A)
- 34. Brady, Arthur Gerald, "Studies of Response to Earthquake Ground Motion," 1966. (N/A)
- 35. Atkinson, John D., "Spectral Density of First Order Piecewise Linear Systems Excited by White Noise," 1967. (N/A)
- 36. Dickerson, John R., "Stability of Parametrically Excited Differential Equations," 1967. (N/A)
- 37. Giberson, Melbourne F., "The Response of Nonlinear Multi-Story Structures Subjected to Earthquake Excitation," 1967. (N/A)
- 38. Hallanger, Lawrence W., "The Dynamic Stability of an Unbalanced Mass Exciter," 1967.
- 39. Husid, Raul, "Gravity Effects on the Earthquake Response of Yielding Structures," 1967. (N/A)
- 40. Kuroiwa, Julio H., "Vibration Test of a Multistory Building," 1967. (N/A)
- 41. Lutes, Loren Daniel, "Stationary Random Response of Bilinear Hysteretic Systems," 1967.

- 42. Nigam, Navin C., "Inelastic Interactions in the Dynamic Response of Structures," 1967.
- 43. Nigam, Navin C. and Paul C. Jennings, "Digital Calculation of Response Spectra from Strong-Motion Earthquake Records," 1968.
- 44. Spencer, Richard A., "The Nonlinear Response of Some Multistory Reinforced and Prestressed Concrete Structures Subjected to Earthquake Excitation," 1968. (N/A)
- 45. Jennings, P.C., G.W. Housner and N.C. Tsai, "Simulated Earthquake Motions," 1968.
- 46. "Strong-Motion Instrumental Data on the Borrego Mountain Earthquake of 9 April 1968," (USGS and EERL Joint Report), 1968.
- 47. Peters, Rex B., "Strong Motion Accelerograph Evaluation," 1969.
- 48. Heitner, Kenneth L., "A Mathematical Model for Calculation of the Run-Up of Tsunamis," 1969.
- 49. Trifunac, Mihailo D., "Investigation of Strong Earthquake Ground Motion," 1969. (N/A)
- 50. Tsai, Nien Chien, "Influence of Local Geology on Earthquake Ground Motion," 1969. (N/A)
- 51. Trifunac, Mihailo D., "Wind and Microtremor Induced Vibrations of a Twenty-Two Steel Frame Building," EERL 70-01, 1970.
- 52. Yang, I-Min, "Stationary Random Response of Multidegree-of-Freedom Systems," DYNL-100, June 1970. (N/A)
- 53. Patula, Edward John, "Equivalent Differential Equations for Nonlinear Dynamical Systems," DYNL-101, June 1970.
- 54. Prelewicz, Daniel Adam, "Range of Validity of the Method of Averaging," DYNL-102, 1970.
- 55. Trifunac, M.D., "On the Statistics and Possible Triggering Mechanism of Earthquakes in Southern California," EERL 70-03, July 1970.
- 56. Heitner, Kenneth Leon, "Additional Investigations on a Mathematical Model for Calculation of the Run-Up of Tsunamis," July 1970.

- 57. Trifunac, Mihailo D., "Ambient Vibration Tests of a Thirty-Nine Story Steel Frame Building," EERL 70-02, July 1970.
- 58. Trifunac, Mihailo and D.E. Hudson, "Laboratory Evaluations and Instrument Corrections of Strong-Motion Accelerographs," EERL 70-04, August 1970. (N/A)
- 59. Trifunac, Mihailo D., "Response Envelope Spectrum and Interpretation of Strong Earthquake Ground Motion," EERL 70-06, August 1970.
- Keightley, W.O., "A Strong-Motion Accelerograph Array with Telephone Line Interconnections," EERL 70-05, September 1970.
- 61. Trifunac, Mihailo D., "Low Frequency Digitization Errors and a New Method for Zero Baseline Correction of Strong-Motion Accelerograms," EERL 70-07, September 1970.
- 62. Vijayaraghavan, A., "Free and Forced Oscillations in a Class of Piecewise-Linear Dynamic Systems," DYNL-103, January 1971.
- 63. Jennings, Paul C., R.B. Matthiesen and J. Brent Hoerner, "Forced Vibrations of a 22-Story Steel Frame Building," EERL 71-01, February 1971. (N/A) (PB 205 161)
- 64. Jennings, Paul C., "Engineering Features of the San Fernando Earthquake of February 9, 1971," EERL 71-02, June 1971. (PB 202 550)
- 65. Bielak, Jacobo, "Earthquake Response of Building-Foundation Systems," EERL 71-04, June 1971. (N/A) (PB 205 305)
- 66. Adu, Randolph Ademola, "Response and Failure of Structures under Stationary Random Excitation," EERL 71-03, June 1971. (N/A) (PB 205 304)
- 67. Skattum, Knut Sverre, "Dynamic Analysis of Coupled Shear Walls and Sandwich Beams," EERL 71-06, June 1971. (N/A) (PB 205 267)
- 68. Hoerner, John Brent, "Modal Coupling and Earthquake Response of Tall Buildings," EERL 71-07, June 1971. (N/A) (PB 207 635)
- 69. Stahl, Karl John, "Dynamic Response of Circular Plates Subjected to Moving Massive Loads," DYNL-104, June 1971. (N/A)

- 70. Trifunac, M.D., F.E. Udwadia and A.G. Brady, "High Frequency Errors and Instrument Corrections of Strong-Motion Accelerograms," EERL 71-05, 1971. (PB 205 369)
- 71. Furuike, D.M., "Dynamic Response of Hysteretic Systems with Application to a System Containing Limited Slip," DYNL-105, September 1971. (N/A)
- 72. Hudson, D.E. (Editor), "Strong-Motion Instrumental Data on the San Fernando Earthquake of February 9, 1971," (Seismological Field Survey, NOAA, C.I.T. Joint Report), September 1971. (PB 204 198)
- 73. Jennings, Paul C. and Jacobo Bielak, "Dynamics of Building-Soil Interaction," EERL 72-01, April 1972. (PB 209 666)
- 74. Kim, Byung-Koo, "Piecewise Linear Dynamic Systems with Time Delays," DYNL-106, April 1972.
- 75. Viano, David Charles, "Wave Propagation in a Symmetrically Layered Elastic Plate," DYNL-107, May 1972.
- 76. Whitney, Albert W., "On Insurance Settlements Incident to the 1906 San Francisco Fire," DRC 72-01, August 1972. (PB 213 256)
- 77. Udwadia, F.E., "Investigation of Earthquake and Microtremor Ground Motions," EERL 72-02, September 1972. (PB 212 853)
- 78. Wood, John H., "Analysis of the Earthquake Response of a Nine-Story Steel Frame Building During the San Fernando Earthquake," EERL 72-04, October 1972. (PB 215 823)
- 79. Jennings, Paul C., "Rapid Calculation of Selected Fourier Spectrum Ordinates," EERL 72-05, November 1972.
- 80. "Research Papers Submitted to Fifth World Conference on Earthquake Engineering, Rome, Italy, 25-29 June 1973," EERL 73-02, March 1973. (PB 220 431)
- 81. Udwadia, F.E. and M.D. Trifunac, "The Fourier Transform, Response Spectra and Their Relationship Through the Statistics of Oscillator Response," EERL 73-01, April 1973. (PB 220 458)
- 82. Housner, George W., "Earthquake-Resistant Design of High-Rise Buildings," DRC 73-01, July 1973. (N/A)

- 83. "Earthquakes and Insurance," Earthquake Research Affiliates Conference, 2-3 April, 1973, DRC 73-02, July 1973. (PB 223 033)
- 84. Wood, John H., "Earthquake-Induced Soil Pressures on Structures," EERL 73-05, August 1973. (N/A)
- 85. Crouse, Charles B., "Engineering Studies of the San Fernando Earthquake," EERL 73-04, March 1973. (N/A)
- 86. Irvine, H. Max, "The Veracruz Earthquake of 28 August 1973," EERL 73-06, October 1973.
- 87. Iemura, H. and P.C. Jennings, "Hysteretic Response of a Nine-Story Reinforced Concrete Building During the San Fernando Earthquake," EERL 73-07, October 1973.
- 88. Trifunac, M.D. and V. Lee, "Routine Computer Processing of Strong-Motion Accelerograms," EERL 73-03, October 1973. (N/A) (PB 226 047/AS)
- 89. Moeller, Thomas Lee, "The Dynamics of a Spinning Elastic Disk with Massive Load," DYNL 73-01, October 1973.
- 90. Blevins, Robert D., "Flow Induced Vibration of Bluff Structures," DYNL 74-01, February 1974.
- 91. Irvine, H. Max, "Studies in the Statics and Dynamics of Simple Cable Systems," DYNL-108, January 1974.
- 92. Jephcott, D.K. and D.E. Hudson, "The Performance of Public School Plants During the San Fernando Earthguake," EERL 74-01, September 1974. (PB 240 000/AS)
- 93. Wong, Hung Leung, "Dynamic Soil-Structure Interaction," EERL 75-01, May 1975. (N/A) (PB 247 233/AS)
- 94. Foutch, D.A., G.W. Housner, P.C. Jennings, "Dynamic Responses of Six Multistory Buildings During the San Fernando Earthquake," EERL 75-02, October 1975. (PB 248 144/AS)
- 95. Miller, Richard Keith, "The Steady-State Response of Multidegree-of-Freedom Systems with a Spatially Localized Nonlinearity," EERL 75-03, October 1975. (PB 252 459/AS)
- 96. Abdel-Ghaffar, Ahmed Mansour, "Dynamic Analyses of Suspension Bridge Structures," EERL 76-01, May 1976. (PB 258 744/AS)

- 97. Foutch, Douglas A., "A Study of the Vibrational Characteristics of Two Multistory Buildings," EERL 76-03, September 1976. (PB 260 874/AS)
- 98. "Strong Motion Earthquake Accelerograms Index Volume," Earthquake Engineering Research Laboratory, EERL 76-02, August 1976. (PB 260 929/AS)
- 99. Spanos, P-T.D., "Linearization Techniques for Non-Linear Dynamical Systems," EERL 76-04, September 1976. (PB 266 083/AS)
- 100. Edwards, Dean Barton, "Time Domain Analysis of Switching Regulators," DYNL 77-01, March 1977.
- 101. Abdel-Ghaffar, Ahmed Mansour, "Studies on the Effect of Differential Motions of Two Foundations upon the Response of the Superstructure of a Bridge," EERL 77-02, January 1977. (PB 271 095/AS)
- 102. Gates, Nathan C., "The Earthquake Response of Deteriorating Systems," EERL 77-03, March 1977. (PB 271 090/AS)
- 103. Daly, W., W. Judd and R. Meade, "Evaluation of Seismicity at U.S. Reservoirs," USCOLD, Committee on Earthquakes, May 1977. (PB 270 036/AS)
- 104. Abdel-Ghaffar, A.M. and G.W. Housner, "An Analysis of the Dynamic Characteristics of a Suspension Bridge by Ambient Vibration Measurements," EERL 77-01, January 1977. (PB 275 063/AS)
- 105. Housner, G.W. and P.C. Jennings, "Earthquake Design Criteria for Structures," EERL 77-06, November 1977. (PB 276 502/AS)
- 106. Morrison, P., R. Maley, G. Brady, R. Porcella, "Earthquake Recordings on or Near Dams," USCOLD, Committee on Earthquakes, November 1977. (PB 285 867/AS)
- 107. Abdel-Ghaffar, A.M., "Engineering Data and Analyses of the Whittier, California Earthquake of January 1, 1976," EERL 77-05, November 1977. (PB 283 750/AS)
- 108. Beck, James L., "Determining Models of Structures from Earthquake Records," EERL 78-01, June 1978. (PB 288 806/AS)
- 109. Psycharis, Ioannis, "The Salonica (Thessaloniki) Earthquake of June 20, 1978," EERL 78-03, October 1978. (PB 290 120/AS)

- 110. Abdel-Ghaffar, A.M. and R.F. Scott, "An Investigation of the Dynamic Characteristics of an Earth Dam," EERL 78-02, August 1978. (PB 288 878/AS)
- 111. Mason, Alfred B., Jr., "Some Observations on the Random Response of Linear and Nonlinear Dynamical Systems," EERL 79-01, January 1979. (PB 290 808/AS)
- 112. Helmberger, D.V. and P.C. Jennings (Organizers), "Strong Ground Motion: N.S.F. Seminar-Workshop," SL-EERL 79-02, February 1978.
- 113. Lee, David M., Paul C. Jennings and George W. Housner, "A Selection of Important Strong Motion Earthquake Records," EERL 80-01, January 1980. (PB 80 169196)
- 114. McVerry, Graeme H., "Frequency Domain Identification of Structural Models from Earthquake Records," EERL 79-02, October 1979.
- 115. Abdel-Ghaffar, A.M., R.F. Scott and M.J. Craig, "Full-Scale Experimental Investigation of a Modern Earth Dam," EERL 80-02, February 1980.
- 116. Rutenberg, Avigdor, Paul C. Jennings and George W. Housner, "The Response of Veterans Hospital Building 41 in the San Fernando Earthquake," EERL 80-03, May 1980.
- 117. Haroun, Medhat Ahmed, "Dynamic Analyses of Liquid Storage Tanks," EERL 80-04, February 1980.
Strong-Motion Earthquake Accelerograms Digitized and Plotted Data

Uncorrected Accelerograms

<u>Volume I</u>

<u>Part</u>
A
B
D D
Ē
F
G
н т
Ĵ
K
L
M N
Õ
P
Q
− K ⊸S
T
U
V
W X
Ŷ

	NTIS
Report No.	Accession No.
EERL 70-20	PB 287 847
EERL 70-21	PB 196 823
EERL 71-20	PB 204 364
EERL 71-21	PB 208 529
EERL 71-22	PB 209 749
EERL 71-23	PB 210 619
EERL 72-20	PB 211 357
EERL 72-21	PB 211 781
EERL 72-22	PB 213 422
EERL 72-23	PB 213 423
EERL 72-24	PB 213 424
EERL 72-25	PB 215 639
EERL 72-26	PB 220 554
EERL 72-27	PB 223 023
EERL 73-20	PB 222 417
EERL 73-21	PB 227 481/AS
EERL 73-22	PB 232 315/AS
EERL 73-23	PB 239 585/AS
EERL 73-24	PB 241 551/AS
EERL 73-25	PB 241 943/AS
EERL 73-26	PB 242 262/AS
EERL 73-27	PB 243 483/AS
EERL 73-28	PB 243 497/AS
EERL 73-29	PB 243 594/AS
EERL 73-30	PB 242 947/AS
	•

Strong-Motion Earthquake Accelerograms Digitized and Plotted Data

Corrected Accelerograms and Integrated Ground Velocity and Displacement Curves

<u>Volume II</u>

		NTIS
Part	Report No.	Accession No.
A	EERL 71-50	PB 208 283
В	EERL 72-50	PB 220 161
С	EERL 72-51	PB 220 162
D	EERL 72-52	PB 220 836
E	EERL 73-50	PB 223 024
F	EERL 73-51	PB 224 977/9AS
G	EERL 73-52	PB 229 239/AS
Н	EERL 74-50	PB 231 225/AS
I	EERL 74-51	PB 232 316/AS
J,K	EERL 74-52	PB 233 257/AS
L,M	EERL 74-53	PB 237 174/AS
Ň	EERL 74-54	PB 236 399/AS
0,P	EERL 74-55	PB 239 586/AS
Q,R	EERL 74-56	PB 239 587/AS
S	EERL 74-57	PB 241 552/AS
Т	EERL 75-50	PB 242 433/AS
U	EERL 75-51	PB 242 949/AS
v	EERL 75-52	PB 242 948/AS
W,Y	EERL 75-53	PB 243 719

Analyses of Strong-Motion Earthquake Accelerograms Response Spectra

Volume III

		NTIS
Part	Report No.	Accession No.
A	EERL 72-80	PB 212 602
В	EERL 73-80	PB 221 256
С	EERL 73-81	PB 223 025
D	EERL 73-82	PB 227 469/AS
Е	EERL 73-83	PB 227 470/AS
F	EERL 73-84	PB 227 471/AS
G	EERL 73-85	PB 231 223/AS
Н	EERL 74-80	PB 231 319/AS
I	EERL 74-81	PB 232 326/AS
J,K,L	EERL 74-82	PB 236 110/AS
M, N	EERL 74-83	PB 236 400/AS
0,P	EERL 74-84	PB 238 102/AS
Q,R	EERL 74-85	PB 240 688/AS
S	EERL 74-86	PB 241 553/AS
T	EERL 75-80	PB 243 698/AS
U	EERL 75-81	PB 242 950/AS
V	EERL 75-82	PB 242 951/AS
W,Y	EERL 75-83	PB 243 492/AS

Analyses of Strong-Motion Earthquake Accelerograms Fourier Amplitude Spectra

<u>Volume IV</u>

		NTIS
Part	Report No.	Accession No.
A	EERL 72-100	PB 212 603
В	EERL 73-100	PB 220 837
С	EERL 73-101	PB 222 514
D	EERL 73-102	PB 222 969/AS
E	EERL 73-103	PB 229 240/AS
F	EERL 73-104	PB 229 241/AS
G	EERL 73-105	PB 231 224/AS
H	EERL 74-100	PB 232 327/AS
Ι	EERL 74-101	PB 232 328/AS
J,K,L,M	EERL 74-102	PB 236 111/AS
N,O,P	EERL 74-103	PB 238 447/AS
Q,R,S	EERL 74-104	PB 241 554/AS
T,U	EERL 75-100	PB 243 493/AS
V,W,Y	EERL 75-101	PB 243 494/AS

Index Volume EERL 76-02 PB 260 929/AS